Economic Studies 208

Simon Ek Structural Change, Match Quality, and Integration in the Labor Market Department of Economics, Uppsala University

Visiting address:	Kyrkogårdsgatan 10, Uppsala, Sweden
Postal address:	Box 513, SE-751 20 Uppsala, Sweden
Telephone:	+46 18 471 00 00
Telefax:	+46 8 47 4 78
Internet:	http://www.nek.uu.se/

ECONOMICS AT UPPSALA UNIVERSITY

The Department of Economics at Uppsala University has a long history. The first chair in Economics in the Nordic countries was instituted at Uppsala University in 1741.

The main focus of research at the department has varied over the years but has typically been oriented towards policy-relevant applied economics, including both theoretical and empirical studies. The currently most active areas of research can be grouped into six categories:

- * Labour economics
- Public economics
- * Macroeconomics
- * Microeconometrics
- * Environmental economics
- Housing and urban economics

Simon Ek

Structural Change, Match Quality, and Integration in the Labor Market



UPPSALA UNIVERSITET Dissertation presented at Uppsala University to be publicly examined in Hörsal 2, Ekonomikum, Kyrkogårdsgatan 10, Uppsala, Friday, 21 April 2023 at 13:15 for the degree of Doctor of Philosophy. The examination will be conducted in English. Faculty examiner: Professor Maarten Goos (Utrecht University, School of Economics).

Abstract

Ek, S. 2023. Structural Change, Match Quality, and Integration in the Labor Market. *Economic studies* 208. 176 pp. Uppsala: Department of Economics, Uppsala University. ISBN 978-91-506-2991-0.

Essay I: Are workers with poor outside opportunities less responsive and more susceptible to negative demand shifts in routine occupations? To answer this, I create and estimate an occupation specialization index (OSI) using Swedish register data and machine learning tools. It measures the expected difference in utility between a worker's occupation and his best nonroutine outside option. This determines the loss he is willing to tolerate to avoid switching. Low-OSI employees disproportionately left routine work. Their future wage growth was akin to comparable workers initially in non-routine occupations. By contrast, routine specialists largely stayed put and experienced lower wage growth than generalists and non-routine specialists.

Essay II (with Adrian Adermon, Georg Graetz, and Yaroslav Yakymovych): Using a new identification strategy, we jointly estimate the growth in occupational wage premia and time-varying occupation-specific life-cycle profiles for Swedish workers 1996–2013. We document a substantial increase in between-occupation wage inequality due to differential growth in premia. The association of wage premium growth and employment growth is positive, suggesting that premium growth is predominantly driven by demand side factors.

Essay III (with Peter Fredriksson, Lena Hensvik, and Oskar Nordström Skans): We provide two pieces of evidence that workers' capacity to extract rents from match-specific productivity hinges on their outside options. Using a measure of match quality, derived from the relationship between workers' multidimensional skills and job-specific skill requirements, we show that: (i) wages within ongoing matches are more closely aligned with match quality following an improvement of local labor market conditions; (ii) wages of job-to-job movers are positively related to the match quality in the previous job, even when controlling for previous wage.

Essay IV (with Mats Hammarstedt, and Per Skedinger): In a field experiment we study the causal effects of previous work experience and language skills when Syrian refugees in Sweden apply for low-skilled jobs. We find no evidence of sizeable effects from experience or completed language classes on the probability of receiving callback from employers. However, female applicants were more likely than males to receive a positive response. As a complement to the experiment, we interview a select number of employers.

Keywords: Structural change, Outside options, Match quality, Integration of immigrants

Simon Ek, Department of Economics, Box 513, Uppsala University, SE-75120 Uppsala, Sweden.

© Simon Ek 2023

ISSN 0283-7668 ISBN 978-91-506-2991-0 URN urn:nbn:se:uu:diva-497751 (http://urn.kb.se/resolve?urn=urn:nbn:se:uu:diva-497751)

To my mother, Ihréne, and grandmother, Elly

Acknowledgements

Writing a thesis is no small feat, nor can it be accomplished swiftly. I am typically a rather restless person and used to things in life moving faster. I still vividly remember my first day in the PhD program, awkwardly consuming breakfast sandwiches at an introductory meeting while failing to memorize any names. Back then, there appeared to be a vast ocean of time between me and graduation. Now that ocean has been reduced to a few droplets, and I cannot comprehend how quickly time has passed.

This journey has been very rewarding, and I am grateful for having had the opportunity to write this thesis. I have had the privilege of interacting with many brilliant scholars in several exceptional academic environments. I have also had the freedom to pursue whatever research I deemed meaningful while having access to extensive support and resources. At the same time, it has been very challenging. I always enjoyed playing the role of student, enthusiastically absorbing material meticulously selected and conveyed by a teacher. Writing a dissertation is a different affair entirely. In doing so, one must do substantial groundwork in order to comprehend what we already know, what we ought to know, what is even knowable, and how one can convincingly bring about the knowledge that is missing. Given this formidable task, I am very glad for having had great people at my side throughout the entire process.

I am perhaps most grateful to my supervisors, Peter Fredriksson and Georg Graetz. Thank you Peter not only for your excellent, painstaking academic guidance, but also for your emotional support during spells of stress and doubt. Your door was always open, your office armchairs felt comfortable no matter the level of thesis progression, and you always had a joke up your sleeve that spoke perfectly to my sense of humour. Thank you Georg for all the help. I am impressed and humbled by your ability to perform sound economic thinking on the spot, the extent of your knowledge, and your dedication to research. Your feedback has been consistently precise and to the point.

My work has also benefited greatly from competent co-authors. In addition to my supervisors, I have the joy of working with Adrian Adermon, Mats Hammarstedt, Lena Hensvik, Oskar Nordström Skans, Per Skedinger, and Yaroslav Yakymovych. At times, I find it hard to see the big picture and deciding how to best advance a project. Thank you Lord for co-authors! In addition, they have all been great sources of inspiration. This thesis is in all respects (at least) a nine-person job.

I hold the Uppsala University economics department in high regard. This may be interlinked with me having obtained all my tertiary education at this department; I am what you could call an in-house product. Researchers who have been particularly helpful and generous with their time include Mikael Carlsson, Stefan Eriksson, Oscar Erixson, Christoph Hedtrich, Luca Repetto, Olof Åslund, and Erik Öberg. Moreover, the department would not function without our excellent administration. All the auxiliary work has been smooth thanks to you, be it registering for courses at other departments, applying for a research visit, or convincing banks that I may presumably be trusted with a mortgage (thank you Ulrika Öjdeby!).

Writing a thesis would be awfully dull without fellow PhD students. My cohort consisted of Anton, Edvin, Elin, Fei, Jan, Maxi, and Zeynep. It has been a delight spending time with you all these years! You also helped me through all the initial coursework. A special thank you to Edvin for being the best of office mates and for introducing me to Ester and Gunnar. Thanks also to all other PhD students, past and present, including some at SLU. Both Sofia and Raoul have functioned as role models and provided helpful feedback on my work. I have had many interesting conversations about economics with Adam, Alice, Akib, Alexander, Bahre, Dogan, Erik, Jacob, Jonathan, Josefin, Lillit, Malin, Melinda, Olle, Rinni, and Xiao, to name a few. Others that made the PhD experience more enjoyable include Adrian, Daniel, Erika, Gaëlle, Hanfeng, Henning, Ida, Jakob, Majken, Markus, Polina, Sebastian, Tianze, Tram, and Tsz Chun.

During my penultimate year of the PhD program, I carried out a brief but inspiring research visit to the Centre for Economic Performance (CEP) at the London School of Economics, generously funded by *Jan Wallanders och Tom Hedelius stiftelse*. Thanks to Stephen Machin for inviting me, to my host Guy Michaels, and to the PhD students I shared offices with at the CEP.

I also wish to extend my gratitude to my former employers Lars Calmfors and Magnus Henrekson. You taught me to care as much about the questions as the methods, the importance of policy relevance, and how rewarding practical economic thinking can be. I do not think I would have pursued a PhD had it not been for my time working for you at the Research Institute of Industrial Economics and the Swedish Labour Market Policy Council in Stockholm. During this period, I also learned much from Niclas Berggren, Fredrik Heyman, Ann-Sofie Kolm, Pehr-Johan Norbäck, Martin Olsson, Tuomas Pekkarinen, Lars Persson, Per Skedinger, Roger Svensson, Björn Tyrefors, and Joacim Tåg.

My family is truly irreplaceable. My partner Hanna has been extraordinary during the last four or so years, lending support whenever I felt overwhelmed and making sure I did not take myself or my work too seriously. I love you. My sister and father are also outstanding. Buying a cottage with you is one of the best decisions I have made, because it means spending more time together. I also appreciate that you care little for what I do professionally. I have therefore never felt much external pressure to perform well in academia, enabling me to do things solely because I find them interesting. I am also blessed with a brilliant extended family through Annette, Birgit, Ebba, Mathias, Molly, Monica, Rikard, and Örjan. Thank you for all the Christmases and birthday celebrations throughout the years. Finally, I want to name a few close friends: Anton and Gabriella, thank you for always making me feel welcome whenever I visit. Spending time with you, Noel, and Isak has provided much-needed distraction from my work; Sebastian, despite the long periods of time in between, when we see each other, it feels like no time has passed. I appreciate our solid friendship; Annika, Alf, Ellen, and Klara, thank you for the many midsummer parties and get-togethers.

Uppsala, March 2023 Simon Ek

Contents

Int	roducti	on	. 13
	1	The essays	. 15
Ι	Worker	Specialization and the Consequences of Occupational Decline Introduction	. 21
	2	Conceptual framework	2.5
	3	Data	30
	4	Empirical analysis	33
	5	Conclusions	. 45
	A	Derivations	. 52
	В	The artificial neural network	. 57
	С	Additional figures	. 58
Π	Under	standing Occupational Wage Growth	. 69
	1	Introduction	. 70
	2	Theoretical framework and empirical strategy	. 72
	3	Data description	. 80
	4	Results	. 81
	5	Conclusion	. 93
	А	Procedure for estimating occupation-specific flat spots	. 97
	В	Additional figures and tables	. 99
III	Outsi	de Options and the Sharing of Match-Specific Rents	109
	1	Introduction	110
	2	Theoretical framework	111
	3	Data, variables and sample selection	116
	4	Empirical analysis	119
	5	Conclusions	129
	А	Simulations	132
	В	Robustness checks for remaining workers	136
	С	Robustness checks for workplace movers	141
IV	Low-	Skilled Jobs, Language Proficiency and Job Opportunities for	
Re	fugees		145
	1	Introduction	146
	2	The immigrant population in Sweden	149
	3	The field experiment	152
	4	The employer interviews	167

5	Conclusions	171
А	Application letter and resume	176

Introduction

The market for labor is crucial in many respects. Nearly all persons interact with it throughout much of their lives. Moreover, individuals make significant life choices and substantial investments in order to succeed in this market. Due to its importance, the labor market warrants much attention and research.

I firmly maintain that, as researchers, we have an obligation to improve knowledge about questions that (we at least believe) have value beyond academia, not least for making sound policy. I think many scholars within the field of labor economics are doing just that: Identifying questions that non-academics and policy makers deem important, and answering them the best they can, whether they concern structural change, education, wage setting, labor market power, the consequences of job loss, discrimination, (in)equality of opportunity and outcomes, social security, or the labor market integration of immigrants. I have tried my best to focus on policy-relevant topics that I believe are of such merit. In doing so, this thesis has come to concern three main themes: Labor market structural change, wage bargaining in matches between firms and workers, and the labor market integration of immigrants.

Throughout history, new technological advancements and ways of organizing work have transformed the market for labor. This has led to continuing advancements in welfare and wealth. But technological transitions have not been seamless, and far from everyone has benefited from them (e.g., Autor, 2015). Still today, there is much concern about how workers will handle future shifts in demand for labor across different types of jobs. The ability to cope will determine both their life-time earnings and the value created in the labor market. Although previous automation mainly concerned routine (i.e., easily codifiable and repetetive) tasks typically found in manufacturing, future technology is expected to be able to perform a broader repertoire of both lowerand higher-skilled work (e.g., Brynjolfsson and McAfee, 2014). Halting technological progress and structural change is neither feasible, nor desirable. But mitigating negative consequences of such shifts for workers should be a priority for policy makers. One key to doing so is to identify which occupations are likely to see negative shifts in demand. Indeed, there exists numerous forecasts on future employment and labor demand at the occupation level (e.g., Frey and Osborne, 2017; Arntz et al., 2017; Webb, 2019). But we do not care about occupations per se, but rather the workers in them. Many workers will likely do well following such shifts, while others will struggle. Thus, it is not sufficient to identify the occupations that will decline. One also needs to identify the workers in them that are vulnerable to negative structural shifts. Despite its

importance, there is little work on this topic. This is what the first chapter of my thesis is about. In it, I develop and substantiate a theory-consistent way of characterizing worker susceptibility to negative shifts in labor demand at the occupation level.

Even after a shift in, say, occupation employment has occurred, it is neither straightforward to determine what the underlying causes of it were, nor is it uncomplicated to gauge the consequences for workers. Key to understanding the causes is the relationship between such shifts and the changes over time in the wage returns paid in different occupations. These wage returns are also critical for outcomes such as wage inequality. But they are notoriously difficult to estimate. One peculiarity that has puzzled researchers is the absence of any clear relationship between the growth in employment and average wages at the occupation level across industrialized countries (see the discussion in, e.g., Böhm et al., 2023). For instance, even though employment has declined substantially, average wages paid in routine occupations in manufacturing have remained relatively high. This is a conundrum considering that the employment changes appear to be mainly caused by shifts in demand, which should result in lower wages. But a recent literature suggests that systematic occupation switching behaviour of workers may be concealing shifts in the wages paid for a unit of skill in these occupations, i.e., the occupation wage premium (Cortes, 2016; Böhm, 2020; Cavaglia and Etheridge, 2020; Böhm et al., 2023). The second chapter adds to this literature by developing a new method for estimating these occupational wage premia.

Another important topic that has received much attention recently is that of employer labor market power (e.g., Lamadon et al., 2022; Berger et al., 2022). The degree to which firms can exert power over wage setting is important for, e.g., the labor share of value added. How then can a worker obtain as high a wage as possible? Sorting into jobs for which a worker is a good match is pivotal. But firms may not have to pay the full value of a worker's labor. A strand of search and matching theory (e.g., Postel-Vinay and Robin, 2002; Cahuc et al., 2006) makes an intuitive proposition: In order for workers to extract rents from a relationship with an employer, it is necessary but not sufficient for that relationship to be of high value. Workers also need strong outside offers to use as a bargaining chip. The current employer will then be forced to bargaining (directly or indirectly) with the employer who gave the outside offer in order to hope to retain the worker, resulting in a higher wage. Anecdotally, this is a common scenario: Workers can bring job offers to their current employer to get a better wage deal. This type of bargaining also has implications at the aggregate level: If outside offers are abundant, for instance when the labor market is tight, workers can extract more value from strong matches. The third chapter presents two sharp tests of this hypothesis.

Finally, for a long time, I have taken an interest in the public debate regarding the labor market integration of immigrants in Sweden. This is one of the major challenges that Swedish policy makers currently face, and it is interlinked with many other public concerns. Two of the questions that have dominated this debate are how to enable immigrants to successfully enter the labor market as quickly as possible, with the hope that this results in long-term establishment, and what role language skills play in facilitating this. The main tool for improving language proficiency among immigrants is language training, specifically the Swedish for Immigrants (SFI) program. Generally, immigrants who enter the labor market soon after arrival with a good proficiency in Swedish tend to do better in the long term. But due to, e.g., unobserved ability, it is difficult to determine to what extent this is caused by labor market experience and language proficiency in itself. The fourth chapter of this thesis studies how much emphasis employers put on previous experience and completed SFI in recruitment processes.

1 The essays

1.1 Worker Specialization and the Consequences of Occupational Decline

According to a standard Roy model, when demand for labor in an occupation declines, the utility loss of an incumbent worker is determined by the initial difference between the utility associated with his current occupation and his best outside option in the set of occupations unaffected by the shock. I refer to this difference as his degree of occupation specialization. Generalist workers with good outside options will leave quickly and be better off, while highly specialized workers willingly remain and tolerate the full effect of the demand shift through lower wages.

This essay shows that under certain distributional assumptions on occupation utility, the expected value of the above utility difference can be inferred from the *ex ante* probability of a worker being observed in the set of unaffected outside occupations, given his traits. I name this function the occupation specialization index (OSI). To construct it empirically, I train an artificial neural network to predict occupation choice probabilities using data from Sweden on detailed worker characteristics, including multidimensional abilities, age, education, region of residence, and industry-specific work experience.

The index is then used to shed light on the historical decline in employment in routine-intensive occupations. I compare the difference in long-run occupation switching behaviour and career outcomes between workers initially employed in routine and non-routine occupations across the specialization distribution by means of a difference-in-differences-styled specification. This allows for holding fixed any general effect of being specialized that is not related to negative demand shifts. The probability of leaving one's initial occupation depends strongly on initial OSI. This suggests that the choice of leaving routine-intensive occupations were often voluntary and guided by outside options. Moreover, the long-run earnings growth penalty of working in a routine-intensive occupation increases with the OSI. This effect acts through wages rather than employment.

Understanding better the historical shift in demand for routine work and its consequences for incumbent workers is important in its own right. But the results also substantiate the general usefulness of the specialization index in determining which workers are susceptible to negative demand shifts. The index is solely based on current information. Therefore, it could be used to characterize workers employed in occupations today that are believed to decline or even disappear in the future by susceptibility to such shifts.

1.2 Understanding Occupational Wage Growth

Co-authored with Adrian Adermon, Georg Graetz, and Yaroslav Yakymovych.

We develop a novel method for estimating occupation-specific wage premium growth—that is, the growth in the return per unit of skill or productivity—which can be applied to relatively small amounts of data. It relies on two fundamental ideas: Observing the wage growth of occupation stayers between two adjacent years in order to address issues with selection in occupation switching; and the notion of a "flat spot" in the experience profiles in each occupation where the return to an additional year of experience equals zero. The method allows for both estimating the wage premium growth between one year and the next as well as the wage-experience profiles at the occupation level. We then accumulate the year-on-year wage premium growth estimates to obtain the long-run growth in wage premia.

We implement our method on data for Sweden for the years 1996–2013. There is a modest, yet clear positive relationship between estimated wage premium growth and employment growth at the occupation level. Moreover, by means of a decomposition exercise, we find that the relative premia changes contributed substantially to the increase in overall wage inequality, but that this is masked by worker sorting. Finally, we document large heterogeneity in life-cycle profiles across occupations, which have also seen substantial shifts across time.

1.3 Outside Options and the Sharing of Match-Specific Rents

Co-authored with Peter Fredriksson, Lena Hensvik, and Oskar Nordström Skans.

A relatively recent strand of search and matching models (e.g., Postel-Vinay and Robin, 2002, Cahuc et al., 2006, Yamaguchi, 2010, Bagger and Lentz, 2019) suggests that workers in productive matches can use outside offers to pit employers against each other, thereby bidding up their wage. Thus, a valuable match and outside options are jointly necessary to reach a high wage, and well-matched workers can extract more rents when outside offers are abundant. Put differently, we should expect a positive interaction effect between workers' match quality and expected outside options on wages. Is this type of bargaining an important feature of the labor market? Although there exists some structural econometric work and survey evidence on bargaining regimes, to the best of our knowledge, there are no reduced-form tests of this theory.

This chapter provides two pieces of evidence on how outside options affect the transmission from match-specific productivity to wages. We begin by constructing an empirical measure of the quality of a match between a job and a worker. To this end, we follow Fredriksson et al. (2018) by leveraging Swedish enlistment data on multidimensional abilities and comparing the conformity between a worker's skill set and those of his tenured co-workers. This metric has multiple desirable empirical traits.

In a first exercise, we show that there is a positive, economically significant interaction effect on wages between match quality and several proxies of labor market tightness, which theoretically governs the probability of receiving outside offers. These proxies include local unemployment, a shift-share instrument based on changes in industry employment and local industry composition, and occupation-specific employment, to name a few. Secondly, we home in on job switchers. Here, our theoretical framework provides a sharp prediction: If workers are able to pit employers against each other, not only should the previous wage matter for the wage following a job transition, previous match quality should also contribute to the new wage. We find that even when conditioning on previous wage, prior match quality increases the wage in the new job. But this is only true in the absence of any interruption in the employment spell. Contrary to canonical bargaining models, these results suggest that match-specific factors and outside options are not additively separable in wage formation.

1.4 Low-skilled Jobs, Language Proficiency and Job Opportunities for Refugees

Co-authored with Mats Hammarstedt, and Per Skedinger.

This chapter focuses on two central aspects of the labor market integration of immigrants with low levels of education: The value of previous labor market experience, and language training. A common idea is that, in itself, entering the labor market improves the subsequent outlook of immigrants by, e.g., providing valuable experience and acting as a productivity signal to other employers. The fact that there is a positive association between language proficiency and employment is not seldom taken as proof of the importance of language training. Despite the relevance for policy, causal evidence on how these factors influence the integration process is scant. One way to better understand what role they play is by studying employers' assessment of job candidates with varying characteristics. More specifically, we analyze the emphasis employers put on completing the full set of courses in the Swedish for Immigrants (SFI) program and having previous work experience as a restaurant assistant by means of a field experiment. We created eight fictitious refugee job seekers with different CV:s who immigrated from Syria in 2016. These were randomly assigned to apply to advertised low-skilled job vacancies. We also complement the experiment with interviews with a handful of employers experienced in handling applications for low-skilled jobs from immigrants.

Previous work experience and completed SFI seem to provide at best a small positive signal when refugees apply for low-skilled jobs through formal channels. This indicates that any potential positive effects must act through other mechanisms such as human capital accumulation, professional networks, or a better comprehension of the Swedish labor market. Moreover, our most salient result is that female applicants receive substantially more callbacks than male ones. This suggests that the lower employment rate among immigrant women compared to men may not be explained by worse employment prospects.

References

- Arntz, M., T. Gregory, and U. Zierahn (2017). "Revisiting the risk of automation". *Economics Letters* 159, 157–160.
- Autor, D. (2015). "Why Are There Still So Many Jobs? The History and Future of Workplace Automation". *Journal of Economic Perspectives* 29.3, 3–30.
- Bagger, J. and R. Lentz (2019). "An empirical model of wage dispersion with sorting". *Review of Economic Studies* 86, 153–190.
- Berger, D., K. Herkenhoff, and S. Mongey (2022). "Labor market power". *American Economic Review* 112.4, 1147–93.
- Böhm, M. (2020). "The price of polarization: Estimating task prices under routine-biased technical change". *Quantitative Economics* 11.2, 761–799.
- Böhm, M., H.-M. v. Gaudecker, and F. Schran (2023). "Occupation Growth, Skill Prices, and Wage Inequality". *Journal of Labor Economics*, forthcoming.
- Brynjolfsson, E. and A. McAfee (2014). *The second machine age: Work, progress, and prosperity in a time of brilliant technologies.* WW Norton & Company.
- Cahuc, P., F. Postel-Vinay, and J.-M. Robin (2006). "Wage bargaining with on-the-job search: Theory and evidence". *Econometrica* 74.2, 323–364.
- Cavaglia, C. and B. Etheridge (2020). "Job polarization and the declining quality of knowledge workers: Evidence from the UK and Germany". *Labour Economics* 66, 101884.
- Cortes, G. M. (2016). "Where have the middle-wage workers gone? A study of polarization using panel data". *Journal of Labor Economics* 34.1, 63–105.
- Fredriksson, P., L. Hensvik, and O. N. Skans (2018). "Mismatch of talent: Evidence on match quality, entry wages, and job mobility". *American Economic Review* 108.11, 3303–38.
- Frey, C. B. and M. A. Osborne (2017). "The future of employment: How susceptible are jobs to computerisation?" *Technological Forecasting and Social Change* 114, 254–280.
- Lamadon, T., M. Mogstad, and B. Setzler (2022). "Imperfect Competition, Compensating Differentials, and Rent Sharing in the US Labor Market". *American Economic Review* 112.1, 169–212.
- Postel-Vinay, F. and J.-M. Robin (2002). "Equilibrium wage dispersion with worker and employer heterogeneity". *Econometrica* 70.6, 2295–2350.
- Webb, M. (2019). "The impact of artificial intelligence on the labor market". *Mimeo*.
- Yamaguchi, S. (2010). "Job search, bargaining, and wage dynamics". *Journal* of Labor Economics 28.3, 595–631.

Essay I. Worker Specialization and the Consequences of Occupational Decline

Acknowledgements: I am very grateful to my supervisors Peter Fredriksson and Georg Graetz for their continued support. I also thank Adrian Adermon, Teodora Borota Milicevic, Michael Böhm, Lars Calmfors, Mikael Carlsson, Tiernan Evans, Christoph Hedtrich, Magnus Henrekson, Sofia Hernnäs, Edvin Hertegård, Raoul van Maarseveen, Alan Manning, Guy Michaels, Oskar Nordström Skans, Per Skedinger, Joacim Tåg, Yaroslav Yakymovych, Olof Åslund, Özge Öner, and participants at workshops and seminars at the Department of Economics at Uppsala University, the Centre for Economic Performance at London School of Economics, the Institute for Evaluation of Labour Market and Education Policy (IFAU), the Research Institute of Industrial Economics (IFN), and the Department of Economics and Statistics, Linnaeus University for valuable comments and discussions. Any errors are my own.

1 Introduction

Many developed economies have experienced large occupational structure shifts in recent decades. Generally, routine-intensive occupations with codifiable tasks have declined in favor of non-routine work. This is partly due to advancements in labor-replacing technologies. The occupations that declined were commonly located in the middle of the wage distribution. These shifts have therefore contributed to wage polarization. Moreover, a recent literature finds that the growth over time in occupation wage premia and employment are positively correlated. This strongly suggests that these occupation structure shifts, including the substantial decline in routine work, were caused by changing demand.¹

There are strong public concerns regarding how well incumbent workers will cope with future shifts in labor demand across occupations.² This paper develops a method grounded in theory for identifying which workers are particularly vulnerable to negative shifts in occupation labor demand.³ It is based on estimating how much value a worker puts on his current occupation relative to his outside options. This difference is referred to as workers' degree of occupation specialization. I then ask two questions: First, how does specialization relate to the career consequences of incumbent employees following the historical decline in routine work? This query is important in its own right. But it is also a means of substantiating the general usefulness of my method for predicting worker susceptibility. Second, what is the nature of worker flows out of routine occupations? This is informative about the process by which these occupations decreased in size. It also speaks to whether mobility was voluntary for workers lacking good alternatives.

I begin by setting up a Roy-style discrete choice model with deterministic (which depend on workers' characteristics) and idiosyncratic utility terms. When a demand shock lowers the wage premium in a worker's occupation, his utility loss is determined by the difference between his utility in that occupation and his best non-shocked outside option, i.e., his specialization. Workers with little specialization move to a now-more attractive option. Highly specialized workers remain, losing more utility. I hypothesize that this effect is work-

¹See, e.g., Goos et al. (2014) and Goos et al. (2019) for occupation structure shifts across countries; Autor et al. (2003), Goos and Manning (2007), Autor et al. (2008), and Adermon and Gustavsson (2015) for the literature on routine-biased technological change and wage polarization; Cortes (2016), Böhm (2020), Cavaglia and Etheridge (2020), and Böhm et al. (2023) for studies on occupational wage premium and employment growth.

²See, e.g., Brynjolfsson and McAfee (2014), Mokyr et al. (2015), Frey and Osborne (2017), and OECD (2019). For example, future technology is envisioned to be able to perform many tasks previously considered impossible, such as writing news articles and driving cars.

³There exists many projections (e.g., the U.S. Bureau of Labor Statistics Occupation Projections) and ample work (e.g., The OECD Future of Work initiative, Frey and Osborne, 2017, Arntz et al., 2017, and Webb, 2019) on future occupation employment. I am not aware of any predictions at the worker level. This paper provides a framework for making such projections.

ing partly through wages rather than only through amenities. For workers in occupations that did not experience any negative demand shift, specialization should be less important for future outcomes. Next, I show that under Gumbel distributed idiosyncratic preferences, expected specialization is a function of workers' *ex ante* probability of working in a non-shocked outside option. I name this expected value the occupation specialization index (OSI). I also demonstrate that ordering workers by the OSI is equivalent to ordering them by the expected utility loss from a negative wage premium shock of any size.

To construct the OSI empirically, I train an artificial neural network (ANN). The network predicts occupation choice probabilities using Swedish register data on male worker characteristics. These include multidimensional abilities (collected during the Swedish enlistment process), educational attainment, region of residence, age, and industry-specific experience. The ANN uses the same formula as the multinomial logit, consistent with the theoretical framework. But it requires less strict functional form assumptions on utility and can accommodate important but unknown non-linear and interaction effects. I use Autor and Dorn (2013)'s routine task intensity index (RTI) to classify occupations as either routine (above) or non-routine (below-median RTI). Between 2001 and 2013, non-routine occupations remained stable or grew while routine occupations declined (by 6 percentage points on average).

The OSI is expected to be more negatively related to occupation switching and wage growth for routine than non-routine workers. To test this, the OSI is related to the long-run (up to twelve years into the future) career outcomes of workers observed in 1997-2001 separately by initial routine and non-routine low- to middle-skilled occupations.⁴ The probability of leaving routine work depends strongly on initial OSI. I interpret this as switching typically being voluntary, guided by the attractiveness of workers' outside options. The wage growth penalty of initially working in a routine occupation increases in the OSI. On average, routine specialists experienced lower wage and earnings growth than both low-OSI workers in either type of occupation and non-routine specialists. These results are consistent with the prediction that low-OSI "generalists" are better able to avoid losses from declining demand by transitioning to more attractive occupations.

The paper contributes to the literature on routine-biased technological change. Several previous papers in this literature highlight that susceptibility to negative demand shifts is determined by the difference between a worker's current utility and potential utility in his non-shocked options. My addition is to show that, under certain assumptions, my occupation specialization index measures the expected value of this difference. The index is simply a monotone transfor-

⁴With only one exception, the routine occupations are classified as low- or middle-skilled. To obtain a more comparable comparison group of non-routine occupations, I exclude higher-skilled occupations in the main analysis. But these are included in robustness checks. The low- to middle-skilled routine occupations are mainly concentrated in manufacturing while the non-routine are found in, e.g., services, construction, and transportation.

mation of the *ex ante* propensity for working in a non-shocked outside option. To the best of my knowledge, this closed-form solution has not been utilized before. I also demonstrate that the index predicts well which incumbents lost from the historical decline in demand for routine work. Hence, the Roy (1951) model seemingly provides an accurate representation of who loses when occupation labor demand declines. Moreover, the observable worker attributes at hand can be used to characterize workers' occupation options well.

There exists a few studies on the consequences of working in declining or routine occupations. Edin et al. (2022a) follow workers in occupations that later experienced arguably unanticipated decline using both Swedish and U.S. data. Ross and Ukil (2021) relate future industry employment to the future earnings of workers in the NLSY. Böhm (2020) and Jaimovich et al. (2021) use cross-sectional AFQT data to study changes in outcomes of workers with skill bundles fit for routine work. Cortes et al. (2017) describe which demographic groups in the U.S. contributed to the decline in routine employment. Bachmann et al. (2019) demonstrate that working in routine occupations in Germany is associated with low job stability and a high risk of unemployment. Cortes et al. (2020) find that the decline in routine work can to a large extent be accounted for by changing transition rates from non-employment to routine occupations.

More generally, there is a large literature explaining occupation switching behaviour (e.g., Jovanovic and Nyarko, 1997; Gathmann and Schönberg, 2010; Groes et al., 2015; Cortes and Gallipoli, 2018). I also relate to research using direct measures of worker attributes to infer job-specific match quality (e.g., Fredriksson et al., 2018; Guvenen et al., 2020; Lise and Postel-Vinay, 2020) and to the literature on the importance of different types of skills (e.g., Lindqvist and Vestman, 2011; Deming, 2017; Roys and Taber, 2022; Edin et al., 2022b). Finally, there are some studies of the role of occupation-specific human capital (often proxied by the distance between the task content of their initial and other occupations) for the ability to adjust to, e.g., mass layoffs (Robinson, 2018) and trade shocks (Traiberman, 2019; Eggenberger et al., 2022).

Section 2 describes the discrete choice model, the specialization index, and the econometric framework for the empirical analysis. Section 3 presents the data. Section 4 reports the empirical results. Section 5 concludes. Appendix A presents the derivation of the specialization index. The Neural Network is described in detail in Appendix B. Additional tables and figures are found in Appendix C.

2 Conceptual framework

2.1 A discrete occupation choice model

Setup

Consider a setting where workers are characterized by multiple attributes collected in the vector \mathbf{x} . There exists a finite number of occupations collected in the set K that are either classified as routine ($R \subset K$) or non-routine ($N \subset K$). The utility of worker i in occupation k is:

$$u_{ik} = \pi_k + m_k(\mathbf{x}) + q_k(\mathbf{x}) + \varepsilon_{ik}.$$
 (1)

 π_k represents the wage premium in k for a unit of skill, $m_k(\mathbf{x})$ determines productivity in k and $q_k(\mathbf{x})$ captures any amenities. ε_{ik} is an idiosyncratic term that may influence both the wage and amenities. Define deterministic utility as $u_k(\mathbf{x}) \equiv \pi_k + m_k(\mathbf{x}) + q_k(\mathbf{x})$. Workers choose the occupation with the highest utility. Denote a worker's initial choice by j.

Utility loss following a routine wage premium shock

Due to automation, there is a wage premium shock to all routine occupations of size $-\delta$, with $\delta > 0$. Define $d_j \equiv \delta \mathbb{1}[j \in R]$. Workers in *R* will choose whether to switch to another occupation. The initial occupation, *j*, yields a higher utility than any other occupation in *R* both before and after the shock. Therefore, the only relevant options are the non-shocked occupations in *N*. Since $d_j = 0$ for workers in *N*, their outside options do not matter. But if their occupation would experience the shock, the relevant option is the best choice in the set of non-routine occupations excluding *j*, i.e., $N \setminus \{j\}$.

The utility loss from staying in j is $-d_j$. The loss from switching is the difference between the initial utility in j and the best, non-shocked, outside option, i.e., the initial utility surplus. The worker will choose the option with the smallest associated loss. Thus, the change in utility is:

$$\Delta u_{ij} = \max\left\{-d_j, -\left(u_{ij} - \max_{n \in N \setminus \{j\}} u_{in}\right)\right\} \le 0.$$
(2)

The loss, in absolute terms, is bounded above by workers' utility surplus relative to their best non-routine outside option. I call this their degree of occupation specialization. Specialists with a large surplus remain and tolerate the full effect of the wage premium shock. Workers with a small surplus will instead move and experience a smaller utility loss.

For routine workers, $N \setminus \{j\}$ equals *N*. Thus, the surplus is defined slightly different for routine and non-routine workers. But it captures the difference between current utility and the utility associated with the best non-routine outside option for both groups.

Gumbel distributed idiosyncratic terms

I henceforth assume that all ε_{ik} are IID standard extreme value type I (or Gumbel) distributed. Although the utility surplus in (2) is never observed, four key properties of the Gumbel distribution allow me to characterize the distribution of this surplus: First, the occupation choice probabilities follow the multinomial logit (see McFadden, 1973):

$$p_k(\mathbf{x}) = \frac{e^{u_k(\mathbf{x})}}{\sum\limits_m e^{u_m(\mathbf{x})}}.$$
(3)

Second, the distribution of maximum utility before conditioning on occupation choice is also Gumbel distributed, with known location and scale. Third, using results from Hanemann (1984), the maximum utility conditional on any optimal choice j can be shown to be distributed the same as the unconditional maximum. Fourth, the difference between two same-scaled Gumbels is known to be logistic distributed. The details of characterizing the utility surplus distribution are reported in Appendix A.

2.2 The occupation specialization index

I now turn to finding a closed-form expression for the expected value of the utility surplus. I name this expression the occupation specialization index, or OSI. It reveals the expected utility a worker would lose if leaving his current occupation. Thus, one may interpret the OSI as how dependent, on average, workers are on their occupation for utility. The derivation of the OSI, and its properties, is described in Appendix A.

The index

First, define the *ex ante* probability of working in a non-routine outside option as follows. It is determined by a worker's characteristics, his observed occupation, and the set of non-routine occupations:

$$\rho(\mathbf{x}, j, N) \equiv \sum_{n \in N \setminus \{j\}} p_n(\mathbf{x}).$$
(4)

The expected value of the utility surplus can then be shown to be a monotonically decreasing function of $\rho(\mathbf{x}, j, N)$. This is the occupation specialization index:

$$OSI(\boldsymbol{x}, j, N) \equiv \mathbb{E}\left[u_{ij} - \max_{n \in N \setminus \{j\}} u_{in} \mid \boldsymbol{x}, j\right] = -\frac{\ln\left(\rho(\boldsymbol{x}, j, N)\right)}{1 - \rho(\boldsymbol{x}, j, N)}.$$
 (5)

To the best of my knowledge, this metric has neither been used previously in any work on occupation decline as well as routine-biased technological change, nor been derived in the theoretical discrete choice literature. The OSI is more general than in my application: It can be used to infer the expected utility surplus of any observed choice relative to a subset of alternatives.

Relationship with expected loss

The expected value of the utility change following the wage premium shock in (2) is a function of only d_i and $\rho(\mathbf{x}, j, N)$:

$$\mathbb{E}\left[\Delta u_{ij} \mid \boldsymbol{x}, j\right] = -\frac{d_j - \ln\left(1 + \left(e^{d_j} - 1\right)\rho\left(\boldsymbol{x}, j, N\right)\right)}{1 - \rho\left(\boldsymbol{x}, j, N\right)}.$$
(6)

For any value of $d_j = \delta > 0$, it is monotonically increasing (i.e., the absolute loss becomes smaller) in $\rho(\mathbf{x}, j, N)$. Since the OSI is decreasing in $\rho(\mathbf{x}, j, N)$, ordering workers by the OSI is equivalent to ordering them by absolute expected loss for any shock size.

Interpretation of the index

To summarize, the utility loss from an occupation wage shock is determined by the difference between a worker's utility in his occupation and his best nonshocked outside option. The expected value of this difference can be inferred from his *ex ante* probability of working in an outside, non-shocked occupation via the OSI. The OSI can also be used to order workers by expected loss from a shock of unknown size.

Intuitively, the OSI can be comprehended as follows. The extent to which a worker's peers with similar characteristics are observed in outside non-routine occupations carries a signal about his non-routine options. If no other workers with, say, a comparable skill set and education background works in an occupation other than his, these options are likely unattractive or unavailable to him. This idea is similar in spirit to the widely used revealed comparative advantage metric by Balassa (1965). It is also related to, e.g., Fredriksson et al. (2018) who measure match quality by the similarity of the skills of workers and their co-workers, Coraggio et al. (2022) who define match quality as the probability of being observed in an occupation-industry cell, and the outside options index developed by Caldwell and Danieli (2022).

Although this framework concerns utility, the empirical section deals with, e.g., wages and earnings. Workers that remain in occupations with decreasing wage premia will experience the effect on utility through wages. For switchers, however, utility may be influenced through wages or amenities. These are difficult to disentangle. But in Section 4.1, I show that specialization correlates positively with the wage surplus that a worker enjoys in his occupation.

Estimation

To construct the OSI empirically, I estimate occupation choice probabilities using the formula from (3). This is also known as the softmax activation function. Instead of using a multinomial logit, it can be used to estimate $p_k(\mathbf{x})$

by an artificial neural network (ANN; Described in detail in appendix B). This flexible machine learning algorithm requires less strict functional form assumptions on $u_k(\mathbf{x})$ regarding non-linear and interaction effects. But to attest that the results are robust to choice of method, I also estimate $p_k(\mathbf{x})$ using a multinomial logit assuming that $u_k(\mathbf{x}) = \vartheta'_k \mathbf{x}$. I then calculate $\rho(\mathbf{x}, j, N)$ in (4) using the predicted probabilities. Finally, $\rho(\mathbf{x}, j, N)$ is inserted into the OSI formula in (5). To avoid outliers, I censor the OSI at the 1st and 99th percentile. The OSI is difficult to interpret in levels, and is therefore standardized to mean zero, standard deviation one.

2.3 Specialization based on observed wages

Definition

As a complement to the OSI, I estimate the expected difference between a worker's log wage in *j* and $N \setminus \{j\}$. I refer to this as the wage surplus, or WOSI. Let $\mathbb{E}[w_{ia} \mid \mathbf{x}, b]$ be the expected log wage in occupation set *a* of workers in set *b*. $\mathbb{E}[w_{iN\setminus\{j\}} \mid \mathbf{x}, j]$ is not observed. But one can assume to what extent workers in $N \setminus \{j\}$ are representative of workers in *j* with similar characteristics. The expected log wage surplus can be written as:

$$WOSI(\mathbf{x}, j, N) = \mathbb{E}[w_{ij} - w_{iN \setminus \{j\}} | \mathbf{x}, j]$$

= $\mathbb{E}[w_{ij} | \mathbf{x}, j] - \sum_{n \in N \setminus \{j\}} \frac{p_n(\mathbf{x})}{\sum_{m \in N \setminus \{j\}} p_m(\mathbf{x})} \mathbb{E}[w_{in} | \mathbf{x}, n] + e(\mathbf{x}, j, N), \quad (7)$

where $e(\mathbf{x}, j, N) = \mathbb{E}[w_{iN \setminus \{j\}} | \mathbf{x}, N \setminus \{j\}] - \mathbb{E}[w_{iN \setminus \{j\}} | \mathbf{x}, j]$ is the error with which a worker's wage in $N \setminus \{j\}$ is predicted.

Estimation

To construct the WOSI empirically, I first demean w_{ijt} separately by year. I then estimate $w_{ijt} = \boldsymbol{\omega}'_j \boldsymbol{x}_{it} + \varepsilon_{ijt}$ using OLS. Next, I substitute all expected values in (7) with $\boldsymbol{\omega}'_k \boldsymbol{x}_{it}$ and the choice probabilities with the ANN predictions. The bias will depend on $e(\boldsymbol{x}, j, N)$, which is omitted. The key assumption for using the WOSI is that it carries a positive and equally strong signal of the true wage surplus for workers in routine and non-routine occupations. As with the OSI, I censor the WOSI at the 1st and 99th percentile and standardize it.

2.4 Predictions

From the above framework, I highlight two predictions:

 Expected loss: The expected loss from a negative demand shift in routine occupations increases in an incumbent's expected degree of specialization. Workers in non-routine occupations will be less affected by such a shock and their specialization matters less than for workers initially observed in routine work. The difference in the average loss between the two groups will therefore increase in specialization. I hypothesize that the loss in utility works partly through wages rather than only through amenities.

2. Expected worker flows: There is negative selection on the OSI in who leaves the routine occupations: Highly specialized workers will to a larger extent remain and tolerate the full effect of the negative shock. Again, specialization should matter less for the probability of switching to another occupation for workers initially in non-routine work; It should primarily be routine generalists that engage in occupation switches.

2.5 Econometric framework

The framework outlined above motivates a difference-in-differences-styled specification comparing the effect of the OSI on the outcomes of workers initially in routine and non-routine work. This holds constant any common effect of being specialized relative to the non-routine outside options by using non-routine workers as a comparison group to the "treated" routine workers.

Main regression model

Let $y_{ijt+\tau}$ represent an outcome in year $t + \tau$ of individual *i* observed in occupation *j* in *t*. The following model is considered the main specification:

$$y_{ijt+\tau} = \psi \widehat{\text{OSI}}(\boldsymbol{x}_{it}, j, N) + \mathbb{1} \left[j \in R \right] \left[\boldsymbol{\beta} + \boldsymbol{\phi} \, \widehat{\text{OSI}}(\boldsymbol{x}_{it}, j, N) \right] \\ + \boldsymbol{\lambda}' \boldsymbol{z}_{it} + \boldsymbol{\varepsilon}_{ijt+\tau}.$$
(8)

 β captures the difference between routine and non-routine workers at average OSI, and ψ captures the common effect of the OSI. The difference in the effect between the two occupation groups is captured by ϕ . The vector z includes year FE and the variables used to estimate the OSI, i.e., x. At times it also includes occupation FE, but then β is no longer identified. The model then mainly relies on workers with similar characteristics sorting into different occupations for identification.

Semi-parametric model

To obtain non-parametric estimates of the effect of specialization, I will also estimate models with indicators for the decile groups of the OSI distribution separately for routine and non-routine workers. The model is:

$$y_{ijt+\tau} = \sum_{q} \theta_{q}^{R} g_{q}^{R} \left(\widehat{OSI}(\boldsymbol{x}_{it}, j, N) \right) + \sum_{q} \theta_{q}^{N} g_{q}^{N} \left(\widehat{OSI}(\boldsymbol{x}_{it}, j, N) \right) + \boldsymbol{\lambda}' \boldsymbol{z}_{it} + \boldsymbol{\varepsilon}_{ijt+\tau}.$$
(9)

 g_q^R and g_q^N are indicators for belonging to OSI decile group q^R and q^N , respectively. The reference category is the lowest non-routine decile group. The θ_q^R, θ_q^N coefficients will be plotted against the average OSI within each decile group. After estimation, I add the mean of the outcome for the reference category to all θ_q^R, θ_q^N coefficients. In some exercises, I also estimate a version of (9) with only one set of decile group indicators.

Standard errors

Since the OSI is an estimate, I cannot obtain correct standard errors for the coefficients in (8) and (9). Bootstrapping is not feasible due to the laboriousness of constructing the metrics. Nevertheless, I report standard errors clustered at the individual level.

3 Data

3.1 Variables

Background characteristics

I collect data on individual characteristics from Swedish population-wide administrative registers. I use 13 different levels of educational attainment,⁵ and 25 different fields of study,⁶ according to the 2-digit categories of the Swedish Standard Classification of Education (SUN, based on ISCED). I also calculate work experience between t - 13 and t - 1 (t = year of observation) in 14 industries according to the Swedish Standard Industry Classification (SNI).⁷

Wage and occupation information

I use workers' full-time equivalent monthly wages from the Swedish Wage Structure Statistics (*Lönestrukturstatistiken*; WSS) survey. I also collect information on occupation at the two-digit level of the Swedish Classification of Occupations (SSYK, based on ISCO). A few, very small, occupations are

⁵Preschool; compulsory < 9 or 9-10 years; secondary < 2, 2, or 3 years; post-secondary < 2, 2, 3, 4 or \geq 5 years; licentiate or similar degree; doctoral degree.

⁶Basic; literacy and numeracy; personal skills; teacher training and education science; arts and media; humanities; social and behavioural science; journalism and information; business and administration; law; life science; physical science; mathematics and statistics; computing; engineering and engineering trades; manufacturing and processing; architecture and building; agriculture, forestry, fishery; veterinary; health; social services; personal services; transport services; environmental protection; security services.

⁷Agriculture and related; mining and quarrying; manufacturing; electricity, gas and water supply; construction; wholesale and retail trade; hotels and restaurants; transport, storage and communication; financial intermediation; real estate and renting; public administration; education; health care and social services; other services.

excluded. I also exclude managers and politicians. The final data include 22 occupations.⁸

Survey weights

Sampling in the WSS occurs at the firm/organization level. All public and almost 50 percent of private sector employees are sampled each year. The data include weights used to make any constructed moments representative of the full employee population.⁹ To mitigate issues with extreme weights being put on a few very small firms in certain industries, I censor the weights at the 99th percentile. I use these survey weights for all empirical exercises, including training the ANN.

Multidimensional skills

I utilize information on cognitive and non-cognitive skills from the Swedish War Archive. These data were collected during the Swedish draft process in 1969-1994. They are available for around 90 percent of males born in 1951-1976 who underwent the draft at the age of 18 or 19.¹⁰ The draftees performed four standardized cognitive tests on: Inductive reasoning; verbal comprehension; spatial ability, and; technical understanding. They also took part in a 25 minute interview with a psychologist. The psychologist evaluated the profile of the draftee and scored them along four dimensions. Mood et al. (2012) interprets these as: Social maturity; psychological energy, focus or perseverance; intensity or activation without pressure; emotional stability or tolerance to stress. The detailed scores were used by the military to produce two aggregate measures of cognitive and non-cognitive ability. I utilize these aggregate metrics in some descriptive exercises. The scaling of the scores varies by test type and draft cohorts. I standardize the skill measures within each cohort following Fredriksson et al. (2018) and Edin et al. (2022b).

Variables used to estimate the OSI

To estimate the OSI, I incorporate a second-order polynomial for each of the eight cognitive and non-cognitive abilities and for experience in the 14 industries in \mathbf{x} .¹¹ I also include 28 age, 13 education level, 25 education field, and 21 region of residence indicators. These variables proxy abilities at labor market entry, human capital acquired through education, experience and age, and differences in, i.a., occupation demand across local labor markets.

⁸Occupation and full-time wages refer to a reference week in September for the private sector and November for the public sector.

⁹All firms with at least 500 employees as well as the whole public sector are sampled. In smaller firms, the sampling probability is positively related to size and stratified by industry.

¹⁰These skill measures are described in detail by Lindqvist and Vestman (2011).

¹¹Individual occupation histories are not observable as occupation information is from a survey.

Occupation routine task intensity

To distinguish between routine and non-routine occupations, I use the routine task intensity (RTI) index from Autor and Dorn (2013). It is based on the five measures of task requirements in 1980 from the U.S. dictionary of occupation titles (DOT) used by Autor et al. (2003): Eye-hand-foot coordination (classified as manual); set limits, tolerances and standards (routine cognitive) and finger dexterity (routine manual), the average of which is routine task requirement; direction control and planning, and GED math, the average of which is abstract task requirement. The RTI for occupation *k* is:

 $\operatorname{RTI}_{k} = \ln(\operatorname{routine input}_{k}) - \ln(\operatorname{manual input}_{k}) - \ln(\operatorname{abstract input}_{k}).$ (10)

I classify occupations as either routine ($r_k = 1$) or non-routine ($r_k = 0$) based on the RTI_k relative to the median M at the worker level: $r_k = \mathbb{1} [\text{RTI}_k \ge M]$.

3.2 Sample

My sample for which I aim to predict the OSI consists of approximately 1.7 million observations of male employees observed in 1997-2001, who are sampled in the WSS, for which information on skills is available and who are therefore aged 23-50. 1997-2001 can be thought of as a pre-period. 2001 is chosen as the final year somewhat arbitrarily. I need a sufficiently large sample to train the Neural Network. Moreover, aggregate statistics on occupation employment are published by Statistics Sweden from this year forward.

I draw a random 30 percent sample of individuals (and not observations) from the pre-period sample. This subsample is used as training data for the ANN, multinomial logit and wage regression underlying the WOSI. The remaining 70 percent are used for out-of-sample evaluation and the empirical analysis in Section 4.

For the empirical analysis, I follow the individual associated with each observation in the pre-period up to twelve years forward in time and collect information on wages, annual earnings, employment and occupation.¹² To obtain better coverage, future occupation and log wage from the WSS are measured in t + 10 through t + 12. I use the most recent observation if available or move back otherwise. As survey weights, I use the inverse of the probability of being observed in t and at least in one year between t + 10 and t + 12.¹³ To account for differences in when workers are observed, I always control for initial \times future year of observation FE.

weight_t × 1/
$$\left(1 - \left(1 - \frac{1}{\text{weight}_{t+\tau}}\right)^3\right)$$
.

¹²In one exercise, I also follow workers five years back in time from the pre-period year of observation and collect information on previous wages.

¹³More precisely, I use the observed initial weight_t and future weight_{t+ τ} and calculate the inverse of the probability of being observed in both t and at least one year between t + 10 and t + 12 as

4 Empirical analysis

This section reports the empirical results. Section 4.1 presents validation exercises. Section 4.3 describes how different characteristics relate to the OSI. Section 4.3 reports the results on long-run worker outcomes.

4.1 Validation exercises

I begin by discussing, e.g., how well the ANN can predict occupation choices. I then describe how the OSI is related to alternative metrics of specialization and to log wage. Finally, I show that the routine indicator predicts well which occupations declined during the studied period.

Explanatory power of the choice and wage models

The neural network has an out-of-sample accuracy, i.e., assigns the highest probability to the actual occupation choice, of 47.9 percent. This is a slight advantage over the accuracy of the multinomial logit, 43.5 percent. These numbers can be compared to when including only a constant in the vector \mathbf{x} . The guess would then be the largest occupation in the training data set (physical and engineering associate professionals), with an accuracy of 11.3 percent.

The ANN acuracy does not matter per se for the OSI. The critical aspect is instead the validity of the probabilities it assigns to *all* potential choices. Figure C3 in the appendix shows that the probabilities assigned by the ANN on average correspond well to the choices of workers in the evaluation sample. In contrast, the multinomial logit predictions slightly overestimate the true probabilities.

The wage regression underlying the WOSI in (7) has a coefficient of determination of 0.56 out-of-sample; occupation FE interacted with age, skills, education, region, and previous experience can explain around half of the variation in log wage (after residualization by year).¹⁴

The alternative specialization indices and wages

Next, I relate the OSI to the metric based on the multinomial logit, the WOSI, and workers' log wages.¹⁵ Figure 1 reports non-parametric estimates from the version of the model from (9) with a common set of decile group indicators. I control for occupation and year FE as well as all variables in x used to construct the OSI (i.e., abilities, industry experience, education level and field, region, and age).

According to panel (a), there is a strong relationship between the two probability based metrics. Panel (b) reveals a positive relationship between the OSI

¹⁴The accuracy and coefficient of determination are very similar for the training and test sample, indicating that the models are not prone to overfitting.

¹⁵Figure C2 reports histograms of these metrics.



Figure 1. Non-parametric relationship between the OSI, alternative specialization metrics and log wage

Notes: The figure relates the OSI based on the neural network to the OSI using the multinomial logit estimates (Panel a), the WOSI which measures the expected wage difference between workers' current and outside options (panel b) and log wage (panel c). The vertical axis reports the decile group coefficients from the version of equation (9) with a common set of indicators. The regressions include controls for a second-order polynomial in the eight skill dimensions and 14 previous industry experience variables as well as age, education and field, region, year, and occupation FE. To all decile group coefficients, I add the average outcome of the lowest decile group reference category. The horizontal axis plots the average OSI within each decile group.

and WOSI. This means that, when a worker's expected wage surplus in an occupation is high, so is his propensity to work there. A one-standard deviation (sd) increase in the OSI is associated with a 0.3-sd higher WOSI. There is also a positive relationship with log wage: A sd increase in the OSI results in 3 percent higher wages.¹⁶ Importantly, this implies that the wage loss associated with leaving one's occupation is increasing in the OSI.

Destinations of occupation switchers

The model assumes that worker characteristics are intrinsically linked to occupation specific utility and therefore occupation choices. One way to corroborate this is to study the choices of workers that transition from their original occupation. Figure C4 shows that the assigned probabilities are highly informative about which outside occupations are the most likely destinations of long-run switchers.

Relationship between routine task intensity and employment growth

Figure 2 ranks all occupations according to their routine task intensity. This rank is then plotted against growth in the employment share between 2001 and 2013 according to Statistics Sweden. The vertical line shows the routine/non-routine cutoff. The occupations that saw the lowest employment growth are routine-intensive. In fact, no routine occupation experienced an increasing employment share and only one non-routine occupation (teaching profession-als)¹⁷ saw declining relative employment.

The figure also plots the employment share growth by occupation for my sample of workers for which I observe occupation in both t and t + 10 to t + 12. The results are quite similar. With the exception of high-skilled physical and engineering professionals and stationary plant operators, all routine occupations that experienced negative employment growth also saw size decreases in my sample of incumbent workers. Overall, routine decline cannot be accounted for only by labor market entrants and leavers.

Education requirements of routine and non-routine occupations

In SSYK, occupations with the leading number 1-3, 4-8, and 9 can be considered high-, medium- and low-skilled, respectively. As evident from Figure 2, with only one exception, routine occupations are low- to middle-skilled. To be able to better compare these to non-routine occupations, I focus on workers initially observed in low- to middle-skilled occupations in Sections 4.2 and 4.3. These are listed in Table 1. Apart from clerks and manual labourers, the routine occupations are concentrated in manufacturing. The non-routine occu-

¹⁶Thus, in line with, e.g., Fredriksson et al. (2018), being well matched (i.e., likely to sort into the chosen occupation) is also associated with an absolute premium.

¹⁷However, associate teaching professionals exhibit a substantial increase in size, suggesting that workers' occupations may have been reclassified.



Figure 2. Occupation employment share growth by routine-intensity rank

Notes: The figure plots the employment share growth at the occupation level on the vertical axis. On the horizontal axis, the occupations are ranked by their routine task intensity (from low to high). The vertical dashed line represents the cutoff between routine and non-routine occupations. "Aggregate growth" refers to relative employment growth between 2001 and 2013 according to Statistics Sweden. "Growth in incumbent sample" instead refers to my sample of workers for which I observe occupation in both *t* (1997-2001) and *t* + 10 to *t* + 12 (depending on when workers are observed in the WSS). The size of the markers is determined by initial employment share.
pations are instead mainly found in construction, services, transportation, and agriculture.¹⁸

4.2 Characteristics associated with specialization

I now briefly describe which characteristics are associated with specialization. Figures C5 and C6 relates the OSI to the variables that are used to construct the OSI for workers in low- to middle-skilled occupations.

For routine occupations, both cognitive and non-cognitive ability is strongly negatively related to specialization. The story for educational attainment is similar. The education fields associated with the highest specialization are general (which includes primary) education and, unsurprisingly, engineering, manufacturing and construction. Regarding regions, average OSI is relatively low in the Stockholm region and other densely populated areas. Specialization is also relatively low in the least populated regions. Specialization is decreasing slightly in age. At the same time, experience in manufacturing is one of the strongest predictors of specialization for routine workers.

For non-routine workers, the ability relationships are not as stark as for routine occupations. But the relationship with education level and age are more pronounced. Specialization is also high for workers with an education related to services, teaching, and construction (classified together with engineering and manufacturing). The results for region of residence are similar as for routine workers. Finally, workers with experience in construction, transportation, public administration, education and health care all exhibit high specialization levels.

4.3 Specialization and long-run career outcomes

How does the OSI relate to the future career outcomes of workers in routine and non-routine occupations? From the theory, I expect routine workers to be more likely to switch to a non-routine outside option than non-routine workers. This difference should be caused primarily by low-OSI workers. Moreover, I expect average wage/earnings growth to be lower for workers in routine than non-routine occupations. This growth penalty should be increasing in the OSI.

Two occupation outcomes are analyzed: Making any occupation transition between t and t + 10 to t + 12, and transitioning to a non-routine outside option. I then analyze three additional outcomes: log wage growth between t and t + 10 to t + 12, annual earnings growth in t to t + 12 relative to the initial level, i.e., $\Delta_{t,t+12}$ earnings/earnings_t, and an employment indicator for t + 12. Earnings and wages are adjusted for CPI.

¹⁸Higher-skilled occupations still enter as outside occupations when calculating the OSI. In robustness checks, I show that the results are similar when including all occupations.

	Routine		Non-routine			
SSYK	Name	Ν	SSYK	Name	N	
74	Other craft & related trades	3,098	61	Skilled agriculture work	9,364	
42	Customer service clerks	4,825	52	Sales & demo.	15,596	
73	Precision, handicraft & printing	5,103	91	Sales & services ele- mentary occupations	22,010	
93	Manual labourers	29,411	83	Drivers & mobile-plant operators	52,904	
81	Stationary-plant & re- lated operators	56,923	51	Service, care & protec- tive work	77,605	
41	Office clerks	66,488	71	Extraction & building	91,859	
72	Metal, machinery & re- lated	101,206				
82	Machine operators & assemblers	116,862				

Table 1. Low- to middle-skilled routine and non-routine occupations

Notes: The table reports the number of observations in the pre-period test sample for all lowto middle-skilled occupations (excluding the high-skilled categories 1-3 at the broadest level of SSYK) separately by occupations below and above median routine intensity.

Non-parametric estimates

Figure 3 reports non-parametric estimates from equation (9) for the OSI and the switching outcomes. Panel (a) shows that leaving the initial occupation is strongly negatively related to specialization for both routine and non-routine occupations. Results are similar when only considering switches to other nonroutine occupations in panel (b). Almost half of the least specialized workers in routine occupations left for a non-routine occupation. This is around three times the mobility rate of the highest-OSI workers. In (a), the relationship is somewhat stronger for routine compared to non-routine workers. But this is to a lesser extent the case in (b). Thus, the OSI indeed predicts which workers leave the set of routine occupations. My interpretation of this is that workers were typically not forced out of routine work; Highly specialized routine were largely able to remain in their initial occupations. However, the hypothesis that primarily routine (and not non-routine) generalists are engaged in occupation switches is not consistently borne out.

Figure 4 instead reports results for wage growth, earnings growth, and employment. Specialization is negatively related to wage and earnings growth. These relationships are markedly stronger for workers in routine compared to non-routine work. At the bottom of the OSI distribution, there is no discernible difference in wage and earnings growth between the two occupation groups. At the top of the distribution, the difference in wage (earnings) growth is around five log points (six percentage points). These results indicate that routine specialists experienced substantial negative consequences of the demand shift.¹⁹ By contrast, routine workers with the lowest OSI levels appear to have been largely shielded from it. There is a slight positive relationship between the OSI and future employment for both occupation groups. This relationship appears marginally stronger for non-routine workers, but the difference in the effect is never significant. The absence of any employment effect suggests that the quality of workers' future jobs is the important margin for future outcomes.

Wage growth can be analyzed separately for occupation stayers and switchers. However, specialization influences the decision to switch. Selection is therefore a key issue to bear in mind. Figure C7 reports non-parametric estimates for log wage growth separately for workers that remained in the same occupation and moved to an outside non-routine occupation. The relationship between the OSI and wage growth is similar for routine and non-routine switchers. However, specialization is more negatively related to wage growth for routine compared to non-routine stayers. This may be due to selection: If the wage premium in routine work decreases, generalists with good outside options should stay only if there is some counteracting idiosyncratic effect in the future that leads to acceptable wage growth. A more troubling explana-

¹⁹This is consistent with the findings for workers who stay in routine occupations in Cortes (2016).

Figure 3. Non-parametric relationship between specialization and leaving the initial occupation by routine and non-routine occupations



(a) Switch to another occupation

Notes: The vertical axis reports the decile group coefficients from equation (9) for separate occupation switching outcomes. All regressions include controls for a second-order polynomial in the eight skill dimensions and 14 previous industry experience variables as well as age, education and field, region, and initial \times future year fixed effects. To all decile group coefficients, I add the average outcome of the lowest OSI non-routine reference category. The horizontal axis plots the average OSI within each decile group. Panel (a) relates the OSI to the probability of being observed in a different occupations in *t* and *t* + 10 to *t* + 12 separately for workers initially observed in routine and non-routine occupations. Panel (b) instead reports the relationship between specialization and switching to another, non-routine occupation.

Figure 4. Non-parametric relationship between specialization and labor market outcomes by routine and non-routine occupations



Notes: The figure relates the OSI to log wage growth between t and t + 10 to t + 12, growth in annual earnings between t and t + 12 divided by initial earnings, and the probability of being classified as employed in t + 12. See Figure 3 for additional information.

tion is if the OSI is related differently to career wage profiles for routine and non-routine occupations. To account for this, I control for age and industry experience fully interacted with the routine indicator in subsequent robustness checks.

Alternative specifications, samples and specialization metrics

Next, I explore the robustness of my results to the choice of specification and specialization metric. To this end, I estimate different versions of the main regression model from equation (8).

Table 2 reports results for the main OSI metric. The first model uses the same controls as in Figures 3-4: Year FE and all variables used to estimate the OSI. The second model additionally controls for occupation FE, thus relying on variation within each occupation. The third model adds a secondorder polynomial in initial log wage fully interacted with the routine indicator. This allows for comparing workers with similar absolute productivity yet different relative occupation advantage. The fourth model includes workers in higher-skilled occupations. In all models, the OSI is negatively associated with occupation switching and earnings and wage growth. The effect is consistently more negative for workers in routine than non-routine work, as captured by the interaction. In the first model, a standard deviation increase in the OSI is associated with around 1.4 percent lower wage growth and 2 percentage points lower earnings growth for routine relative to non-routine workers. These effects decrease slightly when controlling for occupation FE, but remain statistically significant. Moreover, there are no significant effects on future employment.

Table 3 reports results for alternative specialization metrics. All models include controls for the variables used to estimate the OSI and year and occupation FE. The first model uses the OSI based on the multinomial logit. It exhibits very similar results to the comparable model in Table 2. The WOSI used in the second model yields qualitatively similar results, but the effect sizes are more modest. The third model uses the percentile groups of the OSI distribution within each occupation instead of the OSI. I standardize the metric for ease of comparison. This corroborates that the results are not caused by how the OSI scales across occupations. Finally, the metric in the fourth model includes all outside occupations when calculating the OSI. This implies summing over $n \in K \setminus \{j\}$ in equation (5). The effects are much starker than for the baseline OSI. However, this is because a subset of the routine occupations are very similar and could be classified as the same occupations. This leads to much less variation in specialization for routine workers.

Interacted controls

Figure C8 reports estimates from models where subsets of the variables used to estimate the OSI (i.e., x) are fully interacted with the routine indicator. The subsets are: All skill measures; Region of residence FE; Education level and

	Switch to a non-routine occ.	Growth in log wage	Growth in rel. earnings	Future employment				
	(1)	(2)	(3)	(4)				
(a) Controls for a	ll characteristics							
Routine	0.0483***	-0.0271***	-0.0441***	-0.00170*				
	(0.00190)	(0.000808)	(0.00200)	(0.000919)				
OSI	-0.0681***	-0.00419***	-0.00490***	0.00287***				
	(0.00133)	(0.000568)	(0.00132)	(0.000605)				
$\text{OSI} \times \text{Routine}$	-0.0124***	-0.0144***	-0.0214***	-0.00126				
	(0.00180)	(0.000766)	(0.00191)	(0.000878)				
N	446,921	446,921	639,007	639,896				
R ²	0.078	0.119	0.054	0.036				
(b) Additional con	(b) Additional controls for occupation							
OSI	-0.0451***	-0.00257***	-0.00404***	0.00115*				
	(0.00143)	(0.000612)	(0.00143)	(0.000655)				
$\text{OSI} \times \text{Routine}$	-0.0138***	-0.0108***	-0.0189***	-0.000819				
	(0.00187)	(0.000801)	(0.00200)	(0.000916)				
$N_{\mathbf{P}^2}$	446,921	446,921	639,007	639,896				
(c) Additional con	K 0.095 0.127 0.050 0.057 (c) Additional controls for occupation and initial log wage							
OSI	-0.0435***	0.00929***	0.000219	0.000494				
	(0.00143)	(0.000530)	(0.00143)	(0.000656)				
$\text{OSI} \times \text{Routine}$	-0.0135***	-0.0122***	-0.0187***	-0.000845				
	(0.00187)	(0.000692)	(0.00199)	(0.000915)				
N	446,921	446,921	639,007	639,896				
R ²	0.097	0.349	0.061	0.039				
(d) Additional controls for occupation, including all occupations								
OSI	-0.0864***	-0.0138***	-0.0159***	0.00293***				
	(0.000856)	(0.000380)	(0.000895)	(0.000377)				
$OSI \times Routine$	-0.00219	-0.00687***	-0.0142***	-0.00172***				
	(0.00144)	(0.000637)	(0.00150)	(0.000636)				
N	831,991	831,991	1,149,195	1,172,526				
R ²	0.100	0.205	0.079	0.034				

Table 2. Regressions using the main specialization index with varying controls and samples

Notes: The table reports results from estimating equation (8) for different outcomes, sets of controls, and samples. All regressions include controls for a second-order polynomial in the eight skill dimensions and 14 previous industry experience variables as well as age, education and field, region, and initial \times future year fixed effects. Models (b)-(d) additionally include occupation FE. Model (c) includes controls for a second-order polynomial in initial log wage fully interacted with the indicator for routine occupations. Model (d) is estimated using observations of workers in all, and not only low- to middle-skilled, occupations. Standard errors clustered at the individual level are in parenthesis. *, ** and *** represent statistical significance at the 10-, 5- and 1-percent level, respectively.

	Switch to a non-routine occ.	Growth in log wage	Growth in rel. earnings	Future employment					
	(1)	(2)	(3)	(4)					
(a) Specialization based on multinomial logit									
OSI	-0.0376*** (0.00178)	-0.00205*** (0.000763)	0.00224 (0.00179)	0.00180** (0.000819)					
$OSI \times Routine$	-0.0278*** (0.00220)	-0.0126*** (0.000942)	-0.0289*** (0.00233)	-0.00128 (0.00107)					
N R ²	446,915 0.094	446,915 0.127	639,001 0.056	639,890 0.037					
(b) Specialization based on predicted log wage (WOSI)									
OSI	-0.0129*** (0.00153)	-0.00908*** (0.000652)	0.00179 (0.00158)	0.000769 (0.000727)					
$\text{OSI} \times \text{Routine}$	-0.00912*** (0.00175)	-0.00223*** (0.000750)	-0.00777*** (0.00185)	0.00143* (0.000848)					
Ν	446,921	446,921	639,007	639,896					
R ²	0.092	0.127	0.056	0.037					
(c) Specialization	based on OSI percen	tiles within occup	ation						
OSI	-0.0500*** (0.00125)	-0.000825 (0.000535)	-0.00418*** (0.00125)	0.000690 (0.000574)					
OSI imes Routine	-0.00134 (0.00145)	-0.00829*** (0.000622)	-0.0115*** (0.00153)	-0.00137* (0.000702)					
Ν	446,921	446,921	639,007	639,896					
R ²	0.097	0.127	0.056	0.037					
(d) Including routine outside options in OSI									
OSI	-0.0774*** (0.00176)	-0.00243*** (0.000754)	-0.00230 (0.00181)	0.00285*** (0.000829)					
$OSI \times Routine$	-0.0769*** (0.00568)	-0.0531*** (0.00243)	-0.0304*** (0.00617)	0.0135*** (0.00283)					
N	446,919	446,919	639,005	639,894					
R ²	0.097	0.128	0.056	0.037					

 Table 3. Regressions using alternative specialization indices

Notes: The table reports results from estimating equation (8) for different outcomes and specialization metrics. All regressions include controls for a second-order polynomial in the eight skill dimensions and 14 previous industry experience variables as well as age, education and field, region, initial \times future year, and occupation fixed effects. Model (a) uses the OSI metric based on the multinomial logit. (b) uses the WOSI based on predicted wage surplus. (c) uses as specialization the within-occupation percentile groups of the OSI. This metric is standardized to mean zero, standard deviation one for ease of comparison. (d) uses a version of the OSI based on all (and not only non-routine) outside options. See Table 2 for additional information.

field FE; Age FE and industry experience. This exercise reveals if the estimated effect of the OSI is caused by any subset of x. Controlling for age and experience also addresses concerns about whether differences in wage growth is due to differing career wage profiles. Most coefficients are highly similar to the second model in Table 2. The exception is when controlling for interacted age and experience. The interaction effect on wage growth then increases from -0.01 to around -0.005. Thus, age and experience appears to contribute disproportionately to this effect. But the other characteristics are still important. Moreover, the interaction effect on earnings growth is resilient to this set of controls.

Occupation-specific effects

To better understand what role specialization plays in each occupation, I estimate occupation-specific effects of the OSI on outcomes. These are reported in Figure C9. Without exception, the probability of switching to another nonroutine occupation declines with specialization. Furthermore, the OSI is typically negatively associated with both wage and earnings growth. This is especially true for routine occupations, in line with the previous results.

Trends in the returns to specialization

When in workers' careers does the effect of specialization on wage growth arise? To study this, I estimate equation (8) separately for log wage level in $t + \tau$ for all $\tau \in \{-5, ..., 12\}$ where *t* is the year when the OSI and occupation choice is measured. Figure 5 plots the main coefficients against τ . Routine workers enjoyed a high initial wage premium, which was increasing until *t*. This coefficient then decreases from 0.04 to 0.02 in t + 12. There is a positive relationship between specialization and wages: For all relative years, the common effect of the OSI is positive. On average, the estimate is around 0.02 and increases marginally over time. The additional OSI premium of routine relative to non-routine workers is positive at first, but declines to around -0.015 in t + 12. Thus, being specialized in routine occupations used to yield a higher return. Over time, this effect turns into a relative penalty for workers initially observed in routine work.

5 Conclusions

In this paper, I create and estimate an index of worker occupation specialization (OSI) using detailed individual characteristics and machine learning tools. The OSI is derived from a Roy (1951)-styled discrete choice model. Theoretically, it measures the expected difference between a worker's utility in his occupation and his best outside option. The index is simply a monotone transformation of the *ex ante* propensity for working in an outside occupation. This determines the worker's utility loss from a negative wage premium *Figure 5.* Estimated effect of occupation specialization on log wage by routine and non-routine occupations and year relative to year of observation



Notes: The figure reports estimates and 95-percent confidence intervals for the routine indicator, the OSI, and the interaction between the two from equation (8) estimated on log wage in different years relative to the year of observation. I follow individuals five years back and 12 years forward in time and estimate a separate model for each time horizon. All regressions include controls for a second-order polynomial in the eight skill dimensions and 14 previous industry experience variables as well as age, education and field, region, and year fixed effects. Standard errors are clustered at the individual level.

shock: Low-OSI workers with attractive non-shocked options are able to alleviate losses by moving. High-OSI specialists instead willingly remain and experience the full consequences of the shock.

The paper then analyzes to what extent the OSI can explain the consequences for incumbent employees of the falling employment in routine occupations, likely caused by shifting demand, during the period 1997–2013. I find that routine and non-routine generalists with low levels of OSI were highly mobile and did approximately equally well in terms of future earnings growth. Routine specialists instead by and large remained in routine work despite the overall employment decline in these occupations. They also experienced significantly lower earnings growth than both generalists and nonroutine specialists.

These findings are broadly consistent with the predictions from the Roystyle discrete choice model from which the specialization index is derived. Overall, the results indicate that the Roy (1951) model can characterize which workers lose from negative demand shifts. Moreover, the observable worker attributes at hand can be used to infer how dependent workers are on their current occupations.

Exploring the consequences of the historical decline in routine work is important in its own right. But the ability of the OSI to predict which workers experienced negative consequences from this shift also substantiates the general usefulness of the index for describing worker susceptibility. Currently, policy makers have very few tools at hand for forecasting individual consequences of future shifts in occupation demand. The OSI is solely based on current information. It could therefore be used to characterize workers employed in occupations today that we believe will experience negative demand shifts in the future.

References

- Adermon, A. and M. Gustavsson (2015). "Job polarization and task-biased technological change: Evidence from Sweden, 1975–2005". *The Scandinavian Journal of Economics* 117.3, 878–917.
- Arntz, M., T. Gregory, and U. Zierahn (2017). "Revisiting the risk of automation". *Economics Letters* 159, 157–160.
- Autor, D. and D. Dorn (2013). "The growth of low-skill service jobs and the polarization of the US labor market". *American Economic Review* 103.5, 1553–97.
- Autor, D., L. Katz, and M. Kearney (2008). "Trends in U.S. Wage Inequality: Revising the Revisionists". *The Review of Economics and Statistics* 90.2, 300–323.
- Autor, D., F. Levy, and R. Murnane (2003). "The Skill Content of Recent Technological Change: An Empirical Exploration*". *The Quarterly Journal* of Economics 118.4, 1279–1333.
- Bachmann, R., M. Cim, and C. Green (2019). "Long-Run Patterns of Labour Market Polarization: Evidence from German Micro Data". *British Journal* of Industrial Relations 57.2, 350–376.
- Balassa, B. (1965). "Trade liberalisation and "revealed" comparative advantage". *The manchester school* 33.2, 99–123.
- Böhm, M. (2020). "The price of polarization: Estimating task prices under routine-biased technical change". *Quantitative Economics* 11.2, 761–799.
- Böhm, M., H.-M. v. Gaudecker, and F. Schran (2023). "Occupation Growth, Skill Prices, and Wage Inequality". *Journal of Labor Economics*, forthcoming.
- Brynjolfsson, E. and A. McAfee (2014). *The second machine age: Work, progress, and prosperity in a time of brilliant technologies.* WW Norton & Company.
- Caldwell, S. and O. Danieli (2022). "Outside options in the labor market". *Mimeo*.
- Cavaglia, C. and B. Etheridge (2020). "Job polarization and the declining quality of knowledge workers: Evidence from the UK and Germany". *Labour Economics* 66, 101884.
- Coraggio, L., M. Pagano, A. Scognamiglio, and J. Tåg (2022). "JAQ of All Trades: Job Mismatch and Firm Productivity". *Mimeo*.
- Cortes, G. M. (2016). "Where have the middle-wage workers gone? A study of polarization using panel data". *Journal of Labor Economics* 34.1, 63–105.
- Cortes, G. M. and G. Gallipoli (2018). "The costs of occupational mobility: An aggregate analysis". *Journal of the European Economic Association* 16.2, 275–315.

- Cortes, G. M., N. Jaimovich, C. J. Nekarda, and H. E. Siu (2020). "The dynamics of disappearing routine jobs: A flows approach". *Labour Economics* 65, 101823.
- Cortes, G. M., N. Jaimovich, and H. E. Siu (2017). "Disappearing routine jobs: Who, how, and why?" *Journal of Monetary Economics* 91, 69–87.
- De Jong, G., A. Daly, M. Pieters, and T. Van der Hoorn (2007). "The logsum as an evaluation measure: Review of the literature and new results". *Transportation Research Part A: Policy and Practice* 41.9, 874–889.
- Deming, D. (2017). "The growing importance of social skills in the labor market". *The Quarterly Journal of Economics* 132.4, 1593–1640.
- Edin, P.-A., T. Evans, G. Graetz, S. Hernnäs, and G. Michaels (2022a). "Individual consequences of occupational decline". *Mimeo*.
- Edin, P.-A., P. Fredriksson, M. Nybom, and B. Öckert (2022b). "The Rising Return to Noncognitive Skill". American Economic Journal: Applied Economics 14.2, 78–100.
- Eggenberger, C., S. Janssen, and U. Backes-Gellner (2022). "The Value of Specific Skills under Shock: High Risks and High Returns". *Labour Economics*, 102187.
- Fredriksson, P., L. Hensvik, and O. N. Skans (2018). "Mismatch of talent: Evidence on match quality, entry wages, and job mobility". *American Economic Review* 108.11, 3303–38.
- Frey, C. B. and M. A. Osborne (2017). "The future of employment: How susceptible are jobs to computerisation?" *Technological Forecasting and Social Change* 114, 254–280.
- Gathmann, C. and U. Schönberg (2010). "How general is human capital? A task-based approach". *Journal of Labor Economics* 28.1, 1–49.
- Goos, M., M. Arntz, U. Zierahn, T. Gregory, S. C. Gomez, I. G. Vázquez, and K. Jonkers (2019). "The impact of technological innovation on the future of work". Tech. rep. JRC Working Papers Series on Labour, Education and Technology.
- Goos, M. and A. Manning (2007). "Lousy and lovely jobs: The rising polarization of work in Britain". *The Review of Economics and Statistics* 89.1, 118–133.
- Goos, M., A. Manning, and A. Salomons (2014). "Explaining job polarization: Routine-biased technological change and offshoring". *American Economic Review* 104.8, 2509–26.
- Groes, F., P. Kircher, and I. Manovskii (2015). "The U-shapes of occupational mobility". *The Review of Economic Studies* 82.2, 659–692.

- Guvenen, F., B. Kuruscu, S. Tanaka, and D. Wiczer (2020). "Multidimensional skill mismatch". *American Economic Journal: Macroeconomics* 12.1, 210–44.
- Hanemann, W. M. (1984). "Discrete/continuous models of consumer demand". *Econometrica: Journal of the Econometric Society*, 541–561.
- Jaimovich, N., I. Saporta-Eksten, H. Siu, and Y. Yedid-Levi (2021). "The macroeconomics of automation: Data, theory, and policy analysis". *Journal of Monetary Economics* 122, 1–16.
- Jovanovic, B. and Y. Nyarko (1997). "Stepping-stone mobility". *Carnegie-Rochester Conference Series on Public Policy*. Vol. 46. Elsevier, 289–325.
- Kingma, D. P. and J. Ba (2014). "Adam: A method for stochastic optimization". *arXiv preprint arXiv:1412.6980*.
- Lindqvist, E. and R. Vestman (2011). "The labor market returns to cognitive and noncognitive ability: Evidence from the Swedish enlistment". *American Economic Journal: Applied Economics* 3.1, 101–28.
- Lise, J. and F. Postel-Vinay (2020). "Multidimensional skills, sorting, and human capital accumulation". *American Economic Review* 110.8, 2328–76.
- McFadden, D. (1973). "Conditional logit analysis of qualitative choice behavior".
- Mokyr, J., C. Vickers, and N. L. Ziebarth (2015). "The history of technological anxiety and the future of economic growth: Is this time different?" *Journal of Economic Perspectives* 29.3, 31–50.
- Mood, C., J. O. Jonsson, and E. Bihagen (2012). "Socioeconomic persistence across generations : cognitive and noncognitive processes". *From parents to children: the intergenerational transmission of advantage*, 53–84.
- OECD (2019). *OECD employment outlook 2019: The future of work*. Organisation for Economic Co-operation and Development OECD.
- Robinson, C. (2018). "Occupational mobility, occupation distance, and specific human capital". *Journal of Human Resources* 53.2, 513–551.
- Ross, S. L. and P. Ukil (2021). "Initial Industry and Long-Term Earnings Growth". *AEA Papers and Proceedings* 111, 476–80.
- Roy, A. D. (1951). "Some thoughts on the distribution of earnings". Oxford *Economic Papers* 3.2, 135–146.
- Roys, N. A. and C. R. Taber (2022). "Skill prices, occupations, and changes in the wage structure for low skilled men". *Mimeo*.
- Small, K. A. and H. S. Rosen (1981). "Applied welfare economics with discrete choice models". *Econometrica: Journal of the Econometric Society*, 105–130.

- Traiberman, S. (2019). "Occupations and import competition: Evidence from Denmark". *American Economic Review* 109.12, 4260–4301.
- Train, K. E. (2009). *Discrete choice methods with simulation*. Cambridge university press.
- Webb, M. (2019). "The impact of artificial intelligence on the labor market". *Mimeo*.

Appendix A Derivations

This section shows the full derivation of the results discussed in Section 2.1 and 2.2. I begin by discussing some useful properties of the Gumbel distribution. I then characterize the distribution of the utility surplus in equation (2). Using these results, I proceed to derive the occupation specialization index. Finally, I derive a closed-form expression for the expected value of the utility loss following the wage premium in equation (2).

A.1 Properties of the Gumbel distribution

All ε_{ik} are assumed to be IID standard extreme value type I (or Gumbel) distributed. The occupation choice probabilities then follow the multinomial logit formula expressed in equation (3).

Moreover, the maximum utility of a worker (see equation (1)) before conditioning on occupation choice is distributed as:

$$\max_{k \in K} u_{ik} \sim \operatorname{Gumbel}\left(\ln\left(\sum_{k \in K} e^{u_k(\boldsymbol{x})}\right), 1\right).$$
(A1)

The location parameter in this equation is commonly referred to as the logsum. The expected utility takes the form $\mathbb{E}\left[\max_{k \in K} u_{ik}\right] = \ln\left(\sum_{k \in K} e^{u_k(\mathbf{x})}\right) + C$, where *C* is an unknown constant reflecting that absolute utility cannot be measured. See, e.g., Small and Rosen (1981), Train (2009) for a textbook treatment, and De Jong et al. (2007) for a literature review of applications of this expected value.

Using results from Hanemann (1984),²⁰ the maximum utility can be shown to be independent of the chosen occupation j. More specifically, Hanemann (1984) shows that the idiosyncratic term of the best choice j is distributed as:

$$\varepsilon_{ij} \mid \underset{k \in K}{\operatorname{arg\,max}} u_{ik} = j \sim \operatorname{Gumbel}\left(\ln\left(\sum_{k \in K} e^{u_k(\boldsymbol{x})} / e^{u_j(\boldsymbol{x})}\right), 1\right).$$
(A2)

Next, let F(.) be the CDF of this distribution and *s* be some value. Then:

$$\Pr(u_{ij} < s) = F(s - u_j(\mathbf{x}))$$

$$= \exp\left\{-\exp\left\{-\exp\left\{-(s - u_j(\mathbf{x})) + \ln\left(\sum_{k \in K} e^{u_k(\mathbf{x})} / e^{u_j(\mathbf{x})}\right)\right\}\right\}\right\}$$

$$= \exp\left\{-\exp\left\{-s + \ln\left(\sum_{k \in K} e^{u_k(\mathbf{x})}\right)\right\}\right\}.$$
(A3)

²⁰His framework focuses on consumption and incorporates heterogeneous goods prices and a budget constraint.

This is the CDF associated with the distribution in (A1). Thus, the maximum utility conditional on any optimal choice j is distributed the same as the unconditional maximum in (A1).

A final useful property of the Gumbel distribution is that the difference between two independent Gumbel distributed variables with location parameters a, b and common scale parameter c is known to follow a Logistic(a - b, c) distribution.

A.2 Characterizing the distribution of the utility surplus

Finding the distribution of the utility surplus in (2) is done in four steps.

Step 1: Rewriting the utility surplus

I begin by defining P(s) as the probability that the utility surplus in (2) conditional on worker characteristics \mathbf{x} and occupation choice j is smaller than the value s. This represents the CDF of the utility surplus for which I want to obtain an closed-form expression:

$$P(s) \equiv \Pr\left(u_{ij} - \max_{n \in N \setminus \{j\}} u_{in} \le s \mid \boldsymbol{x}, \ \arg\max_{k \in K} u_{ik} = j\right).$$
(A4)

Next, K can be partitioned into two subsets: One including R and a worker's chosen occupation j, and one with his non-routine outside options. The probability may then be rewritten in terms of the difference between the maximum utility in each set conditional on the first maximum being larger, and on the choice in the first set being j:

$$P(s) = \Pr\left(\max_{r \in \mathbb{R} \cup \{j\}} u_{ir} - \max_{n \in \mathbb{N} \setminus \{j\}} u_{in} \le s \mid \mathbf{x}, \max_{r \in \mathbb{R} \cup \{j\}} u_{ir} - \max_{n \in \mathbb{N} \setminus \{j\}} u_{in} \ge 0, \arg\max_{r \in \mathbb{R} \cup \{j\}} u_{ir} = j\right).$$
(A5)

Thus, I recast the maximization problem as a two-step problem where the worker finds the best local options in the two subsets, and then compares them to each other. Notice here that *j* changes interpretation from the best option in *K* to the best option in the set $R \cup \{j\}$.

Step 2: Removing the argmax condition from P(s)

Both $\max_{r \in R \cup \{j\}} u_{ir}$ and $\max_{n \in N \setminus \{j\}} u_{in}$ are Gumbel distributed according to (A1). According to (A3), $\max_{r \in R \cup \{j\}} u_{ir}$ does not depend on $\arg \max_{r \in R \cup \{j\}} u_{ir} = j$. Conditional on the value of $\max_{r \in R \cup \{j\}} u_{ir}$, $\max_{n \in N \setminus \{j\}} u_{in}$ must also be independent of the argmax condition. Hence, this condition can be removed from the set of conditions in (A5):

$$P(s) = \Pr\left(\max_{r \in \mathcal{R} \cup \{j\}} u_{ir} - \max_{n \in \mathcal{N} \setminus \{j\}} u_{in} \mid \boldsymbol{x}, \max_{r \in \mathcal{R} \cup \{j\}} u_{ir} - \max_{n \in \mathcal{N} \setminus \{j\}} u_{in} \ge 0\right).$$
(A6)

One may now think of *j* as a regular occupation rather than as the optimal choice, although it still represents the observed occupation of worker *i*. Finally, define v_{ij} as the difference between the two maxima in (A6):

$$v_{ij} \equiv \max_{r \in R \cup \{j\}} u_{ir} - \max_{n \in N \setminus \{j\}} u_{in}.$$
 (A7)

One may then write:

$$P(s) = \Pr\left(v_{ij} \le s \mid \boldsymbol{x}, v_{ij} \ge 0\right).$$
(A8)

Step 3: Determining the distribution of *v*_{*ii*}

By (A1), $\max_{r \in R \cup \{j\}} u_{ir}$ and $\max_{n \in N \setminus \{j\}} u_{in}$ are both gumbel with scale one. Since by (A6) one may now think of *j* as a regular occupation, they must also be independent. Hence, by the final property of the Gumbel distribution stated in Section A.1, v_{ij} is Logistic distributed with scale one.

From equation (A1), one can also infer the location parameter μ of v_{ij} . It equals the difference between the location parameters of the two Gumbels. Using the multinomial logit formula from (3), μ can then readily be rewritten as a function of only choice probabilities. Finally, recall that equation (4) defines $\rho(\mathbf{x}, j, N)$ as the *ex ante* probability of working in a non-routine outside option, i.e.,

$$\rho(\mathbf{x}, j, N) \equiv \sum_{n \in N \setminus \{j\}} p_n(\mathbf{x}).$$

The location parameter of the distribution of v_{ij} is:

$$\mu = \mu(\mathbf{x}, j, N) = \ln\left(\sum_{r \in R \cup \{j\}} e^{u_r(\mathbf{x})}\right) - \ln\left(\sum_{n \in N \setminus \{j\}} e^{u_n(\mathbf{x})}\right)$$
$$= \ln\left(\sum_{r \in R \cup \{j\}} e^{u_r(\mathbf{x})} / \sum_{n \in N \setminus \{j\}} e^{u_n(\mathbf{x})}\right)$$
$$= \ln\left(\frac{\sum_{r \in R \cup \{j\}} e^{u_r(\mathbf{x})}}{\sum_{k \in K} e^{u_k(\mathbf{x})}} / \frac{\sum_{n \in N \setminus \{j\}} e^{u_n(\mathbf{x})}}{\sum_{k \in K} e^{u_k(\mathbf{x})}}\right)$$
$$= \ln\left(\sum_{r \in R \cup \{j\}} p_r(\mathbf{x}) / \sum_{n \in N \setminus \{j\}} p_n(\mathbf{x})\right)$$
$$= \ln\left(\frac{1}{\sum_{n \in N \setminus \{j\}} p_n(\mathbf{x})} - 1\right)$$
$$= \ln\left(\frac{1}{\rho(\mathbf{x}, j, N)} - 1\right).$$
(A9)

Step 4: Finding the distribution function of the utility surplus

Next, I turn to finding an expression for the CDF and PDF of the utility surplus. First, denote the PDF and CDF of the Logistic($\mu(\mathbf{x}, j, N), 1$) distribution by F'(.) and F(.), respectively. These are:

$$F'(s \mid \mathbf{x}, j, N) = \frac{e^{-s + \mu(\mathbf{x}, j, N)}}{\left(1 + e^{-s + \mu(\mathbf{x}, j, N)}\right)^2}$$
$$F(s \mid \mathbf{x}, j, N) = \frac{1}{1 + e^{-s + \mu(\mathbf{x}, j, N)}}$$
(A10)

Next, by properties of conditional probabilities, one can write the CDF of the utility surplus in terms of F(.):

$$P(s) = \Pr\left(v_{ij} \le s \mid \mathbf{x}, v_{ij} \ge 0\right)$$

=
$$\frac{\Pr\left(0 \le v_{ij} \le s \mid \mathbf{x}\right)}{1 - \Pr\left(v_{ij} \le 0 \mid \mathbf{x}\right)}$$

=
$$\begin{cases} \frac{F(s \mid \mathbf{x}, j, N) - F(0 \mid \mathbf{x}, j, N)}{1 - F(0 \mid \mathbf{x}, j, N)} \text{ for } s \ge 0\\ 0 \text{ otherwise} \end{cases}$$
 (A11)

Finally, to obtain the PDF of the utility surplus, differentiate (A11) with respect to *s*:

$$P'(s) = \begin{cases} \frac{F'(s \mid \mathbf{x}, j, N)}{1 - F(0 \mid \mathbf{x}, j, N)} \text{ for } s \ge 0\\ 0 \text{ otherwise} \end{cases}$$
(A12)

A.3 Deriving the occupation specialization index

Let *S* represent the utility surplus. Using the PDF of the logistic distribution from (A10) and its relationship with the utility surplus PDF in (A12), one can obtain a closed-form solution to the expected value of the surplus. Finally, plugging in the location parameter from (A9) and simplifying gives the OSI:

$$OSI(\mathbf{x}, j, N) \equiv \mathbb{E} \left[u_{ij} - \max_{n \in N \setminus \{j\}} u_{in} \mid \mathbf{x}, \arg \max_{k \in K} u_{ik} = j \right]$$
$$= \left(1 - \frac{1}{1 + e^{\mu(\mathbf{x}, j, N)}} \right)^{-1} \int_{0}^{\infty} S \frac{e^{-S + \mu(\mathbf{x}, j, N)}}{\left(1 + e^{-S + \mu(\mathbf{x}, j, N)} \right)^{2}} dS$$
$$= \frac{1 + e^{\mu(\mathbf{x}, j, N)}}{e^{\mu(\mathbf{x}, j, N)}} \ln \left(1 + e^{\mu(\mathbf{x}, j, N)} \right)$$
$$= -\frac{\ln \left(\rho(\mathbf{x}, j, N) \right)}{1 - \rho(\mathbf{x}, j, N)}$$
(A13)

Both $\ln(\rho(\mathbf{x}, j, N))$ and $1/(1 - \rho(\mathbf{x}, j, N))$ are increasing in $\rho(\mathbf{x}, j, N)$. Therefore, the OSI is a monotonically decreasing function of $\rho(\mathbf{x}, j, N)$.

A.4 Deriving the expected utility loss

The expected value of the utility change following the wage premium shock in (2) can be written as a probability-weighted average of d_j and a conditional expected value of *S*. A closed-form solution can then be derived in a similar way as in (A13). Again, let *S* represents the utility surplus random variable. Then:

$$\mathbb{E}\left[\Delta u_{ij} \mid \mathbf{x}, j\right] = \mathbb{E}\left[\max\left\{-d_j, -\left(u_{ij} - \max_{n \in N \setminus \{j\}} u_{in}\right)\right\} \mid \mathbf{x}, \operatorname*{arg\,max}_{k \in K} u_{ik} = j\right]$$
$$= -d_j - \Pr(S \le d_j) \left(\mathbb{E}\left[S \mid S \le d_j\right] - d_j\right)$$
$$= -\frac{\left(1 - \frac{1}{1 + e^{-d_j + \mu(\mathbf{x}, j, N)}}\right) d_j + \int_0^{d_j} S \frac{e^{-S + \mu(\mathbf{x}, j, N)}}{(1 + e^{-S + \mu(\mathbf{x}, j, N)})^2} dV}{1 - \frac{1}{1 + e^{\mu(\mathbf{x}, j, N)}}}$$
$$= -\frac{d_j - \ln\left(1 + (e^{d_j} - 1)\rho(\mathbf{x}, j, N)\right)}{1 - \rho(\mathbf{x}, j, N)}.$$
(A14)

For any value of $d_j = \delta > 0$, (A14) is monotonically increasing (i.e., the absolute loss becomes smaller) in $\rho(\mathbf{x}, j, N)$. To see this, differentiate (A14) with respect to ρ which is used as shorthand for $\rho(\mathbf{x}, j, N)$:

$$\frac{\partial \mathbb{E} \left[\Delta u_{ij} \mid \mathbf{x}, j\right]}{\partial \rho} = -\frac{d_j - \ln\left(1 + \left(e^{d_j} - 1\right)\rho\right)}{(1 - p)^2} + \frac{\frac{e^{d_j - 1}}{1 + \left(e^{d_j} - 1\right)\rho}}{1 - \rho}$$
$$= \frac{1}{(1 - p)^2} \left[-\left(d_j - \ln\left(1 + \left(e^{d_j} - 1\right)\rho\right)\right) + \frac{(1 - \rho)\left(e^{d_j} - 1\right)}{1 + \left(e^{d_j} - 1\right)\rho} \right]$$
$$= \frac{1}{(1 - p)^2} \left[-\left(\ln\left(e^{d_j}\right) - \ln\left(1 + \left(e^{d_j} - 1\right)\rho\right)\right) + \frac{e^{d_j} - (1 + \left(e^{d_j} - 1\right)\rho)}{1 + \left(e^{d_j} - 1\right)\rho} \right]$$
$$= \frac{1}{(1 - p)^2} \left[-\ln\left(\frac{e^{d_j}}{1 + \left(e^{d_j} - 1\right)\rho}\right) + \frac{e^{d_j}}{1 + \left(e^{d_j} - 1\right)\rho} - 1 \right]$$
(A15)

This may be rewritten as:

$$\frac{\partial \mathbb{E} \left[\Delta u_{ij} \mid \mathbf{x}, j \right]}{\partial \rho} = a \left[b - 1 - \ln \left(b \right) \right],$$

where $a = \frac{1}{(1 - \rho)^2}$ and $b = \frac{e^{d_j}}{(1 + (e^{d_j} - 1)\rho)}.$ (A16)

a > 1 for any $0 < \rho < 1$. For routine workers with $d_j = \delta > 0$, $e^{d_j} \ge 1 + (e^{d_j} - 1)\rho > 1$. This implies that b > 1. In turn, $b - 1 - \ln(b) > 0$. Thus, the derivative is positive, implying monotonicity.

Appendix B The artificial neural network

B.1 Sample and variables

As discussed in Section 3, 30 percent of observations in my full pre-period sample are used for training the neural network. Twenty percent of this sub-sample is used for validation when choosing the specific features of the network, while 80 percent are used for training.

The vector of input variables, \mathbf{x} , is of length 129. It incorporates the following variables: Age fixed effects (FE); A second-order polynomial in all ability measures; Education level FE; Education field FE; Region of residence FE; Separate second-order polynomials in experience between t - 13 and t - 1 in 14 industry categories.

The vector of output variables, y, includes indicators for the 22 different occupations.

B.2 Neural network

The neural network consists of an input layer, a single hidden layer and an output layer. Each layer includes a number of neurons, or variables. The input layer consists of the explanatory variables. Each input neuron is associated with a set of weights (coefficients) to be estimated that map to the second, hidden, layer of neurons, the length of which is discussed below. The value of the *j*th hidden neuron for a particular observation *i* is determined by $z_i^j = \mathbf{w}^{l_z}\mathbf{x}_i + b_j^z$, where \mathbf{w}_j^z is a vector of all the weights mapping the input neurons to the *j*th hidden neuron, \mathbf{x}_i is a vector of the inputs for observation *i* and b_j is the bias (intercept) for *j*. z_i^j is then transformed using a non-linear activation function f(.) before being passed forward to the next layer, i.e., the input into the output layer from the *j*th hidden node for the *i*th observation is $a_i^j = f(z_i^j)$. This process is then repeated for all hidden nodes *j*, and the resulting values are collected in the vector \mathbf{a}_i . To find the value of the *m*th neuron in the output layer, I calculate $y_i^m = \mathbf{w}_m^{ly}\mathbf{a}_i + b_m^y$ Finally, I also apply an activation function g(.) to the values of the output layer neurons.

The network can be described in matrix notation as:

$$\boldsymbol{Y} = g\left(\boldsymbol{W}^{output} \times f\left(\boldsymbol{W}^{hidden} \times \boldsymbol{X} + \boldsymbol{b}^{hidden}\right) + \boldsymbol{b}^{output}\right).$$
(B1)

where g(.) and f(.) are activation functions which operate on the individual cells of the matrices and return matrices of the same dimensions. **X** is a $v^{input} \times N$ matrix, where v^{input} is the length of the input vector and N represents the number of observations. **W**^{hidden} is a $v^{hidden} \times v^{input}$ -dimensional matrix containing all the weights that link the inputs to the hidden layer and where v^{hidden} is the number of hidden neurons. For instance, **W**^{hidden} refers to the weight linking the *j*th hidden neuron and *k*th input. **b**^{hidden} is a vector of length v^{hidden} containing the biases (intercepts) for each hidden neuron. $\boldsymbol{W}_{i,j}^{output}$ contain the $v^{output} \times v^{hidden}$ weights mapping the values of the hidden neurons to the output neurons and \boldsymbol{b}^{output} contains all the output neuron biases.

For the hidden layer, I use a leaky rectified linear unit (RELU) activation function. More specifically, $f(x) = \max\{0.001x, x\}$. The inclusion of such an activation function allows the neural network to accommodate important non-linear effects and interactions between multiple variables. I set the number of hidden neurons to 150.

The output layer is activated using the softmax function, $p_k = e^{x_k} / \sum_m e^{x_m}$, where x_k is the value of the *k*th output neuron and the denominator sums over all output layer neurons.

The loss function used to train the network is the categorical cross-entropy loss:

$$Loss = -\sum_{i \in N} \sum_{k \in K} I_{ik} \times \ln\left(\hat{p}_{ik}\right).$$
(B2)

where I_{ik} is an indicator for if the true choice of observation *i* is occupation *k* and zero otherwise. Thus, the loss function is based on the sum of the log propensities assigned to the true choice of all observations.

The network is trained using the Adam optimization algorithm (Kingma and Ba, 2014), which is an extension of stochastic gradient descent. Training is done for a maximum of 100 epochs (cycling through all observations in the training data) with a batch size (the number of observations passed through the network at each step) of 1000. But to avoid overfitting, I utilize what is known as an early stopping rule: When the loss in the validation sample has not decreased for 10 subsequent epochs, I stop the training algorithm and restore the model from all previous epochs that is associated with the smallest loss in the validation sample. Moreover, I apply dropout on the hidden layer at a rate of 50 percent: Between each weight update cycle, half of all the neurons are randomly chosen and set to zero. This forces the network to be less reliant on a small set of neurons for predictions.

To implement the network, I use the Keras API for R, which uses the Tensorflow machine learning platform.

Appendix C Additional figures

This section presents all the additional figures that are referenced in the empirical analysis section. Please see the main text for a description of the results from each figure.



Figure C1. Simulated relationship between outcomes and the OSI



Notes: To verify the predictive properties of the OSI for the loss, I simulate the model in Section 2 for 10 routine and 10 non-routine occupations. I generate 10 million workers *i* with individual standard Gumbel draws for each occupation. All workers belong to one of 100 equally-sized groups $g \in \{1, ..., 100\}$. Each group draws a deterministic utility term u_{gk} for each occupation *k* from the standard normal distribution. The wage premium shock is set to $\delta = 1$. For each worker, I find $\max_{k \in K} \{u_{ik}\} - \max_{n \in N \setminus \{j\}} \{u_{in}\}$. Next, I calculate group-specific probabilities as the number of workers in *g* in *k* (N_{gk}) divided by the total size of *g* (N_g); $p_{gk} = N_{gk}/N_g$. Workers are classified as either routine or non-routine depending on whether arg $\max_{k \in K} \{u_{ik}\} \in R$. To construct the OSI, I use the group-specific probabilities. Finally, I calculate $\max \{-d_j, -(\max_{k \in K} \{u_{ik}\} - \max_{n \in N \setminus \{j\}} \{u_{in}\})\}$ and $\mathbb{1}[(\max_{k \in K} \{u_{ik}\} - \max_{n \in N \setminus \{j\}} \{u_{in}\}) < d_j]$.



Figure C2. Histograms of the occupation specialization indices

Notes: The figure reports histograms of the main OSI metric based on predictions from the neural network, the alternative OSI based on the multinomial logit, and the WOSI which measures the estimated wage surplus in worker's occupations. All the metrics are censored at the 1st and 99th percentile and standardized to mean zero, standard deviation one.





Notes: The figure plots the assigned occupation choice probability against the true out-ofsample probability. Let I_{ik} represent an indicator for if the true choice of individual *i* is occupation *k*. First, an observation containing (I_{ik}, \hat{p}_{ik}) is constructed for each individual observation *i* × potential occupation choice *k*. I then divide them into 100 bins based on the percentiles of the distribution of \hat{p}_{ik} at this level of observation. Finally, I plot the average \hat{p}_{ik} against the actual probability, i.e., the average of the indicator I_{ik} , for each bin.



Figure C4. Predicted and actual choice probabilities for switchers by destination

Notes: The figure is based on data for occupation switchers between t and t + 10 to t + 12. Separately by destination occupation, I produce binned scatterplots of the relationship between actual probability to switch to a certain occupation and the predicted choice probability conditional on not choosing the source occupation j. More specifically, I use the predicted propensities from the neural network for all outside occupations, rescale them to sum to one, create bins based on the assigned probabilities, and relate them to the actual destination choice probabilities of occupation switchers. For all occupations, there is a strong positive relationship between the predicted probabilities and the actual choices, albeit typically lower than one-to-one.



Figure C5. Occupation specialization by skill level, education, county and age for low- to middle-skilled occupations

Notes: The figures report the share of workers in a certain subgroup based on worker characteristics (e.g., the lowest decile group of cognitive ability) that belong to different OSI quintile groups on the left vertical axis. The right vertical axis instead plots the average OSI for the different groups. This is done separately for workers in routine (left column) and non-routine (right column) occupations.

Figure C6. Occupation specialization by previous industry experience for low- to middle-skilled occupations



Notes: The figures report the share of workers with different years of industry-specific experience that belong to different OSI quintile groups on the left vertical axis. The right vertical axis instead plots the average OSI for the different experience levels. This is done separately for workers in routine (left figure) and non-routine (right figure) occupations.

Figure C7. Non-parametric relationship between specialization and log wage growth by routine and non-routine occupation switchers and stayers



Notes: The figure relates the OSI to log wage growth between t and t + 10 to t + 12 separately for workers initially in routine and non-routine occupations. In panel (a) this is done for workers who remained in the same detailed occupation. Panel (b) instead does so for workers who moved to another, non-routine occupation. See Figure 3 for additional information.



Figure C8. Estimated effect of specialization for different sets of interacted controls

Notes: The figure reports estimates and corresponding 95 percent CIs for the OSI and interaction between OSI and the routine indicator from different versions of equation (8). All regressions include controls for a second-order polynomial in the eight skill dimensions and 14 previous industry experience variables as well as age, education and field, region, initial \times future year, and occupation fixed effects. Each point on the horizontal axis represents a separate model which interacts the routine indicator with one of four sets of control variables: A secondorder polynomial in all skill variables; education level and field FE; region of residence FE; age FE and a second-order polynomial in the previous industry experience variables.



Figure C9. Occupation-specific estimates of occupation specialization on labor market outcomes

Notes: The figure reports point estimates and 95-percent confidence intervals separately by occupation for different outcomes from a version of equation (8) which instead of the routine indicator interacts the OSI with all occupations to obtain occupation-specific OSI estimates. All regressions include controls for a second-order polynomial in the eight skill dimensions and 14 previous industry experience variables as well as age, education and field, region, initial \times future year, and occupation fixed effects. Standard errors are clustered at the individual level. On the horizontal axis, the occupations are ranked by their routine task intensity (from low to high). The vertical dashed line represents the cutoff between routine and non-routine occupations. Marker size is determined by the size of each occupation.

Essay II. Understanding Occupational Wage Growth

Co-authored with Adrian Adermon, Georg Graetz, and Yaroslav Yakymovych

Acknowledgements: We thank Michael Böhm, Stefan Eriksson, Peter Fredriksson, Lena Hensvik, Lisa Laun, Oskar Nordström Skans, as well as seminar participants at the Uppsala Center for Labor Studies, the Future of Labor workshop in Berlin, and the Swedish Conference in Economics for helpful suggestions.

1 Introduction

The past four decades have seen systematic shifts in occupational employment across industrialized countries, with high- and low-paying occupations gaining at the expense of the middle. This is commonly interpreted as reflecting labor demand shifts induced by technological change, consumer demand, or offshoring. However, the impact of such occupation-level demand shifts on the wage structure is far from clear. First, occupations appear to play a minor role in driving changes in wage inequality, at least in terms of descriptive decomposition exercises. Second, occupational employment and wage growth typically do not feature a strong positive correlation. Finally, wage inequality trends differ substantially across countries, while occupational employment shifts are highly similar.¹

In this paper, we shed light on these puzzles by studying occupational wage growth in Sweden 1996–2013. Swedish employment shifts are similar to those elsewhere (Adermon and Gustavsson, 2015), but the wage structure is dramatically compressed compared to most other industrialized countries, and growth in inequality has been moderate and episodic (Graetz, 2020). We show that, as elsewhere, occupations do not appear to play an important role in basic decompositions of changes in wage inequality.

However, as has long been recognized, any analysis of occupational demand shifts and wages must address selection problems arising from workers' systematic sorting into occupations (see for instance Roy, 1951; Acemoglu and Autor, 2011; Böhm, 2020). For example, a positive demand shock to computer programmers may manifest itself as an increase in the price paid for a unit of programming output. At the same time, this increased *occupational wage premium* draws in workers from other occupations, who may be less productive than incumbents, thus leaving *observed* wages approximately unchanged.

Our starting point for overcoming the selection problem is to focus on occupation stayers, whose wage growth comes closer to the growth in premia as time-invariant skills are differenced out and the composition of workers is left unchanged (Cortes, 2016). We address attenuation bias stemming from selection on idiosyncratic shocks using the method developed by Böhm et al. (2023).

The second challenge we face is that occupations may differ in how workers accumulate skills over the life-cycle, so that differential wage growth among

¹The polarization of occupational employment in the US and elsewhere has been documented by Wright and Dwyer (2003), Goos and Manning (2007), and Autor et al. (2006); and Goos et al. (2014). See in particular Adermon and Gustavsson (2015) for the Swedish case. Goos et al. (2014) provide evidence in favor of a technological explanation. Barany and Siegel (2018) emphasize structural change and consumer demand instead. In a decomposition exercise, Hoffmann et al. (2020) find only a minor role for occupations in driving rising wage inequality. Roys and Taber (2019), Böhm (2020), and Böhm et al. (2023) highlight the lack of a strong positive correlation between occupational employment and wage growth in the US and Germany.

occupation stayers may reflect not only differential premium growth (Deming, 2021). Moreover, occupational experience profiles may have shifted over time, for instance due to technology-induced obsolescence of skills (Deming and Noray, 2020). A theoretically motivated restriction that has been suggested as a solution to this identification problem is the concept of a "flat spot", a point in the life-cycle when the derivative of human capital with respect to experience is zero (Heckman et al., 1998; Bowlus and Robinson, 2012). We propose a novel approach for implementing this restriction, namely to re-center the experience profiles around the flat spot. This leaves us with greater statistical power as we are not forced to restrict the sample to workers near the flat spot. More importantly, it allows us to estimate experience profiles for each occupation and point in time.²

Finally, we explore to what extent our estimated premium growth is driven by changes in occupation-specific skill returns. A growing literature documents changing skill returns in the aggregate, and suggests that occupations may be important drivers of such trends (Deming, 2017; Edin et al., 2022). Given the availability of cognitive and psycho-social skill measures from the Swedish military enlistment, we are able to control for differential changes in skill returns in our estimation.

Our findings are as follows. First, premium growth is positively correlated with employment growth (and more strongly so than is raw wage growth). Second, premium growth is also positively correlated with initial wages. These two findings together imply our third finding, namely that in the absence of compositional changes between-occupation wage inequality would have increased more than it actually has. Fourth, experience profiles vary strongly across occupations at any given point in time, and while they are stable in some occupations, in others they show large changes. These results are robust to allowing for changes in specific returns to cognitive and psycho-social skills.

The positive association between premium growth and employment shifts suggests that variation in premium growth is mostly due to demand side factors. At the same time, our results suggest that there is an important life-cycle component to shifts in the occupational wage structure.

Our findings are consistent with a recent and growing literature documenting the importance of compositional changes in counteracting occupationlevel demand shifts (Cortes, 2016; Böhm, 2020; Cavaglia and Etheridge, 2020; Böhm et al., 2023). Our contribution compared to these studies is first, to provide comparable evidence for the Swedish economy, which at first glance features a very different wage structure. Second, to estimate time-varying

 $^{^{2}}$ Böhm et al. (2023) assume that experience profiles are constant—following much of the theoretical literature on task-biased technological change—and use a pre-period of uniform premium growth to estimate these profiles. Our data do not go back in time sufficiently to make this approach feasible.

occupation-specific experience profiles. And third, to allow for time-varying occupation-level skill returns when estimating wage premium growth.

To the best of our knowledge, the joint estimation of premium growth and experience profiles has only been attempted by one other paper, Böhm et al. (2023). Their identification assumption is that the profiles are fixed over time, and that during the decades prior to 1985 any differential wage premium growth was negligible, so that experience profiles can be estimated using a prior period. Our assumptions and identification strategy differ from Böhm et al. (2023), and we view our approach as complementary.³

The remainder of the paper is structured as follows. Section 2 presents our theoretical framework, discusses identification challenges as well as our proposed solutions, and develops counterfactual scenarios. We describe our data in Section 3. Section 4 contains our results, and Section 5 concludes.

2 Theoretical framework and empirical strategy

The theoretical motivation for our empirical exercise is the standard Roy model in which workers sort into occupations based on comparative advantage. Rather than estimating a completely specified model, our point of departure is an assumption about the data-generating process for potential wages. In Section 2.1, we explore how key parameters of this wage equation can be identified under different assumptions about occupational choice. In Section 2.2, we show how changes in overall wage inequality can be attributed to occupationlevel driving forces, and develop counterfactual scenarios based on our estimated wage equation.

2.1 Identifying the parameters of the wage function

Suppose that individual worker *i*'s log wage in occupation *k* and year *t*, w_{ikt} , is given by

$$w_{ikt} = \pi_{kt} + \alpha_{ik} + \beta_k \mathbf{s}_i + g_k (x_{ikt} - x^*) + \varepsilon_{ikt}, \qquad (1)$$

where π_{kt} is a potentially time-varying occupation-specific wage premium; α_{ik} is an unobserved worker-occupation fixed effect; \mathbf{s}_i is a vector of observable skills with its associated occupation-specific returns β_k ; x_{ikt} is the worker's experience in the occupation measured in years and centered around x^* , to be

³Using unusual rich data, Böhm et al. (2023) are able to estimate across-occupation experience profiles, that is, the extent to which a year of work experience in one occupation increases the worker's productivity in this and all other occupations. In contrast, we estimate how wage growth in each occupation and at each point in time varies with overall potential labor market experience. (Given the limited length of our panel, we cannot construct workers' occupational histories.) Therefore, our estimates have a different structural interpretation from those in Böhm et al. (2023).
discussed below; g_k is an occupation-specific experience profile; and ε_{ikt} is an i.i.d. shock. Our main goal is to estimate π_{kt} for each occupation, or at least its *change relative to a reference occupation*.

For the moment, let us assume that workers choose the occupation in which they earn the highest wage in each period, abstracting from dynamic considerations. Furthermore, let us assume for now that the shock ε_{ikt} is realized *after* workers have made their choice. These assumptions are the same as in Cortes (2016). This leaves us with two potential threats to identification: Selection on unobserved time-invariant characteristics, and occupation-specific experience profiles. We address these in turn.

Selection on time-invariant characteristics

Consider the first difference of Equation (1),

$$\Delta w_{ik} = \Delta \pi_k + g_k (x_{ikt} - x^*) - g_k (x_{ikt} - 1 - x^*) + \Delta \varepsilon_{ik}, \qquad (2)$$

where Δ is the first difference operator, so that $\Delta X = X_t - X_{t-1}$. If we estimate Equation (2) using the sample of occupation stayers, we can be sure that selection on time-invariant skills α_{ik} and \mathbf{s}_i is accounted for, since these terms are differenced out. An alternative method accomplishing this is of course to estimate Equation (1) in levels and to include worker-by-occupation fixed effects, as in Cortes (2016). We prefer the first difference specification for two reasons. First, it allows us to run separate regressions for each year, and thus work with datasets of manageable size. Second, our data on wages and occupations come from repeated cross-sectional samples, so that it is difficult to construct long panels of individuals workers, and to accurately capture longer occupational spells (see Section 3).

Occupation-specific experience profiles

For concreteness, we approximate the profile by a polynomial of order M, $g_k(x) = \sum_{m=1}^{M} \gamma_{km} (x - x^*)^m$. Under this assumption, the component of wage growth due to experience—among occupation stayers—now becomes

$$g_k(x_{ikt} - x^*) - g_k(x_{ikt} - 1 - x^*) = \gamma_{k1} + \sum_{m=2}^M \gamma_{km} \left\{ (x_{ikt} - x^*)^m - (x_{ikt} - 1 - x^*)^m \right\}.$$

The wage growth equation to be estimated is thus

$$\Delta w_{ik} = \Delta \pi_k + \gamma_{k1} + \sum_{m=2}^{M} \gamma_{km} \{ (x_{ikt} - x^*)^m - (x_{ikt} - 1 - x^*)^m \} + \Delta \varepsilon_{ik}.$$
(3)

Estimation of Equation (3) for a given occupation yields a constant term $\theta_{kt} = \Delta \pi_k + \gamma_{k1}$. Thus, the challenge is to separate out changes in premia from the constant term of the experience profile. Note that γ_{k1} is the effect of additional experience at the point $x_{it} = x^*$. Human capital theory (Ben-Porath,

1967; Heckman et al., 1998) suggests that there comes a point in a worker's life cycle when human capital accumulation stops, or even reverses due to depreciation—a so-called flat spot where the marginal effect of experience on wages is zero. Thus, if x^* is set to be at the flat spot, then $\gamma_{k1} = 0$, solving the identification problem as we now have $\theta_{kt} = \Delta \pi_k$.⁴

We illustrate this strategy using a concrete example: The wage growth of *physical and engineering science technicians* from 2005–06. Figure 1 plots changes in log wages, together with the fitted polynomial, against potential experience re-centered around different values—the assumed locations of the flat spot. The fitted polynomial comes from estimating Equation (3) choosing m = 4. Grey dashed lines mark the constant term estimated by the regressions, equal to premium growth under the assumption $\gamma_{k1} = 0$. The data reveal a strong downward trend in wage growth, consistent with faster skill accumulation among inexperienced workers, as well as a flattening of this relationship at higher levels of potential experience. The top-left panel does not re-center the data, thus yielding a large estimated premium growth of around 8 percent. But an assumption of zero skill accumulation for labor market entrants is of course highly implausible. Assuming flat spots at higher values such as 25, 30, or 35 all yield estimated premium growth around 2 percent, as shown in the remaining panels.

Figure 1 illustrates that choosing the flat spot means picking a point on the fitted first-differenced experience profile and attributing all wage growth at that point to growth in the premium.⁵ Relying on a parametric prediction for the profile yields greater statistical power compared to simply using average wage growth at the flat spot.

Figure 1 also raises the question whether the flat spot can be determined in a data-driven way. In general, the answer is no. Consider three hypothetical experience-wage profiles plotted in the top row of Figure 2. As we do not observe workers' time-invariant occupation-specific skills, we cannot estimate the profiles in levels. We thus first-difference the profiles, shown in the bottom row. The challenge remains to separate premium growth from skill accumulation. Consider first column (a). The differenced profile reproduces the nearly flat region of the original in-levels profile. While it may not be easy to determine the *exact* location of the flat spot, this would also not matter greatly for the estimated premium growth. However, recall that the econometrician cannot see the top row. As column (b) shows, a flat region in first differences

⁴Our approach is related to Fosse and Winship (2019), who address the identification problem arising in the presence of age, cohort, and time effects. They highlight that it is only linear effects that are unidentified, and explain how one can bound these. However, a single restriction is often sufficient for point identification, as is the case in our context.

⁵To be precise, the flat spot assumption says that $g'_k(x^*) = 0$. In the polynomial case, $g'_k(x^*) = \sum_{m=1}^{M} \gamma_{km} m(x-x^*)^{m-1} = \gamma_{k1}$. Here, the flat spot assumption $\gamma_{k1} = 0$ does not imply that $\Delta w_{ik}|_{x=x^*} = \Delta \pi_k$ exactly, which requires $\sum_{m=2}^{M} \gamma_{km} \{-(-1)^m\} = 0$. However, in practice these equations will hold approximately, as is the case in Figure 1.



Figure 1. Illustration of flat spot identification

Notes: Grey dashed lines mark the constant term from estimating the experience profiles, equal to wage premium growth under the respective flat spot assumptions. The data include all individuals who worked as physical and engineering science technicians in 2005 and 2006. See Section 3 for further details on sample selection.

Figure 2. Simulated wage-experience profiles





can also result from a locally log-linear profile in levels. In this case, the true flat spot at 34 cannot be detected based on first differences. Finally, consider column (c), which shows a profile of roughly constant curvature and hence no flat region. Again, it is not obvious how to choose the flat spot based on the first-differenced profile.

Given that the flat spot cannot be identified without further assumptions, our approach is to set it at 30 for all occupations, while also reporting results for alternative values. In a further robustness check, we estimate flat spots under the additional assumption that the true profiles are strictly concave except for possible flat sections (that is, linear segments with non-zero slope, as in the middle column of Figure 2, are prohibited). In this case, the second derivative of the profile will be maximized (closest to zero) at the flat spot, so that any statistic of interest should change by the least amount—in absolute value—at the true flat spot. See Section A for further details.

A key advantage of our method is that it allows us to jointly estimate experience profiles and premium growth. Moreover, as we estimate separate models for each year, we essentially estimate time-varying experience profiles.⁶ A third advantage is that we retain greater statistical power than existing approaches in the literature which implement the flat spot idea using only data on workers near the flat spot (Bowlus and Robinson, 2012; Cavaglia and Etheridge, 2020).⁷

We note that the interpretation of our estimated profiles is affected by the measurement of occupation-specific experience. With a panel that is relatively short (20 years) relative to the typical length of working lives, it is not possible to construct complete occupational histories for each worker.⁸ In our baseline specification we therefore use potential overall labor market experience, based on age and years of schooling. Given this, the occupation-specificity of the γ_{km} 's means that experience is differently valued across occupations, but it does not matter in which occupation this experience was gained. Alternatively, one can simply interpret the estimated profiles as describing the wage growth in a given occupation and year as a function of potential overall labor market experience. This function will depend not only on deep structural parameters, but also on the characteristics—such as occupational histories—of the workers staying in that occupation in that year (and the year before).

⁶Strictly speaking, the experience profile in *levels* must be constant across the two adjacent years. This would not matter if we had specified the profile in changes in the first place. However, starting with a levels specification is arguably more natural given the Roy framework.

⁷It is also possible to implement flat spot identification via an iterative procedure (Lagakos et al., 2017).

⁸Another challenge is that private sector workers in Sweden are sampled, as we discuss in the data section.

Time-varying skill returns

A key finding in recent research on inequality is that wage returns to various skills have evolved differently over time, with occupations seemingly playing an important role. While this is interesting in its own right, here we are mainly concerned with the impact of such changes on our ability to estimate changes in occupational wage premia. Suppose, then, that returns to portable skills vary over time,

$$w_{ikt} = \pi_{kt} + \alpha_{ik} + \beta_{kt}\mathbf{s}_i + g_k(x_{ikt} - x^*) + \varepsilon_{ikt},$$

so that wage growth now becomes

$$\Delta w_{ik} = \Delta \pi_k + (\Delta \beta_{kt}) \mathbf{s}_i + \sum_{m=2}^{M} \gamma_{km} \{ (x_{ikt} - x^*)^m - (x_{ikt} - 1 - x^*)^m \} + \Delta \varepsilon_{ik}.$$
(4)

For selected cohorts of Swedish men we actually have at our disposal the skill measures for which changing wage returns have been documented. We can thus assess whether our baseline estimates of $\Delta \pi_k$ are robust to controlling for these measures, by estimating Equation (4) where the vector \mathbf{s}_i contains cognitive and psycho-social skills, as described further in the data section.

Selection on idiosyncratic shocks

Let us now allow for selection on the idiosyncratic shock ε_{ikt} , as well. The constant term from estimating Equation (3), imposing the flat spot assumption $\gamma_{k1} = 0$, now becomes $\theta_{kt} = \Delta \pi_k + E[\Delta \varepsilon_{ik} | k_{it} = k_{i,t-1} = k]$. The second term no longer equals zero, due to selection. Other things equal, occupations experiencing relatively fast premium growth will retain more workers with a bad realization of the shock, while occupations in which premia decline only retain those workers with very good realizations. Therefore, selection on idiosyncratic shocks biases downward the between-occupation variance in premium growth. This bias is more severe the larger is the variance of ε_{ikt} . A method to correct for this bias, developed by Böhm et al. (2023), is to include occupation switchers in a regression of wage growth on workers' average choices. We implement this method as a robustness check.

Remaining issues

There are a number of issues which are beyond the scope of this paper. These include forward-looking occupational choice, amenities, search frictions, and long-term wage contracts. We believe that addressing any one of these requires estimation of a fully specified structural model (for recent examples, see Roys and Taber, 2019; Traiberman, 2019).

2.2 Occupational drivers of changes in wage inequality

A key objective of this paper is to assess the importance of occupations for changes in wage inequality. Therefore, we need to formally characterize how changes in inequality relate to occupation-level changes such as differential premium growth and worker re-allocation. We closely follow Böhm et al. (2023).

First, by the Law of Total Variance, $Var(w_{it}) = E[Var(w_{it}|k)] + Var(E[w_{it}|k])$. That is, overall wage inequality can be decomposed into a within-occupation and a between-occupation component. Without specifying the distribution of skills, it is difficult to say much about how changes in premia affect the within component, so we focus on the between component.

To ease notation, let us from now on write $w_{kt} \equiv E[w_{it}|k]$ and $\Delta w_k \equiv \Delta E[w_i|k]$, and similarly for other variables. The difference operator $\Delta X \equiv X_1 - X_0$ denotes changes between two points in time 0 and 1, not necessarily adjacent years.

Note, to integrate out the conditioning variable—occupational choice—we must specify a distribution of occupational employment. When decomposing the variance at a given point in time, the obvious choice is to use the distribution at that point. But when considering changes over time, we need to be explicit about the distribution. We use subscripts to do so.

The change in between-occupation wage inequality can be written as

$$\operatorname{Var}_{1}(w_{k1}) - \operatorname{Var}_{0}(w_{k0}) = \underbrace{\operatorname{Var}_{0}(w_{k1}) - \operatorname{Var}_{0}(w_{k0})}_{\text{change at initial employment}} + \underbrace{\operatorname{Var}_{1}(w_{k1}) - \operatorname{Var}_{0}(w_{k1})}_{\text{re-allocation}}.$$
(5)

Define $y_{kt} \equiv w_{kt} - \pi_{kt}$, which captures workers' skills in the broadest sense all parts of log wages not determined by the occupation premium. The first component on the right-hand side of Equation (5) can be broken down as

$$\operatorname{Var}_{0}(w_{k1}) - \operatorname{Var}_{0}(w_{k0}) = \operatorname{Var}_{0}(\Delta w_{k}) + 2\operatorname{Cov}_{0}(w_{k0}, \Delta w_{k})$$
$$= \operatorname{Var}_{0}(\Delta \pi_{k}) + \operatorname{Var}_{0}(\Delta y_{k}) + 2\operatorname{Cov}_{0}(\Delta \pi_{k}, \Delta y_{k})$$
$$+ 2\operatorname{Cov}_{0}(w_{k0}, \Delta \pi_{k}) + 2\operatorname{Cov}_{0}(w_{k0}, \Delta y_{k}).$$
(6)

From Equation (6), we see how differential changes in premia may affect changes in wage inequality, and at the same time, how their effects may be offset by opposing forces. In particular, all the components of the decomposition involving changes in average skills Δy_k , as well as the re-allocation term from Equation (5), can be seen as potentially countervailing effects due to workers' re-sorting. In contrast, all terms only involving $\Delta \pi_k$ and initial mean wages w_{k0} can be interpreted as giving the counterfactual increase in betweenoccupation wage inequality in the absence of re-sorting. That is, with worker composition unchanged, we have

$$\operatorname{Var}_{0}(w_{k1}) - \operatorname{Var}_{0}(w_{k0}) = \operatorname{Var}_{0}(\Delta \pi_{k}) + 2\operatorname{Cov}_{0}(w_{k0}, \Delta \pi_{k}), \tag{7}$$

which is a key object of interest in our analysis. Equation (7) shows that, holding worker composition constant, changes in wage premia have a large

effect on wage inequality if they are very dispersed, or if they are positively correlated with initial mean wages.

3 Data description

3.1 Data sources

We obtain demographic information (year of birth, sex, municipality of residence, education, immigration status) from Statistics Sweden's LISA database, covering the population of Swedish residents 1985-2016. LISA also contains employment status in November each year, annual salary income, as well as industry and municipality of workplace.

Some information that is key for our purposes is absent from LISA. In particular, LISA does not contain weeks and hours worked, nor occupation. For this, we turn to a database called Swedish Wage Structure Statistics (henceforth WSS). WSS contains three-digit occupation codes according to the *SSYK96* classification 1996–2013, and according to the *SSYK2012* classification 2014-2016.⁹ The two classifications cannot be mapped unambiguously, and breaks in employment trends are apparent even at higher levels of aggregation. We therefore end our main analysis in 2013.

WSS also contains contractual monthly wage rates. This in combination with annual salary income allows us to determine annual labor supply. Most importantly, these contractual wage rates are the main outcome of interest for our analysis, since we are interested in the price of labor.

A drawback of WSS is that outside the public sector, only a sample of workers is available. Sampling is stratified by firms, with large firms being more likely to be drawn. This does not pose any problems for cross-sectional analysis—sampling weights are provided—but makes it more difficult to analyze dynamic phenomena such as occupational mobility. We discuss this issue further in the next sub-section.

For some of our analysis, we use test scores collected during military enlistment in the last decades of the 20th century, after which conscription was gradually phased out. Among birth cohorts 1952-1981, more than 90 percent of Swedish-born males are covered by these data. We use a combined measure of cognitive skills based on four different standardized tests of inductive, verbal and spatial skills, and technical comprehension, and a measure of psychosocial skills (sometimes called "non-cognitive skills") based on a half-hour, semi-structured interview with a certified psychologist.¹⁰ We standardize the

⁹SSYK stands for *Standard för Svensk Yrkesklassificering*, literally "Standard for Swedish occupation classification", a version of the International Standard Classification of Occupations (ISCO).

¹⁰The intent of the interview was to evaluate the psychological fitness for coping with military service. See Lindqvist and Vestman (2011) and Fredriksson et al. (2018) for more details on these data.

two measures within each draft cohort to have mean zero and standard deviation one. To ensure comparability, we estimate our main specification also for the sub-sample of male cohorts for which enlistment data are available.

3.2 Sample selection and construction of variables

Our population of interest includes all Swedish employees aged 18–64 during the years 1996–2013 (sometimes extended to 2016). Employees are individuals who are employed in November and whose annual labor earnings are no less than three times the 10th-percentile monthly wage. We calculate individual wage growth for all adjacent years, dropping anyone with wage growth below the first or above the 99th percentile for each pair of years.¹¹

We calculate potential labor market experience as years elapsed since year of graduation, based on highest education attained and a school starting age of six. To reduce noise, we drop observations with potential experience below two and greater than 40 years. Due to the limited length of the panel as well as due to sampling, we are unable to construct actual occupation-specific experience.

We use sampling weights to adjust for stratification. The raw weights supplied in WSS feature some extremely large values, and this may introduce noise, especially when multiplying the weights for a first-difference analysis using a two-year panel. Whenever we work with individual, two-year panel data, we therefore trim the weights following the procedure of Potter (1990).¹² However, when computing aggregate moments, we use the original weights.

For our baseline analysis we use the 3-digit-level *SSYK96* occupational classification, which includes 101 occupations. However, we sometimes use a coarser classification for descriptive and other purposes.

4 Results

4.1 Raw wages, wage premia, and employment

To set the stage, we document the relationship between growth in averages wages and growth in employment as well as initial wages, across occupations for the period 1996–2013. Panel (a) of Figure 3 plots the long difference in log wages against the long difference in the log of employment, with each marker

¹¹Extreme values of wage growth—five or more standard deviations away from the mean—may occur because individuals enter into and exit from executive positions (Skans et al., 2009). We drop extreme values as these can have a large impact on the results.

¹²The procedure is as follows. We first fit a Beta(α, β) distribution to the weights. Second, weights whose estimated cumulative probability is above 99 percent are trimmed to the estimated 99th percentile. Third, weights are re-scaled such that their sum is unchanged. This procedure is repeated ten times.



Figure 3. Growth in wages, premia, and skills against employment growth, 1996-2013

Notes: The figure plots the growth mean log wages, cumulative estimated wage premia, as well as the implied change in mean skills, against the change in log employment. Wage premia are estimated according to our baseline specification Equation (8). Each marker represents one of 101 occupations. The size of each marker is determined by the employment share in the first year and the regression line is weighted accordingly. We use original survey weights when calculating occupation size and mean log wage.

representing one occupation. First, by moving along the horizontal axis, we see much variation in employment growth. Production, operators, and craft occupations tend to see low (often negative) employment growth, while on average, employment growth appears highest among managers, professionals, and technicians. Clerical and services occupations fall somewhere in between. However, there is much variation even within these broad categories. Turning to wage growth, there is a positive but rather weak relationship with employment growth. Panel (a) of Figure 4 reveals an even weaker relationship of average wage growth with initial (1996) average wages.

However, as discussed in Section 2, average wage growth captures both changes in occupational wage premia and changes in worker composition and hence average skills. In order to isolate changes in wage premia, we operationalize Equation (2) by estimating separate regressions of year-on-year changes in individual log wages on occupation fixed effects and a polynomial in potential experience:

$$\Delta w_{it} = \varphi_{kt} + \sum_{m=2}^{4} \gamma_{km} \{ \tilde{x}_{it}^m - (\tilde{x}_{it} - 1)^m \} + u_{it}, \qquad (8)$$

where φ_{kt} are occupation-specific fixed effects; $\tilde{x}_{it} \equiv x_{it} - x^*$ is potential experience re-centered around the assumed flat spot in the experience profile; and γ_{km} are polynomial coefficients allowed to vary by occupation. In our main specification, we use a fourth-order polynomial and re-center potential experience at 30 years. We report robustness checks with respect to these choices below. We estimate separate regressions for each pair of adjacent years in our sample. In order to control for changes in worker composition, we use only individuals who remained in the same occupation across both years, $k_{it} = k_{i,t-1}$.

Under the assumption that there is no selection on idiosyncratic shocks and that $\gamma_{m1} = 0$ (the flat spot assumption), the fixed effects φ_{kt} consistently estimate premium growth $\Delta_{t-1}^t \pi_k$ for an adjacent pair of years. We estimate premium growth over the full period by simply accumulating the estimated year-on-year changes, $\widehat{\Delta_{1997}^{2013}} \pi_k = \sum_{t=1997}^{2013} \widehat{\varphi}_{kt}$.

Our premium growth estimates are plotted against employment growth in Panel (b) of Figure 3. The relation between premium growth and employment growth is stronger than that of mean wage growth—the slope is steeper, and R^2 almost triples. This pattern implies that while demand factors were pushing up wage premia during this period, changes in the skill composition of workers acted as a counteracting force, resulting in the tempered trend we see in average wage growth. This is consistent with a situation where growing labor demand in certain occupations attracts new workers with lower productivity than the incumbents—and conversely, occupations with falling labor demand might let their lower-productivity workers go first. The implied change in skill composition can be backed out from our estimates by simply subtracting the estimated changes in premia from the observed changes in average wages.



Figure 4. Growth in wages, premia, and skills against initial wages, 1996-2013

Notes: The figure plots the growth mean log wages, cumulative estimated wage premia, as well as the implied change in mean skills, against initial mean log wages. Wage premia are estimated according to our baseline specification Equation (8). Each marker represents one of 101 occupations. The size of each marker is determined by the employment share in the first year and the regression line is weighted accordingly. We use original survey weights when calculating occupation size and mean log wage.



Figure 5. Growth in wages, premia, and skills against growth in schooling, 1996–2013

Notes: The figure plots the growth mean log wages, cumulative estimated wage premia, as well as the implied change in mean skills, against initial the growth in average years of schooling. Wage premia are estimated according to our baseline specification Equation (8). Each marker represents one of 101 occupations. The size of each marker is determined by the employment share in the first year and the regression line is weighted accordingly. We use original survey weights when calculating occupation size and mean log wage.

This is shown in Panel (c) of Figure 3. As expected, faster growing occupations have seen falling implied skill levels in their workforce, although this relationship is not very strong.

Panel (b) of Figure 4 shows that premium growth is strongly positively associated with initial wages. Given Equation (6), this suggests that premium growth would cause an increase in between-occupation wage inequality in the absence of compositional changes. However, panel (c) of Figure 4 already gives an idea of how strong these compositional changes might be—growth in average skills are strongly negatively related to initial wages. We explore these issues in detail in Section 4.2.

One way to assess the plausibility of the estimated growth in skills is to check its association with changes in years of schooling. Panel (c) Figure 5 shows that there is indeed a positive relationship, with a fairly high R^2 of 0.2. On the other hand, panel (b) of the same figure shows a negative association between premium growth and changes in years of schooling, consistent with lower educated workers moving into occupations experiencing positive demand changes.¹³

While the evidence presented so far suggests that the forces predicted by the Roy model are at work, it remains to assess their quantitative importance for the evolution of wage inequality in Sweden. We do so next.

4.2 Decomposing changes in between-occupation wage inequality

To quantify the role of differential premium growth for changes in betweenoccupation inequality in Sweden, we use our estimates to calculate the counterfactual scenarios developed in Section 2.2. We first focus on the long difference 1996–2013 and then examine changes at annual frequency.

The first three lines in column (1) of Table 1 show the change in the observed variance of log wages, the change in between-occupation variance, as well as the change in between-occupation variance holding occupational employment fixed at 1996. The variance of log wages increased by 0.026 between 1996–2013, from 0.073 in 1996. (To avoid excessive decimal places, we multiply the variance and its components by 100 from here on.) Although the wage distribution in Sweden is still highly compressed compared to other countries (Graetz, 2020), this increase is large in relative terms.

Between-occupation wage inequality accounts for just over half of the increase in overall variance. But this is allowing for the employment weights in the calculation of variance to change over time. If employment shifts from

¹³For completeness, Figure B2 displays the respective bi-variate associations of wage growth, premium growth, and implied skill growth, showing positive correlations between wage growth and premium growth, and wage growth and skill growth, and a negative correlation between premium growth and skill growth.

	(1)	(2)	(3)	(4)
	Baseline	Commo	n flat spot	Occspec.
		25	35	flat spot
Total				
$\Delta \text{Var}(w_{ik})$	2.57			
Between				
$\Delta \operatorname{Var}(w_k)$	1.31			
$\Delta \operatorname{Var}_0(w_k)$.39			
Components				
$\operatorname{Var}_{0}(\Delta \pi_{k}) + 2\operatorname{Cov}_{0}(w_{k0}, \Delta \pi_{k})$.94	2.03	.29	.66
$\operatorname{Var}_0(\Delta \pi_k)$.23	.43	.17	.27
$2\mathrm{Cov}_0(w_{k0},\Delta\pi_k)$.71	1.59	.12	.4
$\operatorname{Var}_0(\Delta y_k)$.26	.37	.24	.25
$2\mathrm{Cov}_0(w_{k0},\Delta y_k)$	57	-1.45	.02	26
$2\mathrm{Cov}_0(\Delta\pi_k,\Delta y_k)$	24	55	16	27

Table 1. Decomposition of changes in between-occupation wage inequality

Notes: The table reports results from a decomposition of the change in the betweenoccupation variance in wages between 1996 and 2013 for different flat spot levels. See Equation (6) for the formal statement of the decomposition. Column (1) uses a common flat spot for all occupations, at 30 years of potential experience, when estimating growth in wage premia. Columns (2)–(3) vary this common flat spot as indicated. Column (4) estimates a flat spot for each occupation using the procedure described in Section A. All figures have been multiplied by 100 for readability. middle- to both high- and low-paying occupations, we should expect betweenoccupation inequality to increase even if wage premia do not grow differentially. The phenomenon of job polarization has been extensively documented in the literature (Goos et al., 2014; Adermon and Gustavsson, 2015), and Figure B1 confirms that it is present also in our sample period.

Our main interest, however, is in occupation-level drivers of wage inequality that are due to differential changes in compensation for a fixed set of workers. The third row in Table 1 shows that holding employment fixed at 1996 levels, the contribution of between-occupation variance shrinks by more than two thirds. But, as discussed above, changes in observed wages at the occupation level may mask changes in composition. To assess the role of differential growth in occupational wage premia, we perform the decomposition given by Equation (6).

Column (1) of Table 1 presents our baseline results, with the flat spot set at 30 for all occupations. Holding worker composition constant, the increase in between-occupation variance would have been 0.94 based on our decomposition. This is more than twice the increase in the between-occupation variance of raw wages (at constant employment), and almost 40 percent of the increase in the overall variance of log wages. Most of this effect is due to a positive covariance between initial wages and premium growth, while the variance in premium growth plays a relatively minor role. The last two rows in column (1) of Table 1 show the attenuating forces: Changes in worker skills are negatively correlated with both initial wages and growth in wage premia.

Figure 6 shows the evolution of the variance components year-on-year. Interestingly, during the period 1996–2001, which saw the fastest growth in wage inequality, the attenuating forces of a changing skill composition are absent, and the no-sorting counterfactual closely tracks between-occupation inequality in raw wages (at constant employment). The attenuating forces emerge only after 2001.¹⁴

4.3 Robustness checks

We conduct a number of robustness checks for the results that depend on the estimation of wage premium growth. First, we vary the location of the flat spot. As expected given the shape of wage-experience profiles and the above discussion of Figure 1, the decomposition results are sensitive to the choice of flat spot, as seen in columns (3) and (4) of Table 1. The sensitivity varies

¹⁴Column (1) in panel B of Table B1 displays the decomposition results for the sub-period 2001–2013. Figures B3 and B4 display the relationships between growth in wages, premia, and implied skills on the one hand, and employment growth and initial wages on the other, for 2001–2013. While overall inequality changed little during this time, the pattern of premium growth and compositional changes is qualitatively very similar to that for the whole sample period.



Figure 6. Decomposition of changes in between-occupation inequality 1996-2013

Notes: The figure plots the results from the decomposition given by Equation (6) for every year pair {1996, t} $\forall t \in \{1996, ..., 2013\}$.

by component: The variance of premium growth appears more stable than the covariance of premium growth and initial wages.¹⁵

However, when we attempt to determine occupation-specific flat spots in a data-driven way—based on the assumption of strictly concave wage-experience profiles (except for possible flat regions) as discussed in Section A—we obtain results quite similar to our baseline specification (column (4)). Note also that setting the flat spot at zero, which would be implied if we simply added higher-order terms of potential experience without re-centering them, yields clearly unreasonable results (column (2) of Table B2).

Table B2 displays the decomposition components for a set of further robustness checks. These include changing the order of the polynomial in potential experience; adjusting for endogenous mobility using the method of Böhm et al. (2023); allowing for differential growth in wage premia at the level of regions and industries; pooling the data to estimate time-invariant experience profiles; restricting the data to men with non-missing enlistment scores; controlling for time-varying returns to cognitive and non-cognitive skills within this restricted sample; and dividing the data by gender. The results are robust in the sense that the no-sorting counterfactual in the majority of cases is of similar or even larger magnitude compared to the baseline.¹⁶

Finally, we probe the robustness of the associations of premium growth and implied skills with employment growth, initial wages, and years of schooling. The results are shown in Figures B5 and B6, and once again are largely similar to the baseline specification.

4.4 Changes in occupational experience profiles

A key advantage of our empirical approach is that we are able to estimate occupational experience profiles that vary over time. We estimate profiles for 101 occupations and each pair of years from 1996–2013. Due to space constraints, we only show estimated profiles for the largest (in terms of average employment 1996–2013) 3-digit occupation in each of nine main categories, for the years 1997, 2002, 2008, and 2013.

The estimated profiles are shown in Figure 7. There are several noteworthy findings. First, in all occupations wage growth is fastest for inexperienced workers, but this pattern is much more pronounced in some occupations (finance & sales professionals, building frame workers) than in others (personal care workers). Second, while in some occupations the profiles are stable (building frame workers), in others they show large changes over time (com-

¹⁵As premium growth and skill growth are strongly negatively correlated, this difference in sensitivity is mirrored by the other components.

¹⁶Note that using a polynomial of order one or forcing the experience profiles to be constant over time are more restrictive and thus inferior to our baseline specification.



Figure 7. Estimated occupational experience profiles for selected occupations and years

Notes: The figure plots the estimated experience profiles from Equation (8) for the indicated occupations and years.



Figure 8. Wage-experience profiles over time

(a) Steepness of wage-experience profiles

Notes: The figure characterizes the distribution of the experience profiles estimated by Equation (8) over time (panel (a)) and shows how variance in selected characteristics of experience profiles is related to variance in premium growth (panel (b)).

puting professionals). Third, profiles are steepest in the late 1990s in several cases, but this is not a universal pattern.

To further investigate changes over time, we plot the median as well as quartiles of two measures capturing the steepness of the profiles, namely, the value of the profile at ten years of potential experience as well as the maximum value (both in levels). Panel (a) of Figure 8 reveals that, by both measures, profiles were indeed somewhat steeper in the late 1990s. But even more striking is that the steepness of the profiles was much more dispersed in that period.

Finally, we explore if there is a systematic relationship between dispersion in wage-experience profiles and dispersion in wage premium growth. Panel (b) of Figure 8 plots the variances of the two steepness measures along with the variance of premium growth against time. It appears that years with higher dispersion in profiles also tend to see higher dispersion in premium growth.

5 Conclusion

We contribute to the literature on shifts in the wage structure by jointly estimating growth in occupational wage premia and occupation-specific life cycle wage profiles. We document substantial changes in occupations' relative premia in Sweden in recent decades, which are masked in the raw wage data due to worker sorting. There is a positive association between premium growth and employment growth, suggesting that workers have been responsive to changes in occupational demand. The relative premia changes are estimated to have substantially contributed to the increase in overall wage inequality. We also document large heterogeneity in life-cycle profiles across occupations, as well as substantial shifts of the profiles over time. Allowing for occupation-level changes in returns to cognitive and psycho-social skills has little effect on the results.

Our results suggest that although the overall wage structure in Sweden is highly compressed, forces related to technological change do influence the wage structure and drive workers' occupational choices. An open question is why the increase in Swedish wage inequality was concentrated in the late 1990s. This could be due to a temporary rise in the flexibility of collective bargaining, or it may reflect uneven technological change, for instance a transitional period of technology adoption (Beaudry et al., 2016).

The method we propose to estimate changes in occupational wage premia may fruitfully be applied to other settings, especially those in which experience profiles appear to change over time, and in cases where only short (twoyear) panels of workers are available.

References

- Acemoglu, D. and D. Autor (2011). "Skills, Tasks and Technologies: Implications for Employment and Earnings". *Handbook of Labor Economics*. Ed. by O. Ashenfelter and D. Card. Vol. 4. Elsevier. Chap. 12, 1043–1171.
- Adermon, A. and M. Gustavsson (2015). "Job Polarization and Task-Biased Technological Change: Evidence from Sweden, 1975–2005". *The Scandinavian Journal of Economics* 117.3, 878–917.
- Autor, D., L. Katz, and M. Kearney (2006). "The Polarization of the U.S. Labor Market". *American Economic Review* 96.2, 189–194.
- Barany, Z. L. and C. Siegel (2018). "Job Polarization and Structural Change". *American Economic Journal: Macroeconomics* 10.1, 57–89.
- Beaudry, P., D. A. Green, and B. M. Sand (2016). "The Great Reversal in the Demand for Skill and Cognitive Tasks". *Journal of Labor Economics* 34.S1, S199–S247.
- Ben-Porath, Y. (1967). "The Production of Human Capital and the Life Cycle of Earnings". *Journal of Political Economy* 75.4, 352–365.
- Böhm, M. (2020). "The price of polarization: Estimating task prices under routine-biased technical change". *Quantitative Economics* 11.2, 761–799.
- Böhm, M., H.-M. v. Gaudecker, and F. Schran (2023). "Occupation Growth, Skill Prices, and Wage Inequality". *Journal of Labor Economics*, forthcoming.
- Bowlus, A. J. and C. Robinson (2012). "Human Capital Prices, Productivity, and Growth". *American Economic Review* 102.7, 3483–3515.
- Cavaglia, C. and B. Etheridge (2020). "Job Polarization and the Declining Quality of Knowledge Workers: Evidence from the UK and Germany". *Labour Economics* 66, 101884.
- Cortes, G. M. (2016). "Where Have the Middle-Wage Workers Gone? A Study of Polarization Using Panel Data". *Journal of Labor Economics* 34.1, 63–105.
- Deming, D. J. (2017). "The Growing Importance of Social Skills in the Labor Market". *The Quarterly Journal of Economics* 132.4, 1593–1640.
- Deming, D. J. (2021). "The Growing Importance of Decision-Making on the Job". Working Paper 28733. National Bureau of Economic Research.
- Deming, D. J. and K. Noray (2020). "Earnings Dynamics, Changing Job Skills, and STEM Careers". *The Quarterly Journal of Economics* 135.4, 1965–2005.
- Edin, P.-A., P. Fredriksson, M. Nybom, and B. Öckert (2022). "The Rising Return to Noncognitive Skill". *American Economic Journal: Applied Economics* 14.2, 78–100.

- Fosse, E. and C. Winship (2019). "Bounding Analyses of Age-Period-Cohort Effects". *Demography* 56.5, 1975–2004.
- Fredriksson, P., L. Hensvik, and O. N. Skans (2018). "Mismatch of Talent: Evidence on Match Quality, Entry Wages, and Job Mobility". *American Economic Review* 108.11, 3303–38.
- Goos, M. and A. Manning (2007). "Lousy and Lovely Jobs: The Rising Polarization of Work in Britain". *The Review of Economics and Statistics* 89.1, 118–133.
- Goos, M., A. Manning, and A. Salomons (2014). "Explaining Job Polarization: Routine-Biased Technological Change and Offshoring". American Economic Review 104.8, 2509–26.
- Graetz, G. (2020). "Technological Change and the Swedish Labor Market". Working Paper 2020:19. IFAU.
- Heckman, J. J., L. Lochner, and C. Taber (1998). "Explaining Rising Wage Inequality: Explorations with a Dynamic General Equilibrium Model of Labor Earnings with Heterogeneous Agents". *Review of Economic Dynamics* 1.1, 1–58.
- Hoffmann, F., D. S. Lee, and T. Lemieux (2020). "Growing Income Inequality in the United States and Other Advanced Economies". *Journal of Economic Perspectives* 34.4, 52–78.
- Lagakos, D., B. Moll, T. Porzio, N. Qian, and T. Schoellman (2017). "Life Cycle Wage Growth across Countries". *Journal of Political Economy* 126.2, 797–849.
- Lindqvist, E. and R. Vestman (2011). "The Labor Market Returns to Cognitive and Noncognitive Ability: Evidence from the Swedish Enlistment". *American Economic Journal: Applied Economics* 3.1, 101–128.
- Liu, D. C. and J. Nocedal (1989). "On the limited memory BFGS method for large scale optimization". *Mathematical Programming* 45, 503–528.
- Potter, F. J. (1990). "A study of procedures to identify and trim extreme sampling weights". *Proceedings of the American Statistical Association, Section on Survey Research Methods*. Vol. 225230. American Statistical Association Washington, DC.
- Roy, A. D. (1951). "Some Thoughts on the Distribution of Earnings". *Oxford Economic Papers* 3.2, 135–146.
- Roys, N. and C. Taber (2019). Working Paper 26453. National Bureau of Economic Research.
- Skans, O. N., P.-A. Edin, and B. Holmlund (2009). "Wage Dispersion Between and Within Plants: Sweden 1985-2000". *The Structure of Wages: An International Comparison*. NBER Chapters. National Bureau of Economic Research, Inc, 217–260.

- Traiberman, S. (2019). "Occupations and Import Competition: Evidence from Denmark". *American Economic Review* 109.12, 4260–4301.
- Wright, E. O. and R. E. Dwyer (2003). "The patterns of job expansions in the USA: a comparison of the 1960s and 1990s". *Socio-Economic Review* 1.3, 289–325.

Appendix A Procedure for estimating occupation-specific flat spots

Suppose that experience profiles are strictly concave except for possible flat regions. That is, linear segments with non-zero slope, as in the middle column of Figure 2, are prohibited. Formally, $g''(x) \le 0$, $g''(x) = 0 \Rightarrow g'(x) = 0$. This implies that the second derivative of the profile will be maximized (closest to zero) at the flat spot, so that any statistic of interest should change by the least amount—in absolute value—at the true flat spot. We use this insight to pin down the flat spot in a data-driven way.

Recall from Section 2.2 that the change in between-occupation variance of log wages, at constant employment, can be decomposed as

$$Var_{0}(w_{k1}) - Var_{0}(w_{k0}) = Var_{0}(\Delta w_{k}) + 2Cov_{0}(w_{k0}, \Delta w_{k})$$

$$= Var_{0}(\Delta \pi_{k}) + Var_{0}(\Delta y_{k}) + 2Cov_{0}(\Delta \pi_{k}, \Delta y_{k})$$

$$+ 2Cov_{0}(w_{k0}, \Delta \pi_{k}) + 2Cov_{0}(w_{k0}, \Delta y_{k}).$$

(A1)

Denote by μ the components of the decomposition,

$$\mu \in \mathscr{M} \equiv \{ \operatorname{Var}_0(\Delta \pi_k), \operatorname{Var}_0(\Delta y_k), 2\operatorname{Cov}_0(w_{k0}, \Delta \pi_k), 2\operatorname{Cov}_0(\Delta \pi_k, \Delta y_k) \}.$$

Each of the elements of \mathcal{M} depends on the change in the premia $\Delta \pi_k$, which in turn depend on the chosen flat spots. However, the sum of all components on the right-hand side of Equation (A1) is constant, so we exclude $2\text{Cov}_0(w_{k0}, \Delta y_k)$ from the set \mathcal{M} .

Let $\overline{\omega}$ denote the vector of changes in premia, and let $\tilde{\mathbf{x}}$ denote the vector of candidate flat spots. Both vectors contain *K* elements, where *K* is the total number of occupations, indexed by *k*. We denote the above-mentioned functional dependence by $\mu \equiv \mu(\overline{\omega}(\tilde{\mathbf{x}}))$. Using the chain rule, we define the sensitivity of μ to changing the flat spot, in absolute terms, as

$$|d\mu(\boldsymbol{\varpi}(\mathbf{\tilde{x}}))| \equiv \left|\sum_{k} \frac{\partial \mu}{\partial (\Delta \pi_{k})} \times \sum_{k'} \frac{\partial (\Delta \pi_{k'})}{\partial \tilde{x}_{k'}} \times d\tilde{x}_{k'}\right|.$$

Under strictly concave experience profiles, we conjecture that $|d\mu(\boldsymbol{\varpi}(\tilde{\mathbf{x}}))|$ attains its minimum at or near the vector of true flat spots \mathbf{x}^* , and similarly for the sum over $|d\mu(\boldsymbol{\varpi}(\tilde{\mathbf{x}}))|$,

$$\mathbf{x}^* = \arg\min_{\tilde{\mathbf{x}}} \sum_{\mu \in \mathscr{M}} |d\mu(\boldsymbol{\varpi}(\tilde{\mathbf{x}}))|.$$
(A2)

We implement the optimization problem given by Equation (A2) in practice by solving

$$\mathbf{x}^* = \arg\min_{\tilde{\mathbf{x}}} S \times \sum_{\hat{\mu} \in \mathscr{M}} \left[\left(\widehat{\mu} (\tilde{\mathbf{x}} + \tau) - \widehat{\mu} (\tilde{\mathbf{x}}) \right)^2 + \left(\widehat{\mu} (\tilde{\mathbf{x}} - \tau) - \widehat{\mu} (\tilde{\mathbf{x}}) \right)^2 \right],$$

where $\hat{\mu}$ denote the estimated moments, τ is size-*k* vector with constant elements representing step size, and *S* is a scaling factor chosen for numerical stability. We set the elements of τ to equal 0.01 and S = 1e+7. We use the L-BFGS-B method (Liu and Nocedal, 1989) implemented by the optim package in R. We impose $\tilde{x}_k \in [25, 40] \forall k$. As the procedure appears to be sensitive to initial values, we draw initial values at random from the continuous uniform distribution U(26, 39) for each \tilde{x}_k . This process is repeated 100 times. We then choose the \mathbf{x}^* with the lowest associated loss.

Note that in principle, given strictly concave profiles one should be able to find the flat spots by minimizing the sensitivity of the $\Delta \pi_k$'s instead of a moment that is a function of them. However, approximating the experience profiles by a polynomial does not guarantee that the estimated profiles are actually strictly concave. Alternatively, one could impose a functional form on the profiles that does guarantee strict concavity. We attempted to do this, but the estimation turned out to be highly unstable.





Notes: The figure plots the growth in log employment against mean log wages in 1996. In Panel (b), log wages have been percentile-ranked, weighted by initial employment. Each marker represents one of 101 occupations. The size of each marker is determined by the employment share in the first year and the regression lines are weighted accordingly. We use original survey weights when calculating occupation size and mean log wage.



Figure B2. Relations between growth rates

Notes: The figure plots the bivariate relationships between the growth in mean log wages, cumulative estimated wage premia, and the implied change in mean skills. Each marker represents one of 101 occupations. The size of each marker is determined by the employment share in the first year and the regression lines are weighted accordingly. We use original survey weights when calculating occupation size and mean log wage.



Figure B3. Growth in wages, premia, and skills against employment growth, 2001–2013

Notes: The figure plots the growth mean log wages, cumulative estimated wage premia, as well as the implied change in mean skills, against the change in log employment. Wage premia are estimated according to our baseline specification Equation (8). Each marker represents one of 101 occupations. The size of each marker is determined by the employment share in the first year and the regression line is weighted accordingly. We use original survey weights when calculating occupation size and mean log wage.



Figure B4. Growth in wages, premia, and skills against initial wages, 2001-2013

Notes: The figure plots the growth mean log wages, cumulative estimated wage premia, as well as the implied change in mean skills, against initial mean log wages. Wage premia are estimated according to our baseline specification Equation (8). Each marker represents one of 101 occupations. The size of each marker is determined by the employment share in the first year and the regression line is weighted accordingly. We use original survey weights when calculating occupation size and mean log wage.



Figure B5. Premia, skills, and employment growth—robustness checks

Notes: The table reports the coefficients, 95 percent confidence intervals, and coefficients of determination from separately regressing cumulative estimated wage premia and the implied change in mean skills (growth in average wage minus premium growth) against the change in log employment at the occupation level for different sets of premia estimates. See the text for descriptions of how these estimates are produced. The weight assigned to each occupation is determined by the employment share in the first year. We use original survey weights when calculating occupation size and mean log wage.



Figure B6. Premia, skills, and initial wages-robustness checks

Notes: The figure reports the coefficients, 95 percent confidence intervals, and coefficients of determination from separately regressing cumulative estimated wage premia and the implied change in mean skills (growth in average wage minus premium growth) against initial mean log wage at the occupation level for different sets of premia estimates. See also the notes to Figure B5.



Figure B7. Premia, skills, and schooling-robustness checks

Notes: The figure reports the coefficients, 95 percent confidence intervals, and coefficients of determination from separately regressing cumulative estimated wage premia and the implied change in mean skills (growth in average wage minus premium growth) against growth in average years of schooling at the occupation level for different sets of premia estimates. See also the note to Figure B5.

	(1) Baseline	(2) Common flat spo	(3) ot	(4) Occspec.
	•	25	35	flat spot
(a) 1996–2013				
Total				
$\Delta \mathrm{Var}(w_{ik})$	2.57			
Between				
$\Delta \mathrm{Var}(w_k)$	1.31			
$\Delta \operatorname{Var}_0(w_k)$.39			
Components				
$\mathrm{Var}_0(\Delta \pi_k) + 2\mathrm{Cov}_0(w_{k0},\Delta \pi_k)$.94	2.03	.29	.66
$\operatorname{Var}_0(\Delta\pi_k)$.23	.43	.17	.27
$2 { m Cov}_0(w_{k0},\Delta\pi_k)$.71	1.59	.12	4.
$\operatorname{Var}_0(\Delta y_k)$.26	.37	.24	.25
$2 Cov_0(w_{k0}, \Delta y_k)$	57	-1.45	.02	26
$2\mathrm{Cov}_0(\Delta\pi_k,\Delta y_k)$	24	55	16	27
(b) 2001–2013				
Total				
$\Delta \mathrm{Var}(w_{ik})$.34			
Between				
$\Delta \operatorname{Var}(w_k)$.07			
$\Delta \operatorname{Var}_0(w_k)$	31			
Components				
$\mathrm{Var}_0(\Delta\pi_k)+2\mathrm{Cov}_0(w_{k0},\Delta\pi_k)$.31	.93	08	.15
$\operatorname{Var}_0(\Delta \pi_k)$.00	.13	.08	г.
$2 { m Cov}_0(w_{k0},\Delta\pi_k)$.22	.79	16	.05
$\operatorname{Var}_0(\Delta y_k)$.14	.18	.12	.15
$2 { m Cov}_0(w_{k0},\Delta y_k)$	68	-1.25	3	51
$2\mathrm{Cov}_0(\Delta\pi_k,\Delta_{y_k})$	08	17	06	1
<i>Notes</i> : The table reports results from a decomposition of the C See Equation (6) for the formal statement of the decomposition when estimating growth in wage premia. Columns (2) - (3) values of the table of table o	change in the between-occupation. Column (1) uses a common flar ary this common flat spot as incommon that spot as incomposite the spo	on variance in wages fe lat spot for all occupati dicated. Column (4) es	or different flat spot levels an ons, at 30 years of potential ev timates a flat spot for each o	d periods. xperience, occupation
using the procedure described in Section A. All figures have be	een multiplied by 100 for reada	bility.		

 Table B1. Decomposition results—sub-periods

	(1)	(2)	(3)	(4)	(2)	(9)	E	(8)	6)	(10)	(11)	(12)
	Baseline	Flat spot = 0	Ч	oly. order		BvGS	Reg. &	Pooled	Draft	Draft	Men	Women
			1	3	5		ind.	spec.	sample	scores		
(a) 1996–2013												
Total												
$\Delta \mathrm{Var}(w_{ik})$	2.57											
Between												
$\Delta \mathrm{Var}(w_k)$	1.31											
$\widetilde{\Delta} \operatorname{Var}_0(w_k)$.39											
Components			-					Ĩ				
$\operatorname{Var}_0(\Delta \pi_k) + 2\operatorname{Cov}_0(w_{k0},\Delta \pi_k)$.94	24.49	2.57	1.21	1.07	1.28	.66	.97	1.64	1.59	- 85	2.02
$\operatorname{Var}_0(\Delta \pi_k)$.23	17.6	.67	.31	.25	.31	.17	.24	.41	4.	.25	×.
$2\mathrm{Cov}_0(w_{k0},\Delta\pi_k)$.71	6.89	1.9	80.	.82	76.	.49	.73	1.23	1.19	.59	1.22
$\operatorname{Var}_0(\Delta y_k)$.26	17.09	.56	.31	.25	.29	.25	.27	4.	.39	.28	.83
$2\mathrm{Cov}_0(w_{k0},\Delta y_k)$	57	-6.75	-1.77	75	68	83	35	59	-1.09	-1.05	45	-1.08
$2\mathrm{Cov}_0(\Delta\pi_k,\Delta y_k)$	24	-34.44	98	37	25	35	18	25	55	54	28	-1.38
(b) 2001–2013												
Total												
$\Delta \operatorname{Var}(w_{ik})$.34											
Between												
$\Delta \operatorname{Var}(w_k)$.07											
$\Delta \mathrm{Var}_0(w_k)$	31											
Components												
$\mathrm{Var}_0(\Delta\pi_k)+2\mathrm{Cov}_0(w_{k0},\Delta\pi_k)$.31	14.5	1.22	.45	.41	.58	.17	.1	.51	.53	.26	.84
$\operatorname{Var}_0(\Delta\pi_k)$	60.	9.31	.22	.12	60.	.13	.07	.11	60.	.08	.11	.29
$2\mathrm{Cov}_0(w_{k0},\Delta\pi_k)$.22	5.19	1	.33	.32	.45	.1	01	.41	.45	.15	.55
$\operatorname{Var}_0(\Delta y_k)$.14	9.69	.29	.16	.13	.16	.12	.14	.12	.13	.15	.35
$2\mathrm{Cov}_0(w_{k0},\Delta y_k)$	68	-5.65	-1.45	79	78	91	56	45	87	91	61	-1.01
$2\mathrm{Cov}_0(\Delta\pi_k,\Delta y_k)$	08	-18.85	36	13	08	14	04	1	07	06	11	5
<i>Notes</i> : The table reports results	s from a decc	mposition of th	e change	in the be	etween-c	ccupation	variance	in wages f	or differen	specifica	tions and	periods.
See Equation (6) for the formal	l statement of	f the decomposi	tion and	the text f	or detail	s on the c	lifferent sn	ecification	s. All fion	res have h	een mult	inlied hv
100 for readability.	~	ma ann ann		·		· · · · · · · · · · · · · · · · · · ·	1			· · · · · · · · · · · · · · · · · · ·		buce of

Table B2. Decomposition results—further specifications

107
Essay III. Outside Options and the Sharing of Match-Specific Rents

Co-authored with Peter Fredriksson, Lena Hensvik, and Oskar Nordström Skans

Acknowledgements: We thank seminar participants at EALE Padua 2021, IFAU, BI Norwegian Business School, and SOLE Minneapolis 2022 for comments and discussions.

1 Introduction

Wage inequality is shaped by how labor markets operate and how wages are set. Equally skilled workers earn different wages across workplaces because of imperfect competition and frictions. Furthermore, on-the-job search can generate dispersion across equally productive workers in the same job, and make within-job wage inequality responsive to the aggregate cycle. But the impact of on-the-job search on wage inequality crucially depends on how wages respond to outside offers. If workers in productive matches can use outside offers to bid up their current wage, as in Postel-Vinay and Robin (2002), wellmatched workers will tend to extract a greater share of their match surplus when the state of the labor market improves. In canonical wage bargaining models (e.g. Pissarides, 2000), wages are instead set as weighted averages between match productivity and aggregate conditions with fixed weights. In this case, wages will increase with market tightness without affecting within-job dispersion. This empirical paper sheds light on this issue by providing direct evidence on how outside options affect the transmission from match-specific productivity to wages.

We derive a measure of job-level match quality using Swedish enlistment data on eight distinct skills, following Fredriksson et al. (2018). Consistent with most wage bargaining models, well-matched workers earn higher wages. But what role do outside options play? Our stylized theoretical framework, building on Postel-Vinay and Robin (2002), predicts that, if outside offers are used in bargaining, then match quality for *job stayers* will matter more for wages when the aggregate arrival rate of offers is higher, i.e., when local labor market conditions are favorable. Moreover, for *job movers*, match quality in the previous job should matter for the current wage, but only if the transition was direct (i.e., without intermittent non-employment).

Our empirical results confirm these predictions. The wage returns to match quality are pro-cyclical for workers who remain in their jobs; The return to wages in levels increases by 50 percent of the average return when moving from the highest to the lowest ventile of the local unemployment distribution. The results are qualitatively similar for wages in logs and levels. They are independent of how we define the local labor market, and remain unaltered when we replace unemployment by other indicators of local labor market conditions, including shift-share instruments and occupation-specific employment.

For job-to-job movers, match quality in the previous job is more important for the current wage than current match quality. This holds even when conditioning on the previous wage. By contrast, prior match quality is unrelated to the new wage among workers with interim non-employment.

Our results imply that internal (match quality) and external (outside options) factors are not additively separable as in canonical bargaining models. Instead, workers' returns to match productivity increase when market conditions improve and there are more counteroffers, or when they move directly from a position where they were already well-matched as originally proposed by Postel-Vinay and Robin (2002).

Our main contribution is to offer direct evidence on how outside offers affect wages, an issue which previously exclusively has been studied with structural models. Earlier reduced-form studies (surveyed by Jäger et al., 2020) have analyzed how match-specific factors and outside options affect wages, but not how they interact. Fredriksson et al. (2018) study match quality and wages at the job-level whereas Guvenen et al. (2020) and Lise and Postel-Vinay (2020) use occupations. Studies of outside options analyze unemployment (Blanchflower and Oswald, 1994), outside firms (Lamadon et al., 2022), networks (Caldwell and Harmon, 2019), dual jobs (Lachowska et al., 2022), and benefits (Jäger et al., 2020).¹ Di Addario et al. (2022), study how the previous firm effect (as in Abowd et al., 1999) affect the new wages of job-to-job movers, and find very small effects.

Many papers have used structural approaches to compare wage-setting mechanisms in models with two-sided heterogeneity. Mortensen (2003) found support for wage bargaining over wage-posting in many market segments. Cahuc et al. (2006) embeds outside offers and traditional bargaining, finding more support for the former. Hagedorn and Manovskii (2013) argue that counteroffers do not add additional information once match quality (proxied by previous tightness) is accounted for. More recently, Bagger and Lentz (2019) and Yamaguchi (2010) allow for both endogenous search and human capital accumulation.²

Section 2 presents a theoretical framework. Section 3 describes data and measurement. Section 4 reports the results. Section 5 concludes.

2 Theoretical framework

We outline a stylized theoretical framework that highlights how and why counteroffers can provide an important link between outside options and returns to individual match productivity. The labor market is segmented by skill levels. To simplify, the value of unemployment, b, is homogeneous across workers within each skill segment. For ease of exposition, we focus on a single segment, and let b remain a constant.³ Productivity (p) differs across matches (worker-job pairs) since workers' have different skill bundles and jobs have

¹On Swedish data, see Carlsson et al. (2019) (unemployment), Carlsson et al. (2016) (outside firms), and Fredriksson and Söderström (2020) (benefits).

²A different set of studies have used surveys to directly measure how wages are set (e.g., Barron et al., 2006, Hall and Krueger, 2012, Brenzel et al., 2014). Barron et al. (2006) find that 44 percent of firms would consider making a counteroffer. Hall and Krueger (2012) notes that around 40 percent of workers bargained for a new job while maintaining the option to go back to their previous job.

³Our empirical analysis holds the main effects of skills constant.

different skill requirements. Workers search for jobs randomly and match quality is revealed after they meet.

Consider a two-period set-up. In period 1, unemployed workers meet a firm with probability λ , draw idiosyncratic match quality p from a differentiable distribution F(p), decide on whether to match, and bargain over the wage, w = w(b, p). In period 2, employed workers meet with other firms (again at rate λ), and draw match quality, p'. Since unemployed and employed workers draw matches at the same rate, the reservation wage (and reservation match quality) in period 1 is b.

Workers move if p' > p. When $b < p' \le p$, offers are instead used to bid up the wage in the current match *if* the worker can use the alternative offer in negotiations with the current employer. There are three scenarios when meeting a firm in period 2:

- 1. If $p' \leq b$, nothing happens and the worker retains w(b, p);
- 2. If $b < p' \le p$, the worker (potentially) renegotiates the wage to w(p', p); and
- 3. If p' > p, the worker moves and earns w(p, p').

Following Cahuc et al. (2006), we let wages be affected by Nash bargaining and by counteroffers. Without counteroffers, workers obtain a share η of the (flow) surplus. If employers can respond to counteroffers, wages are instead:

1. $w(b, p) - b = \eta(p - b)$ for $p' \le b$ (and entrants from unemployment);

2.
$$w(p', p) - p' = \eta(p - p')$$
 for $p' \in (b, p]$; and

3. $w(p,p') - p = \eta(p'-p)$ for p' > p.

With a slight change of notation, define p_1 and p_2 as the first and second most productive relationship the worker has found during his current employment spell. Then p_1 is match quality with the current employer and p_2 is the best outside offer. The wage-setting rule becomes:

$$w = w(p_1, p_2, b) = \eta p_1 + (1 - \eta) \max(b, p_2), \tag{1}$$

where $p_2 = 0$ if the worker has received no alternative wage offer. This rule is more general than it may appear. It is derived by Cai (2020) as the result of a strategic alternating bargaining game (of the Rubinstein (1982) type) when the risk of an exogenous break-up of the bargain is ignorable.

In our empirical work, we estimate separate wage regressions for stayers and movers. We begin by considering the wage for a stayer in the second period.

Define θ as the probability of receiving an outside offer that is higher than *b* conditional on remaining in the initial job:

$$\theta = \frac{\lambda(F(p) - F(b))}{1 - \lambda + \lambda F(p)}.$$
(2)

The expected wage for stayers may then be written in terms of the first-period wage, the expected wage conditional on receiving a useful outside offer, and θ :

$$\mathbb{E}[w \mid p' \leq p, p, \lambda] = w(p, b) + \theta \left(\mathbb{E}[w(p, p') \mid p' \leq p, p] - w(p, b) \right)$$
$$= \eta p + (1 - \eta)b + (1 - \eta)\theta \left(\mathbb{E}[p' \mid b \leq p' \leq p] - b \right),$$
(3)

where the second line uses (1).

To see how the expected second-period wage is influenced by initial match quality, we differentiate (3) with respect to p:

$$\frac{\partial \mathbb{E}[w \mid p' \leq p, p, \lambda]}{\partial p} = \eta + (1 - \eta) \left[\underbrace{\frac{\partial \theta}{\partial p} \left(\mathbb{E}(p'_s) - b \right)}_{\text{Effect due to}} + \underbrace{\theta}_{\text{effect due to}}^{\partial \mathbb{E}(p'_s)} \frac{\partial P}{\partial p} \right] \geq \eta,$$
(4)

where we have introduced the notation $\mathbb{E}(p'_s) = \mathbb{E}[p' \mid b \le p' \le p]$

Both θ and $\mathbb{E}(p'_s)$ are increasing in p. Hence, the derivative is larger than the direct effect of increasing match quality on shared surplus, captured by η . This implies that workers receive higher rents when outside offers can be used in bargaining. The difference in the effect of an increase in p on the expected wage between workers that did and did not receive outside offers equals:

$$\frac{\partial \mathbb{E}[w \mid p' \leq p, p, \lambda]}{\partial p} \Big|_{\lambda=1} - \frac{\partial \mathbb{E}[w \mid p' \leq p, p, \lambda]}{\partial p} \Big|_{\lambda=0}$$
$$= (1-\eta) \frac{F'(p)}{F(p)} \left[\frac{F(b)}{F(p)} \left(\mathbb{E}(p'_{s}) - b \right) + p - \mathbb{E}(p'_{s}) \right] \ge 0.$$
(5)

Thus, remaining workers with higher match quality on average benefit more from receiving outside offers.

How does the effect of initial match quality on expected second-period wage change when the arrival rate of offers increases? To determine this, (4) is differentiated with respect to λ :

$$\frac{\partial^{2}\mathbb{E}[w \mid p' \leq p, p, \lambda]}{\partial p \partial \lambda} = (1 - \eta) \left[\underbrace{\frac{\partial \theta}{\partial \lambda} \frac{\partial \mathbb{E}(p'_{s})}{\partial p}}_{\text{Direct interaction}} + \underbrace{\frac{\partial^{2} \theta}{\partial p \partial \lambda} \left(\mathbb{E}(p'_{s}) - b\right)}_{\text{Interaction effect on offer probability}} \right].$$
(6)

113

The direct interaction effect is clearly positive: Workers become more likely to receive useful offers, and, on average, such offers grow more valuable with the quality of the initial match. However, the derivative also depends on the interaction effect on the probability of receiving a useful outside offer. This term is in principle ambiguous in sign, essentially because probabilities are bounded.

To find a closed-form expression for (6), we first calculate the relevant derivatives:

$$\frac{\partial \theta}{\partial \lambda} = \frac{F(p) - F(b)}{(1 - \lambda + \lambda F(p))^2} \ge 0,$$

$$\frac{\partial \mathbb{E}(p'_s)}{\partial p} = \frac{F'(p)}{F(p) - F(b)} (p - \mathbb{E}(p'_s)) \ge 0,$$

$$\frac{\partial^2 \theta}{\partial p \partial \lambda} = \frac{F'(p)}{(1 - \lambda + \lambda F(p))^2} (1 - 2\theta).$$
 (7)

Inserting these into (6) and simplifying yields:

$$\frac{\partial^2 \mathbb{E}[w \mid p' \le p, p, \lambda]}{\partial p \partial \lambda} = \frac{(1 - \eta) F'(p)}{(1 - \lambda + \lambda F(p))^2} \Big[(p - \mathbb{E}(p'_s)) + (1 - 2\theta) (\mathbb{E}(p'_s) - b) \Big]$$
(8)

A sufficient condition for this expression to be positive is that $\theta \le 0.5$. This is true for all values of *p* if $\lambda(1 - F(b)) \le 0.5$. In other words, as long as the probability that the best matched worker can use an outside offer to renegotiate the wage is less than 50 percent, the sign of equation (8) is positive.

Of course, (8) can be positive under more general circumstances. To see this write $\mathbb{E}(p'_s)$ as a weighted average of p and b:

$$\mathbb{E}(p'_s) = \omega p + (1 - \omega)b, \tag{9}$$

where the weight ω depends on the shape of the density function on the (b, p) interval. Use (9) in (8):

$$\frac{\partial^2 \mathbb{E}[w \mid p' \le p, p, \lambda]}{\partial p \partial \lambda} = \frac{2(1-\eta)F'(p)(p-b)}{(1-\lambda+\lambda F(p))^2} \Big[\frac{1}{2} - \theta\omega\Big]. \tag{10}$$

Thus, $\theta \omega \leq 1/2$ implies that (8) is unambiguously positive. In a uniform distribution, for example, $\omega = 1/2$, and thus (8) is positive for all values of λ and *b*.

In the appendix, we explore which values of p, λ , and b lead to a nonpositive value of (8) under various distributional assumptions. We find that this almost never occurs. For example, when the distribution of match quality is log-normal, (8) is positive for all values of λ and b. When the distribution is normal, we find that (8) can become negative only if $\lambda > 0.9$ and F(b) < 0.05. Since this is an extreme case, it seems safe to surmise that an increase in the job-offer arrival rate will benefit well-matched workers more than other workers, i.e., that (8) is positive.

This prediction contrasts with implications of a standard Nash-bargaining sharing framework (e.g., Pissarides, 2000) where $w = \eta p + (1 - \eta)b(\lambda)$. In this case, the only relevant outside option is the aggregate state $(b(\lambda))$; and the wage effects of idiosyncratic productivity would remain constant if λ changes.

Our first empirical part studies whether the wage impact of idiosyncratic productivity is pro-cyclical among workers who remain in their job.

Now, consider the wage for a worker who has moved to a new job in period 2 because p' > p. Then, bargaining revolves around the quality of the old match, regardless of the arrival rates of outside offers:

$$\mathbb{E}\left[w \mid p' > p, \ p, \ p'\right] = \eta p' + (1 - \eta)p.$$
(11)

This wage equation has an extremely simple form. The key prediction is that previous match productivity matters for wages also in the new job. This is, again, different from the standard sharing framework where the outside option *at the market-level* would replace the previous match productivity, p, in equation (11).

Only workers that match with an alternative job before separating from their initial job can use p as an outside option. If leaving/quitting *before* finding alternative employment, the wage equation instead reverts back to w(p',b).

The second empirical part estimates the wage-impact of previous match quality for job-to-job (or EE) movers and for (ENE) movers with an interim non-employment spell.

To summarize, our empirical work tests three predictions:

Prediction 1 (Stayers)

For stayers, the wage return to match quality in the current job is increasing in the probability of obtaining an outside offer.

Prediction 2 (EE movers)

Match quality in the previous job has a positive impact on wages in the current job for workers who have made an employment-to-employment move.

Prediction 3 (ENE movers)

Match quality in the previous job has no impact on wages in the current job for workers who have moved to a new job with an intermittent spell of non-employment.

3 Data, variables and sample selection

3.1 Data sources and information

Our data are drawn from population-wide Swedish administrative registers. They include linked information on individuals, workplaces and firms, and basic individual characteristics such as age, education and municipality. Employment status in November is drawn from tax returns. Data cover 1985-2013. Employment status is used to infer labor market experience (years observed in employment, censored at 12).

The Wage Structure Statistics (WSS), covering 1997-2013, record hoursadjusted wages and three-digit occupations (SSYK-96, corresponding to ISCO-88) annually for half of the private and all public sector employees. Data cover all the workers in sampled firms. The sampling probability increases in firm size and all firms with 500 or more employees are sampled.

From the Swedish War archives, we add detailed measures of cognitive and non-cognitive ability. These were collected during the Swedish military draft procedure and are available for nearly all males in the 1951-1976 cohorts. These were enlisted during 1969-1994 at age 18 or 19.

Finally, we collect aggregate data on unemployment from Statistics Sweden and the Swedish Public Employment Service.

3.2 Variables and definitions

The definition of a job

We focus on match quality at the worker-job level. A *job* is an occupation at a workplace.⁴ This allows skill requirements to vary between different occupations in a given workplace, as well as between different workplaces for a given occupation.⁵

Local unemployment

Unemployment is measured at the workplace municipality level.⁶ It is computed from individual-level information on non-employed job seekers (including participants in active labor market programs) registered at the public employment service. We define *unemployment* as the annual incidence of registered job-seekers among all residents aged 20-64 (i.e., not just labor force participants). Our results remain robust if we use other measures (see the appendix).

⁴Data cover 113 occupations. Our used sample cover 30,000 workplaces and 70,000 jobs.

⁵Sorting across both margins appear empirically relevant, see, e.g., Fredriksson et al. (2018) and Choné et al. (2022).

⁶Sweden has 10 million residents, 290 municipalities and 21 counties.

Multidimensional skills

Mood et al. (2012) and Lindqvist and Vestman (2011) provide extensive details on the test scores. Four skills are based on written tests. These capture *inductive reasoning, verbal comprehension, spatial ability* and *technical understanding* (jointly *cognitive ability*). The remaining four are assessed during a 25-minute interview with a trained psychologist. These capture *social maturity, psychological energy, intensity,* and *emotional stability* according to Mood et al. (2012) (jointly *non-cognitive ability*). We standardize the data to mean zero and standard deviation one for each skill and cohort.

For our purposes, a key aspect of these scores is that workers with similar types of skills tend to be clustered into the same jobs as shown by, e.g., Fredriksson et al. (2018). Workers systematically sort into jobs where the return to their particular skill endowments are higher than in the average job.⁷

Match quality

Our measure of job-specific match quality builds on insights from Fredriksson et al. (2018). The objective is to compare the skill set of a worker to the skill requirements of the job. We use other workers who remain within the same job to measure skill requirements. The presumption is that workers who remain within a job are well matched, i.e., their skill set is likely to be well aligned with the skill requirements of the job. We do not use coworkers who remain for less than three years when calculating the skill requirements as these are likely to be poorly matched.

For each worker *i* in job *j*, we calculate the average skill level along a particular dimension *k* among other, tenured, incumbents in *j*. We denote this average by $\tilde{s}_{j,k}^{-i}$ where "-i" represents all individuals except *i*. In contrast to Fredriksson et al. (2018), we keep $\tilde{s}_{j,k}^{-i}$ constant across time within jobs to ensure that the skill-requirement is not endogenous to the cycle.

For each skill-dimension k, we calculate the absolute difference between individual skills $s_{i,k}$ and $\tilde{s}_{j,k}^{-i}$, and then sum over the eight dimensions. We let $D_{i,i}$ denote worker *i*'s deviation from the skills of tenured co-workers in job *j*:

$$D_{i,j} = \sum_{k=1}^{8} \left| s_{i,k} - \tilde{s}_{-i,j,k} \right|$$

For ease of interpretation we construct our measure of match quality (M) as the negative of the standardized distance D. This implies that a positive oneunit change in our match quality measure is equivalent to a standard deviation reduction in the distance to the skills of the coworkers:

⁷Fredriksson et al. (2018) also estimate significant wage returns to each of the different skills, even conditional on education.

$$M_{i,j} = -\left(\frac{D_{i,j} - \operatorname{mean}\left(D_{i,j}\right)}{\operatorname{sd}(D_{i,j})}\right)$$

To assess the wage impact of match quality, net of constant skill differences across workers and jobs, we always analyze $M_{i,j}$ conditional on controls for worker skills and job fixed effects.

The returns to skills

To understand our measure of match quality, consider a generic version of our wage regressions:

$$w_{i,j} = \alpha_j + \boldsymbol{\beta}' \boldsymbol{s}_i + \mu M_{i,j} + \varepsilon_{i,j}$$
(12)

where α_j is a job-specific fixed effects and s_i is a vector comprising all individual skills. A marginal increase in $s_{i,k}$ will have an effect through the market (via β_k) and an effect from job-match quality (through μ). Assume that *market returns* to skills are positive ($\beta_k > 0 \forall k$) and that wages are increasing in match quality ($\mu > 0$).⁸

Match quality can either become better or worse if we increase $s_{i,k}$. Match quality will increase if the worker is underskilled at job *j* in dimension *k* (i.e. if $s_{i,k} < \tilde{s}_{j,k}^{-i}$) which will generate an additional wage gain. But match quality will deteriorate if worker *i* is overskilled, i.e. if $s_{i,k} > \tilde{s}_{j,k}^{-i}$ and we add further skills. The overall wage impact (market returns + match-quality effects) of additional skills for overskilled workers is therefore lower than the market returns.

The key takeaway is that the model allows the overall wage return to additional skills to *kink* at the point of optimal match quality (i.e., at the point where $s_{i,k} = \tilde{s}_{i,k}^{-i}$).

3.3 Sample and descriptive statistics

Our sample consists of workers who were *i*) employed in year *t*, *ii*) sampled in the WSS, *iii*) Swedish residents during [t - 1, t + 1] and *iv*) in our skills data. We use data for 1997-2012 (to allow for a 1 year follow-up). We only use jobs with at least five tenured coworkers to get a reasonable measure of average skills within jobs.

Column (1) in Table 1 reports means and standard deviations in the used sample. For comparison, column (2), describes all workers in the WSS. The differences between columns (which are mainly driven by firm size) arise because we require jobs to have at least five male workers with skills within each

⁸For the general returns to skills, see Fredriksson et al. (2018). For the returns to match quality, see the next section.

job. The larger firm size makes both job stability and wages somewhat higher than in the overall sample. Reassuringly (for external validity), the distribution of occupations is similar across columns.

4 Empirical analysis

4.1 The overall impact of match quality and unemployment

We first show how match quality relates to wages and the probability of leaving the initial workplace between year t and t + 1 (separations). We then relate local unemployment to wages and to the probability of doing a job-to-job move (among all separations). For wages, we use unemployment during the observed year t. For separations, we instead use unemployment in t + 1 since this better reflects the applicable market conditions between November in years t and t + 1. We illustrate the patterns graphically after removing the impact of key controls.⁹

The top left panel of Figure 1 shows that there is a strong positive relationship between the match quality of a worker and his wage. Moving from the 10th to the 90th percentile of the (residualized) match-quality distribution is associated with approximately two percent higher wages. Similarly, a well-matched worker has a significantly lower risk of separating from his current workplace than poorly matched co-workers. The separation probability is 0.5 percentage points lower for a worker on the 90th percentile compared to a worker on the 10th percentile of the match quality distribution (the average probability is approximately 10 percent). The two lower panels show the relationship between wages and the probability of separating to another employer, respectively, and unemployment. Local unemployment is associated with lower wages—wages are 0.6 percent lower at the 90th percentile than at the 10th percentile of the unemployment distribution-and less employmentto-employment separations—an increase from the 10th to the 90th percentile of the unemployment distribution reduces the job-to-job share by 0.6 percentage points.

4.2 Outside options and the sharing of rents with remaining workers

This section examines Prediction 1 of Section 2. We thus ask how the returns to workers' match quality vary with local unemployment (our proxy for the arrival rate of outside offers). Guided by the theory, we only include workplace

⁹We residualize wages, separations, match quality, and unemployment using: A 2nd order polynomial in each skill, age, education, and experience fixed effects (FEs). In addition, the upper two panels of Figure 1 control for job-by-year FEs (making local unemployment redundant); the lower two panels use additive job and year FEs.

	Main sample A	ll male workers in WSS
	(1)	(2)
Probability of separation	.096	.137
log wage	10.181 (.363)	10.127 (.357)
Age	41.684 (8.285)	42.868 (11.721)
Years of schooling	12.477 (2.399)	12.37 (2.454)
Establishment size	534.1 (1035.7)	410.5 (948.0)
Years of workplace tenure		
—1 to 3	.253	.345
—4 to 6	.193	.188
—7 to 9	.139	.121
—10 or more	.415	.346
Occupation category (one-digit level)		
-Legislators, senior officials, and managers	.068	.084
-Professionals	.229	.214
—Technicians and associate professionals	.216	.204
-Clerks	.05	.062
-Service workers and shop sales workers	.053	.084
-Skilled agricultural and fishery workers	.005	.007
-Craft and related trades workers	.127	.119
-Plant machine operators and assemblers	.215	.172
-Elementary occupations	.038	.054
N	4,315,746	13,359,613

Table 1. Characteristics of the main sample and all workers sampled in the WSS

Notes: Column (1) reports the mean and (when relevant) the standard deviation of key variables in the main sample. For comparison, column (2) reports descriptive statistics for all male workers for which we observe wages and occupation in the Wage Structure Statistics (WSS).



Figure 1. How wages and separations relate to match quality and unemployment

Notes: Panels (a) and (b) relate log wages and the probability of separating from the current workplace between t and t + 1, respectively, to match quality. Panel (c) plots log wages against unemployment, while panel (d) relates the probability of being employed in t + 1, for workers that left the workplace between t and t + 1, to unemployment in t + 1. We have residualized all variables with respect to a 2nd order polynomial in each of the eight included skills, age, as well as education and experience fixed effects (FEs). In addition, panels (a) and (b) control for job-by-year FEs; panels (c) and (d) hold job and year FEs constant.

stayers (at least one year of tenure), but this is not crucial for the results. Job movers are analyzed in the next section.

We regress wages (w) on match quality (M), local unemployment (u), and the interaction between the two, while accounting for job fixed effects and the direct impact of skills. More specifically, we estimate different versions of the following model:

$$w_{i,j,l,t} = \alpha_{j,t} + \phi \, u_{l,t} + \delta M_{i,j} + \mu \, (u_{l,t} \times M_{i,j}) + g(\mathbf{s}_i) + u_{l,t} \times h(\mathbf{s}_i) + \gamma' \mathbf{x}_{i,t} + \varepsilon_{i,j,l,t},$$
(13)

where *i*, *j*, *l* and *t* indexes individual, job, local labor market (municipality) and year, respectively. We include fixed effects $(\alpha_{j,t})$ for each job (separately by year in the tightest specification), which implies that we only identify the return to match quality from within job variation. $g(\mathbf{s}_i)$ and $h(\mathbf{s}_i)$ are second-order polynomials in each of the eight skill measures. we control flexibly for skills to account for changes in market-wide returns to skills (through $g(\mathbf{s}_i)$) and variation across the business cycle in skill returns (through $u_{l,t} \times h(\mathbf{s}_i)$). Finally, $\mathbf{x}_{i,t}$ includes a linear control for age, as well as experience and education fixed effects.

For testing Prediction 1, μ is the key parameter of interest. We expect it to be negative such that workers receive a smaller share of the surplus associated with match quality when the arrival rate of job offers is low (i.e., when unemployment is high).¹⁰

The first three columns of Table 2 report the results from estimating different versions of equation (13) using log wages as the outcome. Column (1) uses additive job and year fixed effects, for which the main effect of unemployment is identified. As previously shown, wages are on average lower when local unemployment is high. Moreover, there is a positive (and precisely estimated) return to match quality.

Crucially, the estimates on the interaction between match quality and unemployment are all negative (and precise). When local unemployment falls, the return to a good match thus grows. The magnitudes are non-trivial. For ease of interpretation, Figure 2 (panel a), *inter alia*, plots the estimated return to match quality (from column 2) along the distribution of local unemployment. The estimates suggest that if unemployment falls from the top to the bottom end of the distribution, the return to a standard deviation increase in match quality rises by 0.275 percent. This increase corresponds to 20 percent (= 0.275/1.38) of the average return to match quality.

A possible concern is that the interaction between match quality and unemployment picks up national trends, or time-invariant differences across munic-

¹⁰To facilitate interpretation of the estimates, we demean unemployment allowing us to interpret the estimates for δ and ϕ as the average effect of match quality and unemployment, respectively. Note, that ϕ is not identified when job-by-year fixed effects are included in the regression.

ipalities, in the returns to match quality that are not caused by unemployment. To address such issues, we interact $M_{i,j}$ with year as well as municipality fixed effects. These interactions imply that the model only relies on variation within municipalities over time when identifying μ . Reassuringly, the estimate on the interaction term is robust to allowing the returns to match quality to vary across municipality and years (see column (3)). If anything, the coefficient on the interaction term increases in absolute size.

In addition to using log wages as the outcome variable—the standard practice in reduced-form empirical work—equation (13) is also estimated using wage levels—in better agreement with the standard theoretical formulation. To make the wage measure comparable across time, and for ease of interpretation, we standardize it (mean = 0, standard deviation = 1) separately by year. Columns (4) to (6) of Table 2 report the results. The results are qualitatively similar to those for log wages, and, thus the results do not depend on functional form. But the interaction effect is more important relative to the average return to match quality in columns (4)-(6) than in columns (1)-(3). Figure 2 (panel b) illustrates this point by showing the estimated return to match quality (from column 5) along the distribution of local unemployment. If unemployment falls from the top to the bottom end of the distribution, the return to an increase in match quality rises by 2.3 percentage points relative to the standard deviation in wages. This increase corresponds to 52 percent (=2.3/4.42) of the average return to match quality.

Figure 2 also provides non-parametric estimates of the return to match quality. Here, we group observations into 20 bins based on the ventiles of the unemployment distribution. The return to match quality is then estimated using a version of equation (13) which, instead of interacting match quality with unemployment, interacts match quality with the dummies for each ventile bin. The non-parametric estimates are well aligned with the predicted returns from the main specification (which we take to be columns 2 and 5 in Table 2).

In the appendix, we report a large number of robustness checks for the relationships presented in this section. Among other things, we vary how standard errors are calculated, use alternative definitions of local labor markets, use alternative proxies for the arrival rate of offers (local employment instrumented by a shift-share instrument, occupation-specific local employment, labor market tightness), relate the return to match quality to the average unemployment during workers' tenure spells, give each municipalities equal weight irrespective of size, and incorporate individual \times job fixed effects in our main specification to hold fixed time-invariant match-specific non-observables. Our results are robust across all these specifications.

		Log wage		Sta	ndardized wa	ge
	(1)	(2)	(3)	(4)	(5)	(9)
Match quality $\times \log$ unemployment	-0.00534 (0.000488)	-0.00264 (0.000497)	-0.00393 (0.00112)	-0.00997 (0.00194)	-0.0217 (0.00194)	-0.0332 (0.00378)
Match quality	0.0134 (0.000136)	0.0138 (0.000135)		0.0472 (0.000545)	0.0442 (0.000533)	
log unemployment	-0.0455 (0.00123)			-0.0579 (0.00447)		
Year fixed effects Tob fixed effects	`			>>		
Job \times year fixed effects	•	>	>	•	>	>
$M \times$ municipality interactions			>			>
$M \times$ year interactions			~			~
N	3,938,598	3,938,598	3,938,598	3,938,598	3 938 598	3 938 598
\mathbb{R}^2	0.806	0.834	0.834	0.604	0.661	0.661

education and experience fixed effects as well as age (linearly) and, for each of the eight skills, a second-order polynomial for the skill level of individual i. All Notes: The table reports versions of the model represented by equation (13). In addition to the control variables listed in the table, all regressions control for models also include interactions between each of the individual skill polynomials and unemployment. Standard errors are heteroskedasticity-robust.

 Table 2. Wage regressions for job stayers

4.3 Match quality in the previous job and the wages of job movers

One key prediction from multilateral wage bargaining is that workers who voluntary move to a new job can use the old job as a distinct outside option in order to bid up the wage in the new job. Thus, the wage in the new job should be a function of match quality in the previous job; see Prediction 2 of Section 2. Moreover, if there is some additional sharing motives, i.e., if the new firm shares more of the surplus than what is required to outbid the old, competing, firm, the quality of the new match should also influence starting wages. The difference between the effects of previous and current match quality, which are measured in the same way, will indicate the relative importance of counteroffers versus generic sharing motives.

However, previous match quality should only be useful for workers whose previous jobs remain available. For workers that are laid off, or quit before finding a new job, we therefore expect previous match quality to play no role in bargaining; see Prediction 3 of Section 2.

To test these predictions, we identify all workers who switched to a new firm between t - 1 and t.¹¹ We refer to this set of workers as job-to-job (or EE) movers. We also identify employed workers in t who were classified as non-employed in t - 1—and were thus likely hired from non-employment—for whom we observe match quality and wages in t - 2.¹² We refer to these workers as movers with an interim non-employment spell (or ENE movers).

For the two groups, $r \in \{EE, ENE\}$, we estimate joint wage regressions with controls for current and previous match quality interacted with indicators for the two groups. The regressions also include the same controls for skills and individual characteristics as the regressions in Table 2, as well as year fixed effects. In addition, for each of our eight skill dimensions, we include a second-order polynomial in the average skill level of the coworkers—for the new job $(g_{j'}^m(\tilde{\mathbf{s}}_{j'}^{-i}))$ as well as the previous job $(g_j^m(\tilde{\mathbf{s}}_j^{-i}))$.¹³ We thus estimate different versions of the following model:

$$w_{i,j',j,t} = \boldsymbol{\rho}_t + \omega I_{i,t}^{EE} + \sum_{r \in \{EE, ENE\}} I_{i,t}^r \left(\boldsymbol{\kappa}^r M_{i,j'} + \boldsymbol{\nu}^r M_{i,j} \right) + g^m(\boldsymbol{s}_i) + \boldsymbol{\psi}' \boldsymbol{x}_{it} + g_{j'}^m(\boldsymbol{\tilde{s}}_{j'}^{-i}) + g_j^m(\boldsymbol{\tilde{s}}_{j}^{-i}) + \boldsymbol{\varepsilon}_{i,j',j,t}, \qquad (14)$$

where j' and j indexes the current and previous job, respectively. $I_{i,t}^r$ is an indicator for belonging to group r. We restrict the background controls to have

¹¹In addition, we require that they have no previous experience from the new workplace, and that we observe match quality and wages in both periods.

¹²We also require that they did not return to their old firm in t.

¹³This is a parsimonious way of controlling for the features of the new and the previous job. In the appendix, we show that the results are robust to instead using unrestricted fixed effects for the old and the new firm.



Figure 2. The return to match quality along the unemployment distribution

Notes: Non-parametric estimates are obtained by grouping data into 20 bins based on the ventiles of the distribution of $u_{j,t}$. The return to match quality is then estimated using a version of equation (13) which interacts match quality with the indicators for each bin. Panel (a) shows the estimates, and the 95-percent confidence intervals based on robust standard errors, for log wages while (b) presents the same information for wage levels standardized (mean = 0, standard deviation = 1) separately by year. The solid lines show the predicted return to match quality based on the estimates in columns (2) and (5) from Table 2.

the same impact across the two groups. ω corresponds to the average wage premium of being hired from employment compared to non-employment; κ^r and ν^r informs us about the importance of new and previous match quality for group *r*.

The results are reported in Table 3. Columns (1)-(3) report results for log wages while the results for standardized wages are reported in columns (4)-(6). In columns (1) and (4), we start by relating wages to current match quality. We then introduce previous match quality in columns (2) and (5). Note, we have 58,053 workers which we classify as EE movers, while (only) 1,784 workers are classified as ENE movers.

The estimated return to match quality for the group of EE movers is large and statistically significant. When we add previous match quality to the regression, the estimated effect of current match quality approximately halves in size. Through the lens of the model, this can be interpreted as current match surplus being shared between workers and firms. The implied estimate of the sharing parameter (η of Section 2) is 0.41.¹⁴ Most importantly, there is a significant and even stronger relationship between previous match quality and the current wage, conditional on current match quality. This strongly suggests that counteroffers from the old employer play an important role in wage setting.

On average, job-to-job movers have a substantially higher wage in their new job compared to movers with interim non-employment (see the coefficient on "EE mover"). For ENE movers, current match quality appears to be important. In relation to Prediction 3, we note that previous match quality for the ENE group appear irrelevant; the estimate for ENE × previous match quality—i.e., V^{ENE} —is virtually zero (see column 2).¹⁵

The impact of the previous match quality may, in principle, be mediated by the previous wage. To see what role previous match quality plays, conditional on previous wages, columns (3) and (6) report results from an augmented version of equation (14) where we introduce group-specific effects of the wage in the previous job. These specifications naturally reduce the impact of match quality, since wages depend on match productivity. Nevertheless, previous match quality is still significant for EE movers.

¹⁴This is obtained as 0.0112/(0.0112+0.0164), where the normalization takes into account that we only observe a proxy for match quality rather than match productivity. In columns (3), (5), and (6), an analogous exercise produces estimates of η equaling 0.26, 0.44, and 0.40, respectively.

¹⁵In the appendix, we report estimates from separate regressions for the two groups. Overall, the results are qualitatively robust, with one exception: Current match quality does not seem to be as important as indicated by the pooled regression. With that said, one should keep in mind that there are only 1,784 workers in the ENE group.

		Log wage		Standardized wage		
	(1)	(2)	(3)	(4)	(5)	(6)
EE mover	0.130 (0.00544)	0.130 (0.00548)	0.115 (0.00864)	0.240 (0.0129)	0.240 (0.0130)	0.250 (0.0408)
$EE \times new$ match quality	0.0201 (0.00163)	0.0112 (0.00198)	0.00109 (0.00129)	0.0657 (0.00517)	0.0395 (0.00614)	0.00793 (0.00401)
$ENE \times new$ match quality	0.00908 (0.00492)	0.0144 (0.00683)	0.00303 (0.00554)	0.0434 (0.0115)	0.0420 (0.0161)	0.0105 (0.0112)
$EE \times prev.$ match quality		0.0164 (0.00191)	0.00305 (0.00127)		0.0493 (0.00596)	0.0117 (0.00399)
ENE \times prev. match quality		-0.00121 (0.00673)	0.00102 (0.00559)		0.0201 (0.0159)	0.00527 (0.0110)
$EE \times prev.$ wage			0.773 (0.00425)			0.745 (0.0187)
ENE × prev. wage			0.455 (0.0219)			0.420 (0.0565)
N R ²	59,837 0.612	59,837 0.612	59,837 0.844	59,837 0.367	59,837 0.368	59,837 0.740

Table 3. Wage regressions for job movers

Notes: "EE movers" are workers who changed firms between t - 1 and t. "ENE movers" are workers who changed firms between t - 2 and t with an intermittent spell of non-employment in November t - 1. We also require that we observe wages and match quality in the current and previous job. All regressions include controls for education and experience fixed effects, a linear control for age, and a second-order polynomial for each of the eight skills. In addition, we control for a second-order polynomial for the average in each skills among tenured workers in the previous as well as current job. The sample includes 58,053 EE movers, and 1,784 ENE movers. Robust standard errors in parentheses.

5 Conclusions

We offer two pieces of evidence that workers can use outside offers to extract rents from match productivity. Using a match quality measure based on the alignment between workers' multidimensional abilities and the skill requirements of their jobs, we show that: (i) Wages within ongoing matches are more closely aligned with match quality following an improvement of local labor market conditions; (ii) Wages of job movers are positively related to the match quality in the previous job, even when controlling for previous wage, while wages of workers who are hired after a non-employment spell are unrelated to the match quality in the last job.

These finding run counter to the standard wage bargaining framework, as well as the typical reduced form rent-sharing set-up, which are based on the assumption that the impact of idiosyncratic rents and outside options are additively separable in the wage equation.

Our findings have clear implications for wage inequality: When firms are prepared to make counteroffers, there will be wage dispersion among workers who are equally productive within a given match. Moreover, since the wage returns to match quality grow larger, within-firm wage inequality due to variation in match productivity among incumbent workers increases when the state of the labor market improves.

References

- Abowd, J. M., F. Kramarz, and D. N. Margolis (1999). "High wage workers and high wage firms". *Econometrica* 67.2, 251–333.
- Bagger, J. and R. Lentz (2019). "An empirical model of wage dispersion with sorting". *Review of Economic Studies* 86, 153–190.
- Barron, J., M. Berger, and D. Black (2006). "Selective counteroffers". *Journal* of Labor Economics 24.3, 385–409.
- Bartik, T. J. (1991). "Who benefits from state and local economic development policies?"
- Blanchflower, D. G. and A. J. Oswald (1994). The Wage Curve. MIT Press.
- Brenzel, H., H. Gartner, and C. Schnabel (2014). "Wage bargaining or wage posting? Evidence from the employer side". *Labour Economics* 29, 41–48.
- Cahuc, P., F. Postel-Vinay, and J.-M. Robin (2006). "Wage bargaining with on-the-job search: Theory and evidence". *Econometrica* 74.2, 323–364.
- Cai, X. (2020). "Efficiency of wage bargaining with on-the-job search". *International Economic Review* 61.4, 1749–1775.
- Caldwell, S. and N. Harmon (2019). "Outside options, bargaining, and wages: Evidence from coworker networks". *Mimeo*.
- Carlsson, M., J. Messina, and O. N. Skans (2016). "Wage adjustment and productivity shocks". *The Economic Journal* 126.595, 1739–1773.
- Carlsson, M., I. H. Skans, and O. N. Skans (2019). "Wage Flexibility in a Unionized Economy with Stable Wage Dispersion". *IZA Discussion Paper Series* 12093.
- Choné, P., F. Kramarz, and O. N. Skans (2022). "When workers' skills become unbundled: Some empirical consequences for sorting and wages". *Working Paper, IFAU*.
- Di Addario, S., P. Kline, R. Saggio, and M. Sølvsten (2022). "It ain't where you're from, it's where you're at: Hiring origins, firm heterogeneity, and wages". *Journal of Economometrics*, forthcoming.
- Faberman, R. J., A. I. Mueller, A. Şahin, and G. Topa (2022). "Job Search Behavior Among the Employed and Non-Employed". *Econometrica* 90.4, 1743–1779.
- Fredriksson, P., L. Hensvik, and O. N. Skans (2018). "Mismatch of talent: Evidence on match quality, entry wages, and job mobility". *American Economic Review* 108.11, 3303–38.
- Fredriksson, P. and M. Söderström (2020). "The Equilibrium Impact of Unemployment Insurance on Unemployment: Evidence from a Non-linear Policy Rule". *Journal of Public Economics* 187.

- Guvenen, F., B. Kuruscu, S. Tanaka, and D. Wiczer (2020). "Multidimensional skill mismatch". *American Economic Journal: Macroeconomics* 12.1, 210–44.
- Hagedorn, M. and I. Manovskii (2013). "Job Selection and Wages over the Business Cycle". American Economic Review 103.2, 771–803.
- Hall, R. and A. Krueger (2012). "Evidence on the incidence of wage posting, wage bargaining and on-the-job search". *American Economic Journal: Macroeconomics* 4.4, 56–67.
- Hornstein, A., P. Krusell, and G. L. Violante (2011). "Frictional wage dispersion in search models: A quantitative assessment". *American Economic Review* 101.7, 2873–2898.
- Jäger, S., B. Schoefer, S. Young, and J. Zweimüller (2020). "Wages and the Value of Nonemployment". *The Quarterly Journal of Economics* 135.4, 1905–1963.
- Lachowska, M., A. Mas, R. Saggio, and S. A. Woodbury (2022). "Wage posting or wage bargaining? A test using dual jobholders". *Journal of Labor Economics* 40.S1, S469–S493.
- Lamadon, T., M. Mogstad, and B. Setzler (2022). "Imperfect Competition, Compensating Differentials, and Rent Sharing in the US Labor Market". *American Economic Review* 112.1, 169–212.
- Lindqvist, E. and R. Vestman (2011). "The labor market returns to cognitive and noncognitive ability: Evidence from the Swedish enlistment". *American Economic Journal: Applied Economics* 3.1, 101–28.
- Lise, J. and F. Postel-Vinay (2020). "Multidimensional skills, sorting, and human capital accumulation". *American Economic Review* 110.8, 2328–76.
- Mood, C., J. O. Jonsson, and E. Bihagen (2012). "Socioeconomic persistence across generations: cognitive and noncognitive processes". *From parents to children: the intergenerational transmission of advantage*, 53–84.
- Mortensen, D. (2003). *Wage dispersion: Why are similar workers paid differently?* MIT Press.
- Pissarides, C. (2000). Equilibrium Unemployment Theory. MIT Press.
- Postel-Vinay, F. and J.-M. Robin (2002). "Equilibrium wage dispersion with worker and employer heterogeneity". *Econometrica* 70.6, 2295–2350.
- Rubinstein, A. (1982). "Perfect Equilibrium in a bargaining model". Econometrica 50.1, 97–109.
- Yamaguchi, S. (2010). "Job search, bargaining, and wage dynamics". *Journal* of Labor Economics 28.3, 595–631.

Appendix A Simulations

This section analyzes the derivative of the expected wage for job stayers with respect to the arrival rate of offers and match quality, i.e., the sign of:

$$\frac{\partial^2 \mathbb{E}[w \mid p' \le p, p, \lambda]}{\partial p \partial \lambda} = \frac{(1 - \eta) F'(p)}{(1 - \lambda + \lambda F(p))^2} \Big[(p - \mathbb{E}(p'_s)) + (1 - 2\theta) (\mathbb{E}(p'_s) - b) \Big]$$
(A1)

More specifically, we investigate which distributional assumptions, as well as values of λ , *b* and *p*, that lead to a negative sign of equation (A1).

To this end, we simulate the model in Section 2 using three different distributions for match quality: The Normal(5, 1); The Log-Normal(0, 1); And the Uniform(0, 1). For each distribution, the simulation considers values of $\lambda, F(b) \in \{0.01, 0.02, \dots, 0.99\}$.

For all possible values of λ , F(b), we randomly generate a value of p_i for 10 000 individuals indexed by *i*, discarding those with $p_i < b$. Each individual is then given the chance of drawing an outside offer in the second period. More specifically, we calculate the best outside option as $\max\{b, -\alpha + B(\lambda) \times (p'_i + \alpha)\}$, where the Bernoulli trial $B(\lambda)$ determines whether the worker matched with a new firm, p'_i is the random value of that match, and α is a large value such that $-\alpha < b$. Observations with outside options outside the range $[b, p_i]$ are removed, i.e., we only retain job stayers.

We proceed by calculating $F(p_i)$, $F'(p_i)$ and $\mathbb{E}(p'_{si}) = \mathbb{E}[p'_i | b < p'_i < p_i, p_i]$ for all individuals. These values are then plugged into (A1). Finally, for all combinations of λ , F(b), we calculate the average derivative over the simulated p_i distribution for stayers. This average is a close approximation of the empirical estimate on the interaction between the arrival rate and match quality in the wage regression for stayers.

For both the uniform and the lognormal distributions, equation (A1) is positive for all values of λ , F(b), and F(p). In fact, this can be proven analytically for the uniform distribution, as briefly discussed in Section 2. Moreover, the right skewness of the lognormal distribution typically causes $\mathbb{E}[p'_i | b < p'_i < p_i, p_i]$ to be closer to *b* than to p_i . In turn, this implies that (A1) is positive as also noted in Section 2. It is only for the normal distribution that (A1) can turn negative. But this only happens at extreme values of the arrival rate—values which are not observed empirically (see Faberman et al., 2022, and Hornstein et al., 2011)¹⁶—and low levels of *b*.

The dark (light) squares in Figure A1 show combinations of F(b) and λ where (A1) is negative (positive) when averaged over the distribution of p.

¹⁶Faberman et al. (2022)'s analysis suggests a monthly arrival rate for employed workers of 0.081. The probability of receiving at least one offer in a year can then be calculated as $(1 - (1 - 0.081)^{12})$. Thus, converted to an annual frequency, this implies $\lambda = 0.637$.

Figure A2 presents the analogous information for different decile groups of the simulated $F(p_i)$ distribution.

Figure A1 shows that a negative sign of (A1) requires $\lambda \ge 0.88$ when F(b) = 0.01. With a slight increase in b to F(b) = 0.05, the requirement is $\lambda \ge 0.97$.

Figure A2 shows that the middle deciles of the F(p) distribution drive the average illustrated in Figure A1. For decile groups with relatively low and high match quality, however, a negative value of (A1) is associated with even more extreme values of λ and F(b).

Figure A1. Sign of equation (A1), averaged over the distribution of p, for different values of λ and F(b)



Notes: The figure reports the sign of the second-order derivative of the expected wage with respect to match quality (*p*) and the arrival rate (λ) by *F*(*b*) and λ , averaged over the distribution of *p*. Dark (light) squares indicate a negative (positive) derivative.



Figure A2. Sign of equation (A1) for different values of λ and F(b), by match quality decile groups



135

Appendix B Robustness checks for remaining workers

This section describes a battery of robustness checks, the majority of which are reported in Table B1, in order to determine the sensitivity of the results reported in Section 4.2. Among other things, we produce different standard errors, vary at which level local unemployment is measured, introduce alternative proxies for local labor market tightness/expected outside option relevant for each individual, and vary the controls and fixed effects that are included in the main specification (see equation (13)). If not stated otherwise, all regressions include the same controls as columns (2) and (5) in Table 2, i.e., job \times year fixed effects as well as the controls that are shared across all models.

Table B1 reports the estimate and standard error for the interaction between match quality and applicable proxy for workers' expected outside options (the equivalent of μ from equation (13)). The table also reports the difference between the 90th and 10th percentile of the outside option proxy distribution, and for ease of interpretation and comparison, the estimated difference in match quality returns between these points, i.e., $\hat{\mu} \times (P90 - P10)$.

Alternative definitions of local labor markets

The main analysis utilizes workplace municipality unemployment. Measuring unemployment at a more aggregate level does not call into question our main results; if anything, the heterogeneity in returns to mismatch with respect to unemployment is marginally larger when using unemployment in the workplace local labor market or county. Moreover, using the annual unemployment incidence in the residential municipality of each worker, which may vary within each job \times year cell, yields very similar results.¹⁷

Actual and instrumented municipality employment

We've also substituted unemployment for the logarithm of the number of employed workers in the workplace municipality. Naturally, employment is expected to have the opposite influence on the return to match quality as unemployment.

Next, we construct a plausibly exogenous measure of local labor market conditions by instrumenting log employment using a shift-share/Bartik (1991) style instrument. The instrument is based on the national deviation in industry s and year t from the industry average logarithm of employment. The employment in each municipality and year is then predicted based on the local industry composition in 1997 using the following formula:

$$\ln(\text{employment})_{l,t}^{Instr.} = \sum_{s} \text{Share}_{l,s,1997} \times \left(\ln(\text{employment})_{s,t} - \overline{\ln(\text{employment})}_{s}\right), \quad (B1)$$

¹⁷The main effect of residential municipality unemployment on wages, which is not reported in any table, is also both statistically and economically significant, which suggests that crossmunicipality-border commuters may be influenced by their residential municipality conditions.

where $\text{Share}_{l,s,1997}$ is the fraction of all workers in municipality *l* in 1997 that were employed in industry *s*. We use the highest level of aggregation in the Swedish Standard Industrial Classification, which includes 21 different industry categories.

Regressing standardized employment, $\ln(\text{employment})_{l,t} - \overline{\ln(\text{employment})}_l$, onto the instrumented standardized log employment using 4913 observations at the municipality-year level returns a coefficient of around 0.9 with a *t*-value of 64, and so the first stage equation is sufficiently precise. Instead of performing the full two-stage least squares analysis, we settle for the reduced form, i.e., we directly substitute unemployment with the instrument in equation (13). The return to match quality increases with both the actual and instrumented normalized logarithm of employment. The interaction effect estimates are also similar in size.

Occupation-specific local employment

The benefit of using occupation-specific local employment is that it is arguably a more relevant measure for each worker of available jobs and proxy of labor market tightness, and that it also exhibits within-municipality and year variation.

We use occupation categories at the three-digit level of SSYK-96 to be consistent with our previously described definition of a job.¹⁸

First, we use the logarithm of the number of employees in each occupation \times municipality \times year cell as the proxy for tightness. Next, we residualize this measure by running a regression at the same level of observation which incorporates municipality \times year and occupation \times year fixed effects, which implies that the variation in the residuals will stem from the deviation from the national occupation-specific employment and local overall employment in a given year. Both variants are reported in Table B1, and in both cases the return to match quality is increasing in occupation-specific employment. Moreover, there is more heterogeneity in match quality returns across the occupation-municipality log employment distribution compared to local unemployment.

Vacancies relative to unemployment

To construct an additional proxy of tightness, we use the number of new vacancies that have been added to the Swedish Public Employment Service job portal *Platsbanken* in a municipality during a particular year and divide it by the number of openly unemployed. The measure is then standardized to mean

¹⁸We have also investigated whether the local unemployment-specific returns to match quality vary by occupation by, in our main specification, interacting M and $M \times u$ (where u is either unemployment or occupation-specific local employment) by broad occupation group (one-digit level) indicators. The match quality and interaction term estimates have the expected sign, and are both statistically and economically significant, in most occupation groups. This suggests that the return to match quality is responsive to workers' outside options across broad occupation categories.

zero, standard deviation one. Again, we find that, when the relative number of vacancies is low, so is the return to match quality. One issue here is that the vacancies likely only represent a subset of available jobs in the local economy.

Share of separations to employment

The final alternative proxy of the arrival rate of outside offers is the share of workers that separate from their initial workplace between t - 1 and t that are categorized as employed in t. The share is calculated at the workplace municipality in t - 1 and year level, and workers are allowed to transition between municipalities. This is equivalent to the measure used in panel (d) in Figure 1.

Unemployment during tenure spell

One may argue that it is the history of unemployment that a worker has experienced while in a particular job that should affect to what extent his match quality can be converted into a high wage. Therefore, we have calculated the average municipality log unemployment during each worker's tenure spell, censored at 15 years. Unfortunately, municipality-level data on unemployment is only available from 1997, and so we're forced to exclude around half of all observations, especially with long tenure and in the early years of our sample.¹⁹

An alternative match quality proxy

To see whether our results are sensitive to how we define match quality, we have constructed an alternative proxy based on the wage returns to abilities within each job compared to the market-wide returns for the same abilities. The idea is that the job-specific returns are informative about the importance of each ability. However, one disadvantage to this method is the noisiness of the estimates for small job cells. To construct the measure, for each job j, we estimate a regression of the following form:

$$w_{i,j,t} = \alpha_{j,t} + \boldsymbol{\beta}'_{j}^{\text{Job}} \boldsymbol{s}_{i} + \varepsilon_{i,j,t}, \qquad (B2)$$

where $\alpha_{j,t}$ is a set of job-times year fixed effects, the vector s_i comprises the eight skill measures, and the vector β'_j^{Job} contains the returns to each skill for job *j*.

We also estimate the same regression for all workers simultaneously, treating the whole market as a single job, to obtain market-wide returns, β'^{Market} .

Our wage-based match quality metric, $M_{i,j}^{\text{wage}}$, is calculated as follows. To remove outliers, we exclude observations below the 1st and above the 99th

¹⁹To be able to compare these estimates to the baseline estimates, we have also re-estimated our main specification for this subset of observations. The interaction estimates for log wage and standardized wage are around -0.00177 and -0.0256, respectively.

percentile of the $M_{i,j}^{\text{wage}}$ distribution. We also standardize it to mean zero, standard deviation one, to make it comparable to our baseline match quality proxy:

$$M_{i,j}^{\text{wage}} = \frac{\left(\hat{\boldsymbol{\beta}'}_{j}^{\text{Job}} - \hat{\boldsymbol{\beta}'}^{\text{Market}}\right)\boldsymbol{s}_{i} - \text{mean}\left(\left(\hat{\boldsymbol{\beta}'}_{j}^{\text{Job}} - \hat{\boldsymbol{\beta}'}^{\text{Market}}\right)\boldsymbol{s}_{i}\right)}{\text{sd}\left(\left(\hat{\boldsymbol{\beta}'}_{j}^{\text{Job}} - \hat{\boldsymbol{\beta}'}^{\text{Market}}\right)\boldsymbol{s}_{i}\right)}.$$
 (B3)

The interaction term between $M_{i,j}^{\text{wage}}$ and unemployment reveals to what extent the relationship between predicted skill-bundle return in workers' jobs relative to the market and their actual wages relates to unemployment.

Alternative specifications

To verify that our results are not driven solely by large municipalities (e.g., Stockholm), and hold also in smaller settings, we have weighted the regressions with the inverse of the number of observations in each municipality \times year cell, thus putting equal weight on each cell. The interaction effect remains highly statistically significant and similar in size.

Finally, individual \times job fixed effects have been added to the specification which also incorporates job \times year fixed effects. This removes any constant wage differences between individuals and jobs. Put differently, we then rely on within individual and job variation in wages and local unemployment across time to identify the interaction between match quality and unemployment. The resulting interaction estimate for log wages is somewhat more negative, while the estimate for standardized wages is less negative, than the preferred specifications reported in Table 2, columns (2) and (5).²⁰

²⁰In unreported regressions, we have also included individual and job \times year fixed effects as well as only individual \times job fixed effects.

		Log wage	Stan	dardized wage
	Coef. (SE)	Coef.×(P90 - P10) [P90 - P10]	Coef. (SE)	Coef.×(P90 - P10) [P90 - P10]
	(1)	(2)	(3)	(4)
(a) Standard errors				
Clustering at municipality	-0.00263	-0.00177	-0.02162	-0.01455
\times year level	(0.00072)	[0.67315]	(0.00432)	[0.67315]
Clustering at job	-0.00263	-0.00177	-0.02162	-0.01455
level	(0.00126)	[0.67315]	(0.00535)	[0.67315]
Twoway clustering at	-0.00263	-0.00177	-0.02162	-0.01455
muni. \times year and job	(0.00131)	[0.67315]	(0.00635)	[0.67315]
(b) Alternative local labor markets				
Residential	-0.00425	-0.00316	-0.03706	-0.02761
municipality	(0.00049)	[0.74497]	(0.00243)	[0.74497]
Workplace local	-0.00458	-0.00296	-0.03301	-0.02132
labor market	(0.00055)	[0.64581]	(0.00239)	[0.64581]
Workplace	-0.00473	-0.00291	-0.03637	-0.02236
county	(0.00059)	[0.61468]	(0.00258)	[0.61468]
(c) Alternative proxies of expected of	utside option			
$\ln(\text{employment})_{l,t}$ –	0.01968	0.00331	0.02585	0.00435
In(employment) _l	(0.00203)	[0.16817]	(0.00879)	[0.16817]
Instrumented	0.02309	0.00297	0.02435	0.00313
$\ln(\text{employment})_{l,t}$	(0.00276)	[0.12858]	(0.01205)	[0.12858]
Occupation-specific	0.00167	0.00770	0.00996	0.04592
ln(employment)	(0.00009)	[4.60852]	(0.00047)	[4.60852]
Residualized occ.	0.00137	0.00373	0.00558	0.01517
ln(employment)	(0.00012)	[2.72120]	(0.00041)	[2.72120]
# New vacancies $_{l,t}$ /	0.00046	0.00094	0.00727	0.01485
# Unemployed _{l,t}	(0.00016)	[2.04292]	(0.00082)	[2.04292]
Share of separations	0.01841	0.00235	0.11400	0.01458
to employment	(0.00256)	[0.12/93]	(0.00995)	[0.12793]
Mean local unemployment	-0.00309	-0.00200	-0.02851	-0.01847
during tenure spell	(0.00082)	[0.64791]	(0.00297)	[0.64791]
(d) Alternative match quality proxy				
Wage-based	-0.00361	-0.00243	-0.02611	-0.01757
match quality	(0.00052)	[0.67315]	(0.00209)	[0.67315]
(e) Alternative specifications				
Weighted using $1/N$	-0.00334	-0.00225	-0.00937	-0.00631
at muni. \times year level	(0.00080)	[0.67315]	(0.00190)	[0.67315]
Individual \times job	-0.00598	-0.00403	-0.01409	-0.00948
and job \times year FE	(0.00056)	[0.67315]	(0.00248)	[0.67315]

Table B1. Robustness checks for job stayers

Notes: Columns (1) and (3) report the estimates and standard errors (in parenthesis) for the interaction between match quality and proxy of workers' expected outside options. Columns (2) and (4) report the predicted difference in the effect of match quality between the 90th and 10th percentile of the outside option proxy distribution, and the difference between the 90th and 10th percentile (in brackets). See also the description of Table 2. All regressions include the same controls as the model reported in column (2) in that table. See the description of the robustness checks in the main text.

Appendix C Robustness checks for workplace movers

This section reports results from robustness checks for the wages of workplace and firm switchers. Table C1 reports estimates from regressions estimated separately for job-to-job switchers between t - 1 and t and switchers between t - 2 and t with interim non-employment in t - 1 of the following form:

$$w_{i,j',j,t} = \rho_t + \kappa M_{i,j'} + \nu M_{i,j} + g^m(s_i) + \psi' x_{it} + g^m_{j'}(\tilde{s}_{j'}^{-i}) + g^m_j(\tilde{s}_{j}^{-i}) + \varepsilon_{i,j',j,t}.$$
(C1)

See the description of equation (C1) in the main text for definitions of each variable and subscript.

The results from these regressions are reported in Table C1. Again, we see that previous match quality is more important for EE relative to ENE switchers. However, now match quality in the new firm does not appear to be important for ENE switchers either, which is different from when estimating joint regressions for the two switcher groups. However, given the small sample of ENE switchers, we are cautious regarding how to interpret these results.

We have also estimated joint regressions for EE and ENE switchers where we augment the model from equation (C1) by introducing job fixed effects. Whenever such job fixed effects are included, the average skill controls for co-workers are rendered superfluous. These results are presented in Table C2. The regressions are again estimated both for log wages (columns (1)-(3)) and standardized wages (columns (4)-(6)). In columns (1) and (4), we control for fixed effects for the new job of the worker, in (2) and (4) we instead include fixed effects for the previous job, and both types of fixed effects are included in columns (3) and (6). In the last version of the model, we rely on variation stemming from multiple workers leaving the same job with different destinations and multiple workers entering into the same job from multiple sources. In this model, around one third of our sample is in effect lost due to lack of variation.

The results are well in line with those from Table 3: For EE switchers, both new and previous match quality is important for wages, and the estimate for previous match quality is larger than that for current match quality throughout the table. For ENE switchers, previous match quality appears to mostly play a minor role for wages, while the estimate for current match quality is often relatively large and on par with the estimate for EE switchers.

		Log	wage			Standardize	ed wage	
	ш	ίΕ	EI	E E	Ē	Е	E	Æ
	(1)	(2)	(3)	(4)	(5)	(9)	(7)	(8)
New match quality	0.0194 (0.00166)	0.0112 (0.00199)	0.00119 (0.00613)	-0.000739 (0.00637)	0.0645 (0.00530)	0.0392 (0.00619)	0.0112 (0.0133)	0.00577 (0.0138)
Previous match quality		0.0158 (0.00192)		0.00653 (0.00687)		0.0481 (0.00600)		0.0184 (0.0150)
m N $ m R^2$	58 053 0.609	58 053 0.610	1 784 0.501	1 784 0.501	58 053 0.363	58 053 0.364	1 784 0.316	1 784 0.317

 Table C1. Separate regressions for EE and ENE job movers

Notes: The table reports the results from separate regressions on wage in t for workers that moved between workplaces and firms between t - 1 and t (EE switchers) and workers who moved between t - 2 and t who were classified as non-employed in November in t - 1 (ENE switchers) for which we observe wages and match quality in the last job and current job. All regressions include controls for education and experience fixed effects, a linear control for age, a second-order polynomial for each of the eight skills and skills squared, and a second-order polynomial for the average of all skills among tenured workers in the previous as well as new job. Robust standard errors in parenthesis.

		Log wage		Sta	undardized wa	age
	(1)	(2)	(3)	(4)	(5)	(6)
EE switcher	0.0659	0.0696	0.0631	0.134	0.106	0.138
	(0.00658)	(0.00768)	(0.00790)	(0.0177)	(0.0190)	(0.0244)
$EE \times new$	0.00593	0.00426	0.00576	0.0190	0.0183	0.0233
match quality	(0.00228)	(0.00240)	(0.00242)	(0.00759)	(0.00756)	(0.00892)
	0.0216	0.0102	0.00119	0.0611	0.0255	0.0194
ENE \times new match quality	(0.0216)	(0.0103)	(0.00118)	(0.0249)	(0.0255)	(0.0184)
	(0000000)	(010100)	(0.0101)	(0.02.7)	(010_00)	(0.0_) ()
$\mathrm{EE} imes \mathrm{prev.}$	0.0101	0.0119	0.0107	0.0320	0.0354	0.0260
match quality	(0.00209)	(0.00227)	(0.00233)	(0.00711)	(0.00734)	(0.00884)
$FNE \times prev$	-0.00783	-0.00415	0.00950	-0.0166	0.0101	0.0168
match quality	(0.00703)	(0.00113)	(0.0111)	(0.0195)	(0.0252)	(0.0303)
materi quanty	(0.00771)	(0.0101)	(0.0111)	(0.0195)	(0.0202)	(0.0505)
New job FE	\checkmark		\checkmark	\checkmark		\checkmark
Prev. job FE		\checkmark	\checkmark		\checkmark	\checkmark
N	50 625	49 723	39 540	50 625	49 723	39 540
R ²	0.866	0.856	0.889	0.715	0.710	0.743

Table C2. Robustness checks for job movers

Notes: The table reports the results from regressions on wage in t for workers that moved between workplaces and firms between t - 1 and t and workers who moved between t - 2 and t who were classified as non-employed in November in t - 1 for which we observe wages and match quality in the last job and current job. All regressions include controls for education and experience fixed effects, a linear control for age, a second-order polynomial for each of the eight skills and skills squared, and a second-order polynomial for the average of all skills among tenured workers in the previous as well as new job. Each column incorporates fixed effects for the current job, previous job or both. Whenever fixed effects are incorporated, the average skill controls are rendered superfluous. The sample includes 58 053 observations with employment in t - 1 and 1 784 observations with non-employment in t - 1. Standard errors are clustered at the current job level in columns (1), (3), (4) and (6) and at the previous job level in columns (2) and (5).
Essay IV. Low-Skilled Jobs, Language Proficiency and Job Opportunities for Refugees *Co-authored with Mats Hammarstedt, and Per Skedinger*

Acknowledgements: We thank Gabriella Chirico Willstedt, Stefan Eriksson, Peter Fredriksson, Georg Graetz, Edvin Hertegård, seminar participants at IFN and Uppsala University, the editor and three referees for helpful comments, the employers who kindly participated in our interviews, and Emma Gunnarsson, Jörgen Nilson, Kasper Nilson and Sara Norbäck for excellent research assistance. The experiment was registered at the American Economic Association (for more details, see https://www.socialscienceregistry.org/trials/4832).

1 Introduction

The refugee crisis of 2015–16 resulted in large immigrant inflows from countries in the Middle East and Africa to Europe. It is well documented in several countries that refugees, and especially female refugees, experience poorer labor market outcomes—e.g., lower employment and higher unemployment rates—than both natives and other groups of foreign born (see, e.g., Brell et al., 2020). In many cases it takes a long time after arrival for both refugees and their relatives to find stable employment. A large share of immigrants from the Middle East and Africa lack higher levels of education, preventing them from entering the skilled segments of the labor market. Poor language proficiency may also be an obstacle for labor market integration.

There is an ongoing discussion, both in policy circles and in academia, regarding the value of labor market experience and language skills for the labor market integration of immigrants (see, e.g., OECD, 2018). A key idea is that the first job significantly improves labor market prospects, making a worker better suited for, and more able to find, subsequent employment opportunities. The question then becomes how to facilitate labor market entry as soon after arrival as possible. The fact that language skills are strongly correlated with favorable labor market outcomes is often used as confirmation of the importance of language training. However, despite the obvious policy relevance of these issues, causal evidence on the impact of labor market experience and language training on the integration of immigrants is scarce.

This article studies job opportunities for refugee immigrants in Sweden, a country which has experienced a very large inflow of refugee immigrants in recent years. We focus on the effects of language training provided via the Swedish for Immigrants (SFI) program and labor market experience in low-skilled jobs in a field experiment. In the experiment, we investigate the impact of experience from jobs as restaurant assistants and completed SFI for foreign-born job seekers. Applications were sent from randomly assigned fictitious Syrian refugees with different levels of previous experience and language training, to employers who advertised low-skilled job vacancies. Syrians constitute the largest group of foreign born in Sweden. This allows us to put more focus on the effects of skill variation within a particular refugee group.

We complement the field experiment with interviews with a select number of employers with extensive experience of handling applications for lowskilled jobs from persons originating from Middle Eastern and African countries. While this evidence is only suggestive in nature, it nevertheless provides some insights into what employers look for when judging job candidates. It is also informative about how actual applications are typically written. In these respects, the interviews serve as a check on our results from the field experiment. But they should also be of interest in their own right. In the econometric analysis, we are unable to demonstrate sizeable effects of previous experience or completed language training on the probability of callback from employers. However, females were more likely than males to receive a positive response on their applications. Most of the respondents in our employer interviews reported that they do not attach much value to previous experience and completed SFI. When judging applicants, the respondents had a very functional approach, considering the requirements of the task at hand and the potential for a long-term relationship rather than formal qualifications. They also put much emphasis on how motivated job candidates are. Moreover, some of the employers disclosed a preference for hiring females over males, because the former were regarded as more conscientious and adaptable. These qualitative results support our findings in the experiment.

Our study contributes to several literatures on the impact of work experience and language skills on labor market prospects for immigrants. Economic theory suggests that low-skilled jobs may lead to more qualified jobs if individuals increase their human capital by means of on-the-job-training or learning-by-doing (Becker, 1962). The transferability of skills between jobs is then of crucial importance. Moreover, such jobs may improve social capital through an expanded professional network (see, e.g., Calvo-Armengol and Jackson, 2004). Previous experience may also serve as a productivity signal when applying for other jobs (see, e.g., Spence, 1973). This may be so even if a worker's human capital is unaffected-simply exhibiting sufficient skills to handle a certain job may make a worker more attractive to other firms. Language skills can be considered an investment in the individual's human capital, and may also signal higher productivity (Chiswick and Miller, 2015). Taken together, these theories suggest that foreign-born persons, and especially refugee immigrants, may be disadvantaged by poor language skills, little work experience and inadequate professional networks.

The role of labor market experience, occupational sorting, and job mobility for the labor market assimilation process has been studied extensively in empirical work (see, e.g., Husted et al., 2001, Chiswick et al., 2005, Barth et al., 2012, and Brenzel and Reichelt, 2018). Other observational studies show that proficiency in the language spoken in the host country is associated with higher employment and higher wages for immigrants (see, e.g., Chiswick and Miller, 2015, and Yao and Ours, 2015, for surveys). There is also some evidence indicating that a large part of the difference in labor market outcomes between immigrants and natives can be explained by differences in language proficiency, as measured by tests, and not by differences in returns to these skills (see Ferrer et al., 2006, and Himmler and Jäckle, 2018). Consequently, it should be of great policy interest to investigate the labor market effects of government-sponsored, formal language training for adult immigrants. However, there are few such studies that allow a causal interpretation. Two recent exceptions are Lochmann et al. (2019) and Arendt et al. (2020), utilizing regression discontinuity designs to show that language classes improve the labor market integration of immigrants.

In general, though, the findings in the literatures that we have discussed cannot be interpreted as necessarily reflecting causal relationships. Labor market experience and language skills may be correlated with other unobserved characteristics that influence outcomes under investigation. Unlike previous observational studies on immigrants, our experimental approach enables us to identify causal relationships between, on the one hand, experience from a lowskilled job and language skills, and, on the other hand, employment prospects. Our design also allows for examining the impact of combinations of the two qualifications.

Our most salient result is that female applicants receive more callbacks than males. This is in line with other correspondence studies for the Swedish labor market, documenting that, compared to females, male applicants with foreign-sounding names are less likely to receive a positive response from employers (Arai et al., 2016; Vernby and Dancygier, 2019; Erlandsson, 2022). This evidence appears to be consistent with theories in social psychology postulating that mainly males are subject to stereotypes about foreign nationalities (see Manzi, 2019, for a literature survey). But it is inconclusive as to whether gendered ethnic discrimination is more pervasive in female-dominated occupations.¹

Previous correspondence studies dealing with assimilation in the labor market typically do not concern refugee immigrants (but native-born persons with foreign or minority background), rarely consider variations in work experience, usually do not focus on typical entry occupations for immigrants, and typically do not examine the impact of variations in language skills within a minority group. However, some correspondence studies compare the returns to work experience for foreign born or a minority group to those of natives or members of the majority group (see, e.g., Bertrand and Mullainathan, 2004, Vernby and Dancygier, 2019, and Adermon and Hensvik, 2022). The typical finding in these studies is that experience is less rewarded, in terms of callback rates, for foreign born or minority groups. However, Baert et al. (2017) find that differences in returns to experience diminish and eventually disappear with longer experience in skilled jobs. To the best of our knowledge, the only comparable previous correspondence study on language proficiency is Edo et al. (2019), which, unlike our experiment, focuses on skilled jobs and has job applicants in France signaling language skills by participation in language related activities other than language classes, such as tutoring in French and membership in reading clubs. The results indicate that signal inclusion

¹There are also Swedish correspondence studies with native applicants only that document higher callback rates for females in female-dominated occupations (Carlsson, 2011; Carlsson and Eriksson, 2019).

reduces discrimination against females with non-French-sounding names, but not against male minorities.

Based on our findings, we conclude that previous experience and completed SFI seem to provide at best a small positive signaling value when refugees apply for low-skilled jobs through formal channels. Effects of low-skilled job experience and language training may, however, be driven by other mechanisms outside our experimental setting, such as better access to informal career paths or personal networks. The fact that females from Middle Eastern and African countries have lower employment rates than males from these regions cannot be explained by females being less likely to be contacted for an interview, according to our experimental data. This indicates that the integration of foreign-born females could be improved if they apply for jobs to a greater extent—and more so than for males.

The remainder of the paper is organized as follows. The next section provides a background on the labor market situation for immigrants in Sweden. The correspondence study, involving newly arrived immigrants from Syria, is presented in Section 3. Section 4 reports the findings from the employer interviews and Section 5 concludes the paper.

2 The immigrant population in Sweden

During recent decades, immigration to Sweden has consisted mainly of refugees and their relatives, most of them originating from countries outside Europe. Of Sweden's total population of somewhat more than 10 million in 2021, around 2 million, or about 20 percent, are foreign born. The foreign-born population has doubled in size during the last 20 years, but Sweden has a fairly long history of immigration and its characteristics have changed over time. (For an overview of Sweden's immigration history, see, e.g., Boguslaw, 2012.) Since the turn of the millennium, immigration from certain countries in the Middle East (Syria and Iraq) and Africa (Somalia) have accounted for most of the migration to Sweden. The increase of people with background in non-European countries has been considerable during the 2000s. In 2000, about 220,000 and 55,000 individuals in the Swedish population were born in Asian and African countries, respectively. In 2021, the corresponding numbers have increased to about 780,000 and 230,000 individuals. The refugee immigration reached historically high levels in 2015 and 2016 when refugee immigration from countries in the Middle East, with Syria and Iraq as the dominating countries, increased as a result of the civil wars in the region. During the peak of what is known as the "refugee crisis" in 2016, more than 70,000 individuals were granted residence permit as refugees in Sweden and an additional 40,000 were granted such permit as "tied movers". Most of the residence permits during 2016 were admitted to refugees from Syria, and around 60 percent to males. Today, Syria is the dominating immigrant country in Sweden and

about 200,000 individuals in Sweden are born in Syria. Furthermore, around 150,000 individuals are born in Iraq and 70,000 are born in Somalia.

2.1 Educational attainment

Table 1 shows the educational attainment for the entire immigrant population in Sweden and for immigrants originating from Iraq, Somalia and Syria, respectively, in the age span 25 to 64 years. The foreign-born population are overrepresented among individuals with nine years of compulsory schooling or shorter education. This picture is most pronounced for individuals from Iraq, Somalia and Syria. Among immigrants from Syria and Iraq, around 30 percent had nine years of compulsory schooling or less and the corresponding figure for immigrants from Somalia is over 50 percent. This can be compared to 10 percent in the native-born population. Table 1 also highlights the fact that only 5 percent of the immigrants from Somalia have a university education that is three years or longer. For immigrants from Syria this share amounts to 15 percent. The figures can be compared to 27 percent in the native population.

2.2 Labor market integration

Several studies have documented differences in labor market outcomes between groups of immigrants in Sweden (see, e.g., Aldén and Hammarstedt, 2015, and Calmfors et al., 2018, for an overview). While immigrants originating from countries in Western Europe are doing about as well as natives on the labor market, low employment rates and high rates of unemployment characterize immigrants born in the Middle East and Africa. Table 2 shows the labor market situation for immigrants from the Middle East and Africa and also for immigrants originating from the three major non-European immigrant countries in 2017. Employment rates are considerably lower for immigrants from these regions than for natives, and this pattern is even more pronounced for females. For male immigrants, the employment rate differential to natives ranges between 25 (Iraqis) and 46 percentage points (Syrians), whereas the corresponding interval for females is 35 (Iraqis) to 61 percentage points (Syrians). A similarly bleak picture emerges for unemployment. It should be noted that the figures refer to 2017, i.e., immediately after the "refugee crisis", implying that a large number of immigrants from especially Syria only have been resident in Sweden for a very short time when we observe them in the data. This contributes to the employment rate being much lower and the unemployment rate considerably higher for Syrian immigrants than for immigrants from Iraq and Somalia in this particular year.

		Educatio	onal attainment	(percent)	
	\leq 9 years schooling	Upper secondary	University <3 years	University ≥ 3 years	Unknown
Born in Sweden	10	0			
Foreign born	20	33	14	26	7
Born in:					
—Iraq	29	30	15	22	3
—Somalia	52	29	7	5	7
—Syria	35	22	21	15	6

Table 1. Educational attainment, by region of birth, percent, 2016

Notes: The data refer to 25-64-year-olds. Source: Statistics Sweden.

	Sweden	Africa	Middle East	Iraq	Somalia	Syria
(a) Males:						
Employment rate	86.9	60.9	55.9	62.0	57.6	40.6
Share in unemployment	5.5	33.7	39.2	27.0	38.3	68.1
(b) Females:						
Employment rate	85.5	48.8	45.6	50.2	34.7	24.3
Share in unemployment	4.4	31.5	33.6	24.7	41.1	60.5

 Table 2. Labor market status by region of birth, percent, 2017

Notes: The data refer to 25-64-year-olds. The definition of employment is based on annual income taxation records. The cutoff for being classified as employed is based on a model which incorporates taxation records and data from the Swedish Labor Force Surveys for October-November. The method is designed to produce an employment measure that corresponds to the definition of employment according to the International Labour Organization as closely as possible. Unemployment is defined as "total" unemployment, i.e., being registered at the Swedish Public Employment Service as full-time unemployed or participating in any labor market program, including subsidized employment, on the 30th of November. *Source:* Own calculations, based on register data from Statistics Sweden.

2.3 Employment in low-skilled jobs

Approximately one in twenty employees in Sweden work in elementary occupations, which typically do not require more than primary education. These include, i.a., cleaners, restaurant assistants and home care assistants. It can be concluded from Figure 1 that immigrants from Africa and the Middle East are strongly overrepresented in these jobs. This is especially true for males, and for individuals who immigrated recently; despite representing only around five percent of all jobs in Sweden, elementary occupations employ about 40 percent of male immigrant employees in the studied group who immigrated the year before. After four years, the number is still above 25 percent. Even after ten years, the share of employees in elementary occupations is still around three times larger than the share for all workers in Sweden. Our data indicate that elementary occupations are an important gateway to the labor market for newly arrived immigrants from Africa and the Middle East and continue to be of significance long after immigration.

To what extent are low-skilled jobs female dominated? Table 3 reports the percentage of female workers in the largest low-skilled occupations by region of birth. The highest share of female workers is in cleaning and related services and restaurant jobs, where, e.g., three-quarters of native workers are women. Elementary occupations in construction, manufacturing and transportation instead exhibit the lowest overall shares. The female share is notably lower for workers born in Africa and the Middle East than for natives. This is particularly true for restaurant and café assistants, where only around a third of workers from Africa and the Middle East are women.

3 The field experiment

The aim of the experiment is to study the impact of language training and experience from low-skilled jobs for foreign-born persons who apply for low-skilled jobs in the Swedish labor market. Eight fictional job applicants are included in the experiment: Four males and four females, all of whom with unique Arabic-sounding names, born in Syria, 23 years old, single, living in the same suburb of Stockholm, with a high school diploma from their country of origin and with a residence permit granted in 2016. As we noted in the Introduction, a very large number of the refugees who were granted residence permits in Sweden in recent years have a Syrian background.² In order to

²It is not explicitly stated in the applications that the applicants have been refugees. Strictly speaking, they could have been granted residence permits also as "tied movers" (see Section 2). (Around 24 percent of granted residence permits for Syrians during 2015-16 were for "tied movers" and very few, if any, were for work or studies, according to the Swedish Migration Agency.) The distinction between refugees and "tied movers" is not likely to matter much for employers in our experiment. For simplicity, we use the term "refugees" forthwith to include also migrants for family reasons.

Figure 1. Probability to be in elementary occupation for employees born in Africa and the Middle East, by gender and years since immigration



Notes: The data refer to individuals aged 20 to 64, observed between the years 2000 and 2013. Occupation based on the 1996 version of SSYK. *Source:* Own calculations, based on register data from Statistics Sweden.

Table 3. Share of females in low-skilled jobs, by occupation and region of birth, 2017

Occupation	Ν	Native	Foreign born	Africa	Middle East	Iraq	Somalia	Syria
Cleaning and re- lated services (91)	87,392	75.1	68.6	47.6	38.1	37.9	43.7	34.9
Construction, man- ufacturing and transport (93)	22,395	20.0	22.4	8.7	12.7	16.5	2.4	8.9
Restaurant and café assistants (94)	59,781	77.9	44.6	29.3	34.7	41.9	34.9	23.2
Other service work- ers (96)	47,757	30.7	31.5	21.2	26.6	25.8	23.7	22.0
All low-skilled oc- cupations	220,505	54.5	53.2	36.6	33.0	36.6	36.7	24.9

Notes: The data refer to 25-64-year-olds. Elementary occupations, as defined by the International Standard Classification of Occupations, require at most primary education. Occupation based on the 2012 version of SSYK. Each cell reports the percentage of females of all workers by occupation and region of birth. The figures for all low-skilled occupations include the minor occupations "Berry pickers and planters" and "Market salespersons". *Source:* Own calculations, based on register data from Statistics Sweden. avoid having job applicants with a long work history, we have chosen them to be relatively young. The experiment was carried out during the period January to December 2019 (i.e., before the Covid-19 pandemic broke out).

3.1 Design of the experiment

All eight job applicants were registered at the Swedish Public Employment Service in August 2016. Four of them signaled work experience by stating employment in a low-skilled job—restaurant assistant in a well-known fast-food restaurant chain (starting in November 2017)—in their applications, while the other four instead continued to be registered at the employment service. Four of the applicants signaled language proficiency by stating that they had completed the entire Swedish for immigrants (SFI) program, while the other four did not mention anything about such training. The applicants were randomly distributed to advertisements for low-skilled jobs. Through this procedure, we thus get the following four types of applicants of each gender:

- 1. One who has been registered with the Public Employment Service until the time of application and who does not mention anything about completed SFI.
- 2. One who has been registered with the Public Employment Service until the time of application and who claims to have completed SFI.
- 3. One who, after being registered with the Public Employment Service, worked in a low-skilled job until the time of application and who does not mention anything about completed SFI.
- 4. One who, after being registered with the Swedish Public Employment Service, worked in a low-skilled job until the time of application and who claims to have completed SFI.

Our hypothesis is that experience from a low-skilled job and completed SFI should increase the probability that employers respond positively to an application, as these two characteristics should signal higher productivity, relative to continued unemployment and not having completed language training, respectively.

We chose to signal labor market experience by having the applicants refer to a well-known fast food chain since the majority of people in Sweden should understand roughly what tasks are performed and what level of effort is required to carry out a low-skilled job in such a restaurant. Thus, the employer should relatively easily be able to infer the value of such experience for the advertised job at hand.

The purpose of the SFI education, which is the responsibility of municipalities and is free of charge, is to provide basic proficiency in the Swedish language to adult immigrants (Swedish National Agency for Education, 2018). SFI is structured in three different paths depending on the individual student's prior general skill level. Within each path, students can advance to courses with a gradually higher degree of difficulty (courses A to D for path 1, B to D for path 2 and C to D for path 3), but with the same knowledge requirements regardless of the path. The courses deal with listening and reading comprehension, oral proficiency and writing skills. While courses A and B focus on informal language in everyday situations, courses C and D also teach students more formal language used in workplaces, educational institutions and the wider community. SFI can be vocationally oriented, with courses sandwiched with practical work, but it seems that only a minority of students—data at the national level are not available—get access to these vocational orientations (Swedish Schools Inspectorate, 2010).

Although SFI is a compulsory part of the Establishment program (*etableringsprogrammet*) that certain refugee immigrants are supposed to participate in, a substantial share do not complete the training.³ During the period 2014–18, only between 35 and 37 percent of the participants completed the SFI courses, according to statistics published by the Swedish National Agency for Education. Participants drop out for various reasons, not necessarily detrimental to the individual. Some may have found a job, while others may have entered other types of education, moved to another part of Sweden or returned to their home country. Notwithstanding, it seems plausible that an explicit reference to having actually completed the language training can be perceived as a positive signal regarding language proficiency by the employers.

Employers responded to applications via email or phone message. We distinguish between two types of callback: (i) Invitation to an interview or request for more information about the applicant (than what appears from the cover letters and attached resumés), and; (ii) Invitation to an interview. As expected, no employers responded with direct job offers, so no such callbacks are categorized. To minimize the inconvenience for employers, we promptly declined any callback offers. In line with other correspondence studies, we interpret requests for more information from the applicant as a positive signal from the employer, even if it does not lead to an interview or a job offer.

The application letters were designed to be simple and rather short, but were written in grammatically correct Swedish without spelling mistakes. We decided not to signal language proficiency by variations in how correct the language in the application letters was. Arguably, this signal can be weak, for several reasons. First, checks for grammatical and spelling errors are integral parts of most word processing programs. Second, templates of correctly formulated applications are readily available on the Internet. Finally, relatives and friends with good language skills may assist in the formulation of the applications. As we show in Section 4, employers seem well aware of these possibilities, implying that they are inclined not to draw strong conclusions

³The Establishment program is organized by the Public Employment Service and consists of support in the form of activities and education, directed to newly arrived refugee immigrants aged 20–64 with residence permits.

regarding language proficiency from an application in correct Swedish until they have talked to the applicant in person. There are also other concerns with varying the quality of the language in the application letters. Such variation may, apart from language proficiency, signal other personal characteristics like ambition and conscientiousness, making the interpretation of any effects on callback complicated. Focusing on language training only is arguably also more policy relevant.

An example of an application with cover letter and resumé, translated to English, is shown in Appendix. The brief letters may also have motivated some employers to request additional information instead of forthwith inviting the applicant for an interview.

The eight applicants were randomly assigned to low-skilled job openings across the country, which were advertised on the Swedish Public Employment Service's portal *Platsbanken*. We selected five occupations at the lowest skill level (i.e., elementary occupations) according to the Swedish Standard Classification of Occupations (SSYK): Restaurant/café assistant, cleaner, newspaper/leaflet deliverer, home care assistant and hand packer. At this skill level, these occupations are among the most common among foreign-born, according to Statistics Sweden. These occupations are also characterized by lower language requirements than more skilled occupations (Ek et al., 2020).

Advertised jobs were not applied for if qualifications were explicitly required that were not clear in all applications (for example, previous work experience, good knowledge of Swedish or other languages, special training, driving license or local knowledge) or if information about applicants were requested that we could not provide (for example, social security number or photo). We also excluded jobs advertised by staffing firms, as we lack the necessary information about the client firm in which the employee will work. In addition, we excluded jobs in the fast-food chain at which some of the fictitious applicants were already employed.

It turned out that many jobs in three of the occupations—newspaper/leaflet deliverer, home care assistant and hand packer—could not be applied for, to a large extent for the reasons stated above, and they were also relatively few in number. Consequently, almost all applications concerned jobs as restaurant/café assistants or cleaners. However, these occupations account for a very large proportion of the low-skilled jobs among foreign born, 86 percent for males and 61 percent for females in 2017 (Ek et al., 2020). In practice, the restrictions also meant that only jobs in the private sector could be applied for, as social security numbers are requested in job advertisements in the public sector.

The experiment was registered with the *American Economic Association's* registry for randomized controlled trials before performing any analyses using the collected data, which means that we specified in advance the regressions to be estimated and for which groups. We also performed some power calculations (to be discussed in Section 3.4), although the sample size was

not determined by these, but rather by a predetermined start and end date of the trial. Moreover, we submitted the experimental design in advance to *Etikprövningsnämnden* (the Ethics Review Board) in Stockholm for ethical approval, which is standard procedure for research projects involving experimental subjects in Sweden. They decided that no ethical review was necessary.

In total, we sent out 2,184 applications. For 1,958 of these, we were able to determine the geographical location (municipality) of the job and whether the advertisements referred to an open-ended or fixed-term contract and/or a full-time or part-time job and in the analyses below we only include these observations.⁴

3.2 Descriptive results

Table 4 displays descriptive statistics for the variables included in our analysis. A first observation is that the callback rates were low: 3.9 percent for interview or more information and 1.4 percent for interview. However, the callback rates are fairly similar to those for non-European immigrants in other Swedish correspondence studies, although these results are not strictly comparable.⁵ Furthermore, 63.3 (44.5) percent of the jobs were open-ended contracts (full-time), whereas the corresponding figure for the labor market as a whole is 83.4 (78.5) percent, according to Statistics Sweden. The jobs in the experiment are thus not only low-skilled, employment contracts are also atypical to a greater extent than is the case for the labor market as a whole. Some jobs could only be applied for via the employer's own web portal, and not by email. However, only 9 percent of applications were made through such online forms.

Table 5 reports balancing checks, where the characteristics of the job vacancies are related to those of the fictitious applicants. Since applicants are randomly assigned to each vacancy, there should be no systematic differences in job characteristics across them. We include indicators for open-ended and full-time contracts, if jobs were applied to via online forms, if the job was as a cleaner or restaurant assistant, if the job was located in the Stockholm local labor market area, and the distance from job to home. The upper part of the table reports averages for each of the eight applicants, while the lower part shows coefficients and standard errors from regressing the job character-

⁴The qualitative results regarding the randomized variables (SFI completion, experience and gender) are not affected by this choice, and the regression estimates are very similar when instead using all 2,184 observations and not including controls for job characteristics.

⁵There is no previous Swedish study that is fully comparable to ours, in terms of applicant groups and types of jobs. Some of the results in Carlsson (2010) and Vernby and Dancygier (2019) come closest. The former study reports a callback rate of 7 percent for persons born in the Middle East applying for low-skilled jobs in the restaurant sector. In the latter study, callback rates for Iraqi- and Somali-born turn out to be 10 and 5 percent, respectively, but the restaurant and café jobs applied for include not only low-skilled ones, as in our study, but also higher-skilled jobs.

	Mean	Standard deviation
(a) Type of callback:		
Interview/more information	0.039	0.194
Interview	0.014	0.117
(b) Characteristics of job applicants:		
Completed language training (SFI)	0.508	0.500
Experience as restaurant assistant	0.500	0.500
Female	0.508	0.500
(c) Characteristics of jobs applied for:		
Open-ended contract	0.633	0.482
Full-time schedule	0.445	0.497
Online form	0.089	0.285
Distance to job from home	284.0	249.3
Stockholm area market	0.369	0.483
Hand packer	0.006	0.078
Home care assistant	0.002	0.045
Restaurant/café assistant	0.697	0.460
Cleaner	0.291	0.454
Newspaper/leaflet deliverer	0.004	0.060

Table 4. Descriptive statistics for the experiment

Notes: 1,958 observations. Distance is in kilometers between the residential municipality and the municipality in which the job is located, as indicated by Google Maps.

istics onto indicators for having previous labor market experience, completed language training, and for being female, separately for each outcome and explanatory variable. Overall, the treatment is balanced over job characteristics. The only exception is that applicants with previous experience are four percentage points less likely to apply for jobs as restaurant assistants, with an average probability of just below 70 percent, which is mirrored in the higher probability of applying to jobs as cleaners.

Figure 2 shows the callback rates in our experiment by gender and type of job applicant, together with 95 percent confidence intervals, for the broad definition of callback (interview or request for more information), while Figure 3 shows corresponding rates for the narrow definition (interview).⁶ The callback rate for females (around six percent) is three times as large as for male applicants (two percent). Within genders, there are no apparent differences across types, suggesting no large returns in the form of higher callback rates for applicants with completed language training, work experience or with both of these qualifications, relative to those with neither of them.

3.3 Econometric framework

The econometric analysis is based on linear probability models estimated with OLS. As the main dependent variable, we use an indicator variable for if there was a callback from the employer, either regarding invitation to an interview or a request for more information. We will also conduct analyses with just invitation to an interview as the dependent variable.

In the econometric analysis, our basic model is represented by the following equation:

$$y_i = \beta_0 + \beta_1 \times SFI_i + \beta_2 \times EXP_i + \beta_3 \times FEMALE_i + \gamma' X_i + \varepsilon_i, \qquad (2)$$

where y is the outcome of interest, *SFI* is an indicator for whether the applicant successfully finished language training, *EXP* is an indicator for whether the job applicant has experience from a low-skilled occupation, *FEMALE* is an indicator for female applicants, and ε is the error term. Job applications are indexed by *i*. Although not necessary for identification, the model also includes a vector of additional, non-randomized controls, **X**, which comprises indicator variables for whether the employment contract is open-ended or fixed-term, for whether the job is full-time or part-time, the distance from job to home, using data from Google Maps and scaled to lie between 0 and 1, as well as indicator variables for the occupations. As hand packers, home care assistants and newspaper/leaflet deliverers accounted for very few observations, these occupations have been merged into a single category, "Other occupations".

⁶This is equivalent to estimating a fully saturated, non-parametric model of callback for each of the eight applicants.

	(1) Open- ended	(2) Full- time	(3) Online form	(4) Cleaner	(5) Restaur- ant ass.	(6) Stock- holm	(7) Distance to job
(a) Applicants:							
Male	0.631	0.416	0.077	0.266	0.717	0.348	301
Male with language training (SFI)	0.617	0.424	0.074	0.288	0.700	0.329	300
Male with experience as restaurant assistant	0.658	0.440	0.115	0.272	0.720	0.395	263
Male with SFI and experience	0.669	0.429	0.118	0.294	0.682	0.376	276
Female	0.606	0.469	0.095	0.276	0.720	0.331	297
Female with language training (SFI)	0.639	0.430	0.100	0.265	0.731	0.430	257
Female with experience as restaurant assistant	0.627	0.464	0.047	0.326	0.652	0.356	299
Female with SFI and experience	0.620	0.488	0.081	0.341	0.655	0.388	279
(b) Estimated effect of:							
Female	-0.021 (0.022)	0.035 (0.022)	-0.015 (0.013)	0.022 (0.021)	-0.014 (0.021)	0.014 (0.022)	-2.242 (11.272)
EXP	0.020 (0.022)	0.020 (0.022)	0.004 (0.013)	0.035* (0.021)	-0.040* (0.021)	0.019 (0.022)	-9.376 (11.267)
SFI	0.006 (0.022)	-0.004 (0.022)	0.009 (0.013)	0.013 (0.021)	-0.012 (0.021)	0.024 (0.022)	-12.392 (11.264)

Table 5. Characteristics of vacancies, by applicant type

Notes: The upper part of the table reports average job characteristics of the applications sent from the eight applicants, while the lower part shows estimates from separately regressing the job characteristics onto the characteristics of the applicants. Distance is in kilometers between the residential municipality and the municipality in which the job is located, as indicated by Google Maps. Robust standard errors in parentheses. * indicates statistical significance at the 10-percent level.

Figure 2. Callback rates for broad definition of callback, by gender and type of job applicant



Notes: Callback refers to invitation to an interview from the employer or enquiry for more information about the applicant. SFI and EXP stands for completed education in Swedish for immigrants and experience as restaurant assistant, respectively. 994 observations for female applicants and 964 for male. 95 percent confidence intervals.

Figure 3. Callback rates for narrow definition of callback, by gender and type of job applicant



Notes: Callback refers to invitation to an interview from the employer. SFI and EXP stands for completed education in Swedish for immigrants and experience as restaurant assistant, respectively. 994 observations for female applicants and 964 for male. 95 percent confidence intervals.

In alternative specifications, we add an interaction between *SFI* and *EXP* to the model above, in order to capture heterogeneous effects for different combinations of qualifications. Later on, we also examine if there are heterogeneous effects with regard to gender, occupation, the region in which the job is located, and mode of application, i.e., whether the job could be applied for by email or only by using the employer's own web portal.

3.4 Econometric results

In the registered analysis plan, our point of departure for the power calculations was a sample of 3,000 applications, a callback rate of 5 percent and a 5 percent significance level, implying that we can detect an effect of about 2.5 percentage points with an 80 percent probability (the standard power level used in experimental studies). The actual number of observations in our study turned out to be lower (2,184).⁷ The actual callback rate for all applicants was slightly below 4 percent (for interview or more information). With that as the correct underlying baseline, the real effect of any intervention would need to be around 2.6 percentage points in order for us to reject with 80 percent probability the null hypothesis that there is no effect of, for example, completed SFI.

The first set of regressions, for all job applicants and both definitions of callback, is shown in Table 6, while Table 7 reports regressions by gender, using the broader response measure only. Neither SFI nor EXP contributes significantly to higher (or lower) callback rates and this is true also for the interaction $SFI \times EXP$, although the latter is borderline significant in the sample with male job applicants. In some of the regressions, the estimated effects are even negative, but they are never significant. The only variable that seems to systematically affect callback rates is the applicant's gender: Females are about 3.8 percentage points more likely than males to be asked for an interview or for more information. The difference is 1.3 percentage points for the narrower measure of invitation to an interview. Relatively speaking, the gender difference is very large; the probability of being invited to an interview or asked for more information increases by approximately 190 percent if the applicant is female instead of male, while the corresponding increase for invitation to an interview amounts to around 180 percent.⁸ The higher callback rate for females is also consistent with our finding in the employer interviews

⁷Basically all jobs were applied for that appeared on *Platsbanken* during the period of study and fulfilled our selection criteria.

⁸To obtain the relative effects, we divide the estimated coefficient for the *FEMALE* indicator by the average probability of receiving a callback for male applicants. The average callback rate for invitation to an interview or request for more information was 2.0 percent for males and 5.8 percent for females, while the corresponding rate for invitation to interview was 0.7 and 2.0 percent, respectively.

that some employers prefer hiring females over males in the low-skilled jobs we study (see Section 4).

With all applicants included and without interaction between *SFI* and *EXP*, the confidence interval for *SFI* ranges from -0.027 to 0.009, while that for *EXP* is between -0.018 and 0.018. Since the overall callback is only 4 percent, we cannot rule out sizeable relative effects. In absolute terms, however, the effects appear small; based on the upper bound of the confidence intervals, when signaling experience one would receive at most one additional positive callback per $\frac{1}{0.018} \approx 56$ applications and when signaling *SFI* the corresponding number is $\frac{1}{0.009} \approx 111$. With separate regressions by gender and with interaction effects the confidence intervals become even larger, since we are then in effect comparing either four or eight groups.

What about the control variables in the regressions? It should be noted that job characteristics were not randomized in the experiment, so it is not possible to interpret our findings for these variables in causal terms out of hand. If the advertisement refers to an open-ended contract, the callback rate for the more broadly defined response decreases by 1.9 percentage points, but the coefficient is close to zero for the interview alone. The fact that jobs with openended contracts are associated with lower callback rates may reflect that these positions are more attractive to apply for, but employers may also become more demanding when recruiting to such jobs as costs of dismissal are substantially higher. The mode of application could reflect unobserved employer characteristics correlated with callback rates, such as size, the number of expected applicants and the resources devoted to the screening of applicants. However, neither for full-time jobs nor online forms do we see a relationship with callback. While full-time jobs may also be more attractive to apply for, they are not associated with higher dismissal costs than part-time positions. Since all applicants resided in the Stockholm area but applied for jobs all over the country, we examine whether geographical distance affects callback rates. For example, the low callback rates in general may reflect that many jobs (over 60 percent) were located outside of Stockholm. The distance variable is rescaled to run between 0 and 1, so the coefficient can be interpreted as the difference between applying for jobs in the residential municipality and the municipality farthest from this municipality. We find a negative effect of the distance variable, but it is significant only for the narrow definition of callback. Employers may be reluctant to directly invite faraway applicants to interviews for low-skilled jobs, out of misgivings that the applicants are not genuinely interested in the position (see Section 4, where this matter is discussed in more detail).

Examination of heterogeneous effects with regard to independent variables is rendered somewhat problematic because of low power, but outlines of such analyses were part of our pre-registration plan and the results are provided in

	(1)	(2)	(3)	(4)
	Interview/m	ore information	Inter	view
Language training (SFI)	-0.009	-0.014	-0.002	0.006
	(0.009)	(0.012)	(0.005)	(0.007)
Experience as restaurant assistant	-0.000	-0.006	0.001	0.009
	(0.009)	(0.013)	(0.005)	(0.008)
SEL × Experience		0.012		-0.015
		(0.012)		(0.011)
Female	0.038***	0.038***	0.013**	0.013**
	(0.009)	(0.009)	(0.005)	(0.005)
Open-ended	-0.019*	-0.019*	-0.006	-0.006
L L	(0.010)	(0.010)	(0.006)	(0.006)
Full-time	-0.010	-0.011	-0.008	-0.008
	(0.009)	(0.009)	(0.005)	(0.005)
Online form	-0.010	-0.010	0.010	0.011
	(0.016)	(0.016)	(0.012)	(0.012)
Scaled distance to job from home	-0.022	-0.022	-0.034***	-0.034***
j.	(0.021)	(0.021)	(0.012)	(0.012)
Cleaner	-0.006	-0.006	0.015**	0.015**
	(0.043)	(0.044)	(0.006)	(0.006)
Restaurant/café assistant	-0.010	-0.010	0.015***	0.015***
Restaurant/care assistant	(0.043)	(0.043)	(0.005)	(0.006)
Constant	0.05(0.050	0.007	0.002
Constant	(0.056)	0.059 (0.044)	(0.007)	(0.003)
Number of observations	1,958	1,958	1,958	1,958
R ²	0.014	0.015	0.009	0.010

Table 6. Callback regressions, by type of response

Notes: The regression models are estimated with OLS. The reference category for the occupations is Other occupations (hand packers, home care assistants and newspaper/leaflet deliverers). Robust standard errors in parentheses. *, ** and *** indicate statistical significance at the 10-, 5- and 1-percent level, respectively.

	(1)	(2)	(3)	(4)
	Female a	applicants	Male ap	oplicants
Language training (SFI)	-0.018	-0.015	0.002	-0.014
	(0.015)	(0.021)	(0.009)	(0.013)
Experience as restaurant assistant	-0.003	0.000	0.002	-0.013
	(0.015)	(0.023)	(0.009)	(0.013)
$SFI \times Experience$		-0.006 (0.030)		0.031* (0.018)
Open-ended	-0.027	-0.027	-0.012	-0.012
	(0.017)	(0.017)	(0.011)	(0.011)
Full-time	-0.013	-0.013	-0.009	-0.008
	(0.015)	(0.015)	(0.010)	(0.010)
Online form	-0.008	-0.008	-0.012	-0.012
	(0.031)	(0.031)	(0.013)	(0.013)
Scaled distance to job from home	-0.019	-0.018	-0.028	-0.028
	(0.038)	(0.038)	(0.020)	(0.020)
Cleaner	-0.058	-0.057	0.017*	0.020*
	(0.118)	(0.118)	(0.009)	(0.010)
Restaurant/café assistant	-0.070	-0.069	0.022***	0.025***
	(0.118)	(0.118)	(0.007)	(0.008)
Constant	0.162	0.160	0.016	0.022
	(0.120)	(0.120)	(0.014)	(0.015)
Number of observations	994	994	964	964
R ²	0.008	0.008	0.005	0.008

 Table 7. Callback regressions for broad definition of callback, by gender

Notes: The regression models are estimated with OLS. The reference category for the occupations is Other occupations (hand packers, home care assistants and newspaper/leaflet deliverers). Robust standard errors in parentheses. *, ** and *** indicate statistical significance at the 10-, 5- and 1-percent level, respectively.

Table 8. All regressions use the broader measure of callback as the dependent variable.

Separate regressions for cleaners and restaurant/café assistants reveal no sizeable positive effects from signaling SFI or experience for any occupation. The coefficient for *SFI* is significantly negative for the latter group. (Given the large number of estimates in the table, we are of course likely to find some significant effects just by chance.) Surprisingly, we find no positive effect of non-trivial magnitude on callback from signaling previous experience as a restaurant/café assistant even when such jobs were applied for.

We have examined heterogeneity relating to geographical distance in two ways. First, we have added interactions between the measure of the traveling distance between job and home and the *SFI*, *EXP* and *FEMALE* indicators. Second, we estimate separate models for job postings inside and outside the Stockholm local labor market as well as a joint model where the randomized explanatory variables are interacted with an indicator for if the job was located in Stockholm. However, we find no relationship between, on the one hand, the traveling distance from the residential municipality and whether the jobs are in Stockholm and, on the other hand, effect sizes for the randomized variables. Furthermore, we find no difference in the returns from signaling completed SFI or experience with respect to the mode of application, but there is a negative effect of the distance variable when only online forms were used.

Finally, in unreported regressions (available on request from the authors) we investigated whether the month of application during the year-long experiment and repeat applications to the same employer matter. The month of application reflects pure calendar effects, on the one hand, as well as effects stemming from the fact that both length of previous experience and duration of unemployment increases over time, on the other hand. It is not possible to distinguish between the two effects with our data. We divided the sample according to month of application and estimated the regressions corresponding to Table 6, column 1, separately for each month. We see no clear trends in the estimates for SFI or EXP, but there is a weak tendency for the estimates for females to be somewhat smaller in the second half-year. Moreover, it turned out that many employers received more than one application-38 percent of all applications were made to an employer whose name and/or contact details appear multiple times in our sample. However, a large number of these occurrences were due to applications being sent to jobs at large chains/firms with multiple establishments across the country, hiring locally. Only about half of the occurrences (21 percent) in our main sample were associated with an email address that appears multiple times. These were often addresses used specifically for recruitment purposes. Although the applications were for different job postings and over the course of a year, it is conceivable that these firms discovered that an experiment was going on and consequently differ from other firms in their response. However, separate regressions, corresponding to Table 6, columns 1-2, for applications to employers who received only one application throughout the experiment show only small differences to the regressions using the full sample in the estimates for *SFI*, *EXP* and *FEMALE*.

4 The employer interviews

After the field experiment was completed, we conducted interviews with employers with extensive experience of handling and judging applications for low-skilled jobs from persons of non-European origin. The purpose of the interviews was to shed additional light on what employers look for in such applications and how these are typically written. Contact information to suitable employers was provided by two Swedish employer associations, *Visita* and *Almega*. The former consists of firms in the hospitality industry and the latter directs itself to various other service industries, including cleaning firms.

In total, we contacted ten employers, five of whom did not respond or declined to participate. We carried out interviews with four employers from the hospitality industry and one cleaning firm. These were conducted via Zoom and recorded (with the consent of the interviewees) and lasted between 30 minutes and one hour. After the Zoom interviews, some follow-up questions were communicated via email. All the respondents were directly involved in recruitment, either as owners of the firms, chief operating officers or heads of human resources departments. The participating firms were located in different geographical areas of Sweden and of different size in terms of the number of employees: One small firm (49 employees or less), two medium-sized firms (50–249 employees) and two large firms (250 employees or more).

Before the interviews, the respondents were informed about the purpose of the interviews and that the identity of the firms would not be revealed. The interviews were semi-structured, and based on a questionnaire (that the respondents were given access to in advance), but allowing for follow-up questions depending on the answers given.

The number of interviews is small and the employers were not chosen randomly, implying that the evidence we collected should be regarded as suggestive in nature. Our respondents were, however, quite unanimous in several important respects.

All firms except one (a former user) reported that they use the web portal of the Public Employment Service, *Platsbanken*, as the main recruitment channel. Some interviewees also use social media, e.g., Facebook and LinkedIn, or the firm's own website. The type of low-skilled jobs our respondents advertise include restaurant assistants, cleaners and, in one firm, janitors/park-tenders.

All of the employers stated that they receive many applications for lowskilled jobs: From 30–40 applications for a single position in one firm up to 1,500 applications for a couple of hundred positions in another firm. Handling such large amounts of applications obviously require a great deal of resources on part of the firms. Some employers argued that the task is made more diffi-

	(1)	(2)	(3)	(4)	(5)	(9)	(L)	; ; (8)	(6) .	(10)
	Stock	holm vs. other	local labor ma	rket	Heterogenec	ous effects by	occupation	Applicatic	ons via online f	orms
	Distance interaction	Stockholm interaction	Stockholm only	Other only	Rest. ass. interaction	Rest. ass. only	Cleaners only	Online form interaction	Other applications	Online forms
Language training (SFI)	-0.015	-0.008	-0.007	-0.007	0.021	-0.021**	0.027	-0.011	-0.011	0.013
	(0.013)	(0.011)	(0.014)	(0.011)	(0.016)	(0.010)	(0.017)	(600.0)	(600.0)	(0.027)
Experience as restaurant assistant	-0.001	0.001	-0.003	0.001	-0.00- (210.0)	100.0	-0.008	-0.003	-0.003	0.020
Female	(0.01 <i>5</i>) 0.034**	(0.011) 0.045^{***}	(0.014) 0.030^{**}	(0.011) 0.045^{***}	(0.016) 0.055***	(0.010) 0.032^{***}	(0.018) 0.052^{***}	(0.00)	(0.09) 0.037***	(0.029) 0.072**
	(0.013)	(0.011)	(0.014)	(0.011)	(0.016)	(0.010)	(0.017)	(0.00)	(600.0)	(0.033)
Open-ended	-0.019^{*}	-0.018^{*}	-0.042*	-0.008	-0.019*	-0.014	-0.034*	-0.019*	-0.022**	0.014
Full-time	-0.010	-0.010	0.003	-0.012	-0.011	-0.006	-0.016	-0.011	-0.009	-0.014
	(6000)	(0.00)	(0.014)	(0.012)	(0.00)	(0.010)	(0.019)	(600.0)	(0.010)	(0.028)
Online form	-0.010	-0.011	0.047	-0.026^{*}	-0.012	0.004	-0.031	-0.049**		
	(0.016)	(0.016)	(0.047)	(0.015)	(0.016)	(0.022)	(0.024)	(0.023)		
Scaled distance to job from home	-0.049	-0.053	0.728	-0.060*	-0.018	-0.028	0.004	-0.022	-0.006	-0.184^{**}
	(0.043)	(0.037)	(0.504)	(0.035)	(0.022)	(0.023)	(0.045)	(0.021)	(0.022)	(0.080)
Cleaner	-0.006	-0.006	0.065^{**}	-0.019	-0.011			-0.006	-0.024	0.029
	(0.041)	(0.041)	(0.031)	(0.056)	(0.041)			(0.044)	(0.065)	(0.028)
Restaurant/café assistant	-0.011	-0.010	0.068**	-0.030	0.015			-0.010	-0.030	0.045*
Stockholm area	(140.0)	0.041)	(000.0)	(000.0)	(+++0.0)			(0.40.0)	(+00.0)	(0.024)
		(0.022)								
SFI interaction	0.029	-0.001			-0.042**			0.021		
	(0.044)	(0.018)			(0.019)			(0.028)		
EXP interaction	0.003	-0.005			0.007			0.037		
-	(0.044)	(0.018)			(0.019)			(0.031)		
Female interaction	0.022	-0.017			-0.023			0.020		
	(0.044)	(0.018)			(610.0)			(0.032)		
Number of observations	1,958	1,958	723	1,235	1,958	1,365	570	1,958	1,784	174
R ²	0.015	0.015	0.025	0.019	0.018	0.013	0.030	0.015	0.015	0.078
<i>Notes:</i> The table reports the results of va	uriations of the r	egression mode	el reported in	the first colu	umn of Table	6. estimated l	ov OLS. In c	olumn (1), we ad	dd interactions	between
the distance between the residential mun	icipality and the	municipality	in which the je	ob is located	and language	e training, wo	ork experienc	e and female inc	dicators (hence	forth the
randomized indicators). The distance me	asure is rescaled	to run betwee	n 0 and 1. Col	lumn (2) add	ls interactions	between an i	indicator for j	if a job is located	d in the Stockh	olm area
and the randomized indicators. Column	(3) and (4) only	y include jobs	located in and	outside the	Stockholm au	rea, respectiv	ely. Column	(5) interacts the	e randomized ir	idicators

with an indicator for if the job is as a restaurant assistant. Column (6) and (7) only include jobs as restaurant assistants and cleaners, respectively. In column (8), the randomized indicators are interacted with an indicator for if the job was applied to via a web portal instead of by sending a letter and a resumé via email. Column (9) excludes all applications made via web portals, while Column (10) only includes such applications. The reference category for the occupations is Other occupations (hand packers, home care assistants and

newspaper/leaflet deliverers). Robust standard errors in parentheses. *, ** and *** indicate statistical significance at the 10-, 5- and 1-percent level, respectively.

Table 8. Robustness regressions for broad definition of callback

cult by the perception that some applications are not seriously intended, only serving as a means to fulfill requirements for unemployment benefits or activity support, i.e., monetary compensation for those participating in active labor market programs, sometimes resulting in no-shows for job interviews. These difficulties were also the reason why the former user of *Platsbanken* stopped using it entirely.

A clear majority of the applications for low-skilled jobs come from persons with perceived origin in non-European countries, according to all but one of our respondents. Estimated shares range between 50 and 90 percent. Regarding the share of applications from females, responses were more dispersed, between 20 and 80 percent. Three of the firms were of the opinion that there are no important differences in job performance between females and males, but two respondents regarded females as being more conscientious and adaptive. One of these respondents, with only 20 percent of the applications from females, would like to see the share of females in the firm increase. This respondent also claimed that females are more meticulous than males in cleaning jobs, and that this possibly reflects habits from their home countries, where females traditionally assume full responsibility for household chores.

Two respondents reported that, on average, there are differences in social skills between females and males. One of them stated that women can be relatively quiet and unobtrusive, in line with traditional gender roles. Another respondent was of the opinion that males connect faster to co-workers than females.

Without exception, the firms reported that job applications from non-Europeans tend to be very short and simple, containing only basic information. A length of six to seven sentences in the cover letter is typical for persons from Africa and the Middle East and some applicants do not even include a letter, only the resumé. Some applications are more standardized and formal than others, and appear to be written using templates, from SFI or the Internet, or with the assistance of a job coach from the Public Employment Service. While our respondents informed us that grammatical and spelling errors are common in the application letters, not much emphasis seems to be put on this in the screening process.

Most of the firms did not regard completed SFI as important. A common view was that proficiency in Swedish varies a great deal among those who have completed SFI, implying that the language skills signal is rather weak. The best way to assess language proficiency is to meet the applicant in person. One interviewee claimed that the quality of SFI is not always first-rate and that there are other ways to learn Swedish. Another one mentioned that many positions do not require advanced knowledge of Swedish. A third respondent observed that completed SFI does not necessarily imply good communication skills—some employees with fluency in Swedish do worse when it comes to communicating with co-workers and customers, while some employees lacking in language skills are better at this. One respondent, however, had a more

positive attitude toward the merit of completing SFI and argued that it shows dedication to and ability for learning. There is little to suggest that the responses were due to the interviewees being completely ignorant about SFI or lacking in interest; two employers had personal experience of SFI, as a visitor to language classes and co-arranger of vocationally oriented SFI, respectively. The extent to which our respondents put emphasis on proficiency in Swedish seems to vary with occupation. For jobs requiring direct contact with customers basic knowledge of Swedish is seen as quite important, while other jobs are less demanding in this respect (but some proficiency in English, at least, is desirable).

Regarding the value attributed to previous labor market experience in Sweden, opinions differed. Three of the firms do not attach much importance to this when recruiting, be it from the same type of job as the vacant position or not and regardless of length. One respondent emphasized the importance of on-the-job training. Under supervision from an experienced co-worker, it is possible to learn the job from scratch within a couple of weeks. Another interviewee noted that many positions in the firm pertain to summer or weekend jobs, which typically are applied for by new entrants in the labor market. Fostering an employee into a valuable member of a working team is not dependent on previous experience, according to a third respondent. For instance, in his/her firm, some of the restaurant staff are former carpenters. Two respondents reported that at least some labor market experience is valuable, but not necessarily from the same occupation as the one advertised.

Overall, our respondents regarded personal traits and attitudes—something that they try to figure out from the application letters but is better assessed during a job interview—as more important than formal qualifications. Being motivated, service minded, conscientious, interested in a long-term employment relationship with the firm, and able to fit in with the workplace culture, are characteristics highly sought after by employers in our survey.

Although strong conclusions cannot be drawn from our small survey, the findings do lend support to our simple design of the application letters in the field experiment. They also contribute to the understanding of the reasons behind the low callback rates in general, why they were higher for females than males, and why applicants with completed SFI and previous experience did not receive more callbacks than applicants without these qualifications. Our findings suggest that employers hiring applicants from the Middle East and Africa in low-skilled jobs view SFI and experience as weak signals of productivity. When judging such applicants, employers seem to have a functional approach, considering the requirements of the task at hand and the potential for a long-term relationship rather than formal qualifications.

5 Conclusions

Using a correspondence test, we have investigated the impact of completed language training in Swedish (SFI) and experience from low-skilled jobs for recently arrived foreign-born job applicants in the Swedish labor market. Applications were sent from fictitious Syrian refugees with different language skills and previous work experiences to employers advertising low-skilled job vacancies. We are unable to demonstrate large positive effects of SFI or previous experience on callback rates. However, female job applicants were significantly more likely than male applicants to receive callback from employers. We have complemented the correspondence study with interviews with a select number of employers, in order to shed light on potential mechanisms behind our experimental results.

A review of previous research indicates that initial labor market experience is associated with improved long-term labor market outcomes for foreign-born individuals. Language proficiency also seems to be related to better prospects in the labor market for immigrants according to the literature. But employers in our experiment did not pay all that much attention to whether or not a job applicant has completed language training in Swedish or have any previous work experience. Neither completed SFI nor a low-skilled job thus seems to provide any significant positive signaling value when refugees from Syria apply for low-skilled jobs through formal channels. One interpretation of these findings is that the positive effects suggested in the literature of these qualifications are driven by other mechanisms than signaling, which are not possible to account for in our experimental setting, such as better access to informal career paths, information, human capital accumulation, or improved professional networks. The respondents in our employer interviews reported that they regard the link between completed SFI and language proficiency as being rather weak and that applicants' personal traits, like motivation and conscientiousness, are more important than previous experience. Applications for low-skilled jobs are typically very short and simple, according to our respondents, and more information about the personal characteristics that employers are looking for could contribute to higher callback rates.

The observation (in Section 2) that females from Middle Eastern and African countries exhibit lower employment rates compared to men from these regions does not seem to be explained by females being less likely to be contacted for an interview. The two most important occupations in our experiment, restaurant assistant and cleaner, are dominated by females and there are also other correspondence studies that find that females have a higher callback rate than males in female-dominated occupations, as discussed in the Introduction. The interviews revealed that some employers regard females as more conscientious and adaptable than males. Disregarding any general equilibrium effects and differences in selection into employment for males and females, our results in-

dicate that the integration of foreign-born females would be improved if they to a greater extent apply for jobs.

As is usual in correspondence studies, a number of caveats are in order. It should be emphasized that we do not test the effect of language skills per se, but the signal from completed language training. It is not obvious how employers interpret the formulation that the job applicant has completed "the entire SFI program" in terms of language skills, as employers may be illinformed about the contents and structure of SFI (although the employer interviews showed that some of them were practically involved in the program). We cannot distinguish between effects due to ignorance or misconceptions regarding SFI on part of employers and effects based on actual knowledge. Moreover, given previous experience, completed formal language training may not be seen as a large additional advantage. It is also conceivable that SFI has no positive effect on the callback rate if an explicit reference to such training reminds the employer that the applicant belongs to a group with a perceived low productivity (foreign born) or if the employer is reminded of a training to which he or she is skeptical, even if completed by the applicant. This skepticism may be due to the requirements or quality in SFI perceived as being too low.

The fact that unemployment is assigned such a small role by employers in our study may be related to the fact that the group we examine has a generally low employment rate. For the group we are investigating, it may be that the work experience is considered to be too short (between 14 and 25 months, depending on time of application) or not sufficiently qualified. However, it is not possible to distinguish between effects of different lengths of work experience and calendar effects in our experiment.

Furthermore, the labor market we study is characterized by stronger competition for available jobs than in more skilled occupations, according to the Swedish Public Employment Service (2019), whose regularly published indices indicate the extent of labor shortages or excess supply in various jobs. The observation that competition for low-skilled jobs can be fierce is also supported by our employer interviews, in which the respondents reported that they receive a large number of applications per vacant position, and by public statements from other employers (Bergfors, 2011; Jureskog, 2022). Fierce competition may have particularly negative consequences for the vulnerable group included in the experiment and not only lead to a generally low callback rate, but also to a small return on the signals of Swedish language proficiency and productivity acquired through work experience in the applications.

References

- Adermon, A. and L. Hensvik (2022). "Gig-jobs: stepping stones or dead ends?" *Labour Economics* 76, 102171.
- Aldén, L. and M. Hammarstedt (2015). "Utrikes födda på 2000-talets arbetsmarknaden översikt och förklaringar till situationen". *Ekonomisk Debatt* 43.3, 77– 89.
- Arai, M., M. Bursell, and L. Nekby (2016). "The reverse gender gap in ethnic discrimination: Employer stereotypes of men and women with Arabic names". *International Migration Review* 50.2, 385–412.
- Arendt, J. N., I. Bolvig, M. Foged, L. Hasager, and G. Peri (2020). "Language training and refugees' integration". Tech. rep. National Bureau of Economic Research.
- Baert, S., A. Albanese, S. du Gardein, J. Ovaere, and J. Stappers (2017). "Does work experience mitigate discrimination?" *Economics Letters* 155, 35–38.
- Barth, E., B. Bratsberg, and O. Raaum (2012). "Immigrant wage profiles within and between establishments". *Labour Economics* 19.4, 541–556.
- Becker, G. S. (1962). "Investment in human capital: A theoretical analysis". *Journal of political economy* 70.5, Part 2, 9–49.
- Bergfors, R. (2011). "Sänk momsen och förbättra attityden". Dagens Industri.
- Bertrand, M. and S. Mullainathan (2004). "Are Emily and Greg more employable than Lakisha and Jamal? A field experiment on labor market discrimination". *American Economic Review* 94.4, 991–1013.
- Boguslaw, J. (2012). *Svensk invandringspolitik under 500 år: 1512-2012*. Studentlitteratur.
- Brell, C., C. Dustmann, and I. Preston (2020). "The labor market integration of refugee migrants in high-income countries". *Journal of Economic Perspectives* 34.1, 94–121.
- Brenzel, H. and M. Reichelt (2018). "Job Mobility as a New Explanation for the Immigrant-Native Wage Gap: A Longitudinal Analysis of the German Labor Market1". *International Migration Review* 52.3, 724–749.
- Calmfors, L., P. Danielsson, S. Ek, A.-S. Kolm, T. Pekkarinen, and P. Skedinger (2018). *Hur ska fler komma in på arbetsmarknaden?* Dialogos Förlag.
- Calvo-Armengol, A. and M. O. Jackson (2004). "The effects of social networks on employment and inequality". *American Economic Review* 94.3, 426–454.
- Carlsson, M. (2010). "Experimental evidence of discrimination in the hiring of first-and second-generation immigrants". *Labour* 24.3, 263–278.
- Carlsson, M. (2011). "Does hiring discrimination cause gender segregation in the Swedish labor market?" *Feminist Economics* 17.3, 71–102.

- Carlsson, M. and S. Eriksson (2019). "Age discrimination in hiring decisions: Evidence from a field experiment in the labor market". *Labour Economics* 59, 173–183.
- Chiswick, B. R., Y. L. Lee, and P. W. Miller (2005). "Immigrant earnings: a longitudinal analysis". *Review of Income and Wealth* 51.4, 485–503.
- Chiswick, B. R. and P. W. Miller (2015). "International migration and the economics of language". *Handbook of the economics of international migration*. Vol. 1. Elsevier, 211–269.
- Edo, A., N. Jacquemet, and C. Yannelis (2019). "Language skills and homophilous hiring discrimination: Evidence from gender and racially differentiated applications". *Review of Economics of the Household* 17, 349– 376.
- Ek, S., M. Hammarstedt, and P. Skedinger (2020). *Enkla jobb och kunskaper i svenska—nycklar till integration?* Centre for Business and Policy Research (SNS), Stockholm.
- Erlandsson, A. (2022). "Gender, Parenthood, Ethnicity and Discrimination in the Labor Market". PhD thesis. Swedish Institute of Social Research, Stockholm University.
- Ferrer, A., D. A. Green, and W. C. Riddell (2006). "The effect of literacy on immigrant earnings". *Journal of Human Resources* 41.2, 380–410.
- Himmler, O. and R. Jäckle (2018). "Literacy and the migrant–native wage gap". *Review of Income and Wealth* 64.3, 592–625.
- Husted, L., H. Skyt Nielsen, M. Rosholm, and N. Smith (2001). "Employment and wage assimilation of male first-generation immigrants in Denmark". *International Journal of Manpower* 22.1/2, 39–71.
- Jureskog, J. (2022). "Säkra återväxten i restaurangbranschen". *Dagens Industri*.
- Lochmann, A., H. Rapoport, and B. Speciale (2019). "The effect of language training on immigrants' economic integration: Empirical evidence from France". *European Economic Review* 113, 265–296.
- Manzi, F. (2019). "Are the processes underlying discrimination the same for women and men? A critical review of congruity models of gender discrimination". *Frontiers in psychology* 10, 469.
- OECD (2018). Settling in 2018: Indicators of immigrant integration.
- Spence, M. (1973). "Job Market Signaling". *The Quarterly Journal of Economics* 87.3, 355–374.
- Swedish National Agency for Education (2018). *Kommunal vuxenutbildning i svenska för invandrare*.

- Swedish Public Employment Service (2019). Var finns jobben? Bedömning för 2019 och på fem års sikt.
- Swedish Schools Inspectorate (2010). Svenskundervisning för invandrare (sfi) —en granskning av hur utbildningen formas efter deltagarnas förutsättningar och mål.
- Vernby, K. and R. Dancygier (2019). "Can immigrants counteract employer discrimination? A factorial field experiment reveals the immutability of ethnic hierarchies". *PloS one* 14.7, e0218044.
- Yao, Y. and J. C. van Ours (2015). "Language skills and labor market performance of immigrants in the Netherlands". *Labour Economics* 34, 76–85.

Appendix A Application letter and resume

Job application, with cover letter and resumé. [] indicates text not shown here. Text without {} refers to applicant 1 (see Section 3.1) and text with {} added refers to the other applicants.

Page 1

To: [Employer]

In your ad, I read that you are looking for a [Professional role]. I'm very interested in the job. I'm 23 years old. I was born in Damascus in Syria and received a residence permit in Sweden in 2016. I currently live in Stockholm. {I have completed the entire education in Swedish for immigrants (SFI).}¹

I have a high school diploma from my home country. {I currently work as a restaurant assistant at [Fast-food chain] in Stockholm (references provided on request).}²

I'm single and in my spare time I like to work out, listen to music and meet friends.

I hope to meet you in person and send my application.

Page 2

Resumé

Personal Information: Name: [Name] Date of birth: [Date of birth] Place of birth: Damascus, Syria Address: [Residential address] Email: [Email address] Telephone: [Telephone number]

Education: High school diploma from Damascus, Syria {Completed the entire education Swedish for immigrants (SFI)}¹

Work experience: 201608 - Registered at the Swedish Public Employment Service {201711 - Employed as a restaurant assistant at [Fast food chain]}²

 $^{^1\{}$ } indicates text included for applicant 2 and 4, see Section 3.1.

 $^{^{2}}$ { } indicates text included for applicant 3 and 4, see Section 3.1.

Economic Studies

- 1987:1 Haraldson, Marty. To Care and To Cure. A linear programming approach to national health planning in developing countries. 98 pp.
- 1989:1 Chryssanthou, Nikos. The Portfolio Demand for the ECU. A Transaction Cost Approach. 42 pp.
- 1989:2 Hansson, Bengt. Construction of Swedish Capital Stocks, 1963-87. An Application of the Hulten-Wykoff Studies. 37 pp.
- 1989:3 Choe, Byung-Tae. Some Notes on Utility Functions Demand and Aggregation. 39 pp.
- 1989:4 Skedinger, Per. Studies of Wage and Employment Determination in the Swedish Wood Industry. 89 pp.
- 1990:1 Gustafson, Claes-Håkan. Inventory Investment in Manufacturing Firms. Theory and Evidence. 98 pp.
- 1990:2 Bantekas, Apostolos. The Demand for Male and Female Workers in Swedish Manufacturing. 56 pp.
- 1991:1 Lundholm, Michael. Compulsory Social Insurance. A Critical Review. 109 pp.
- 1992:1 Sundberg, Gun. The Demand for Health and Medical Care in Sweden. 58 pp.
- 1992:2 Gustavsson, Thomas. No Arbitrage Pricing and the Term Structure of Interest Rates. 47 pp.
- 1992:3 Elvander, Nils. Labour Market Relations in Sweden and Great Britain. A Comparative Study of Local Wage Formation in the Private Sector during the 1980s. 43 pp.
- 12 Dillén, Mats. Studies in Optimal Taxation, Stabilization, and Imperfect Competition. 1993. 143 pp.
- 13 Banks, Ferdinand E., A Modern Introduction to International Money, Banking and Finance. 1993. 303 pp.
- 14 Mellander, Erik. Measuring Productivity and Inefficiency Without Quantitative Output Data. 1993. 140 pp.
- 15 Ackum Agell. Susanne. Essays on Work and Pay. 1993. 116 pp.
- 16 Eriksson, Claes. Essays on Growth and Distribution. 1994. 129 pp.
- 17 Banks, Ferdinand E., A Modern Introduction to International Money, Banking and Finance. 2nd version, 1994. 313 pp.

- 18 Apel, Mikael. Essays on Taxation and Economic Behavior. 1994. 144 pp.
- 19 Dillén, Hans. Asset Prices in Open Monetary Economies. A Contingent Claims Approach. 1994. 100 pp.
- 20 Jansson, Per. Essays on Empirical Macroeconomics. 1994. 146 pp.
- 21 Banks, Ferdinand E., A Modern Introduction to International Money, Banking, and Finance. 3rd version, 1995. 313 pp.
- 22 Dufwenberg, Martin. On Rationality and Belief Formation in Games. 1995. 93 pp.
- 23 Lindén, Johan. Job Search and Wage Bargaining. 1995. 127 pp.
- 24 Shahnazarian, Hovick. Three Essays on Corporate Taxation. 1996. 112 pp.
- Svensson, Roger. Foreign Activities of Swedish Multinational Corporations. 1996.
 166 pp.
- 26 Sundberg, Gun. Essays on Health Economics. 1996. 174 pp.
- 27 Sacklén, Hans. Essays on Empirical Models of Labor Supply. 1996. 168 pp.
- 28 Fredriksson, Peter. Education, Migration and Active Labor Market Policy. 1997. 106 pp.
- 29 Ekman, Erik. Household and Corporate Behaviour under Uncertainty. 1997. 160 pp.
- 30 Stoltz, Bo. Essays on Portfolio Behavior and Asset Pricing. 1997. 122 pp.
- 31 Dahlberg, Matz. Essays on Estimation Methods and Local Public Economics. 1997. 179 pp.
- 32 Kolm, Ann-Sofie. Taxation, Wage Formation, Unemployment and Welfare. 1997. 162 pp.
- Boije, Robert. Capitalisation, Efficiency and the Demand for Local Public Services. 1997.
 148 pp.
- 34 Hort, Katinka. On Price Formation and Quantity Adjustment in Swedish Housing Markets. 1997. 185 pp.
- 35 Lindström, Thomas. Studies in Empirical Macroeconomics. 1998. 113 pp.
- 36 Hemström, Maria. Salary Determination in Professional Labour Markets. 1998. 127 pp.
- 37 Forsling, Gunnar. Utilization of Tax Allowances and Corporate Borrowing. 1998. 96 pp.
- 38 Nydahl, Stefan. Essays on Stock Prices and Exchange Rates. 1998. 133 pp.
- 39 Bergström, Pål. Essays on Labour Economics and Econometrics. 1998. 163 pp.

- 40 Heiborn, Marie. Essays on Demographic Factors and Housing Markets. 1998. 138 pp.
- 41 Åsberg, Per. Four Essays in Housing Economics. 1998. 166 pp.
- 42 Hokkanen, Jyry. Interpreting Budget Deficits and Productivity Fluctuations. 1998. 146 pp.
- 43 Lunander, Anders. Bids and Values. 1999. 127 pp.
- 44 Eklöf, Matias. Studies in Empirical Microeconomics. 1999. 213 pp.
- Johansson, Eva. Essays on Local Public Finance and Intergovernmental Grants. 1999.
 156 pp.
- 46 Lundin, Douglas. Studies in Empirical Public Economics. 1999. 97 pp.
- 47 Hansen, Sten. Essays on Finance, Taxation and Corporate Investment. 1999. 140 pp.
- 48 Widmalm, Frida. Studies in Growth and Household Allocation. 2000. 100 pp.
- 49 Arslanogullari, Sebastian. Household Adjustment to Unemployment. 2000. 153 pp.
- 50 Lindberg, Sara. Studies in Credit Constraints and Economic Behavior. 2000. 135 pp.
- 51 Nordblom, Katarina. Essays on Fiscal Policy, Growth, and the Importance of Family Altruism. 2000. 105 pp.
- 52 Andersson, Björn. Growth, Saving, and Demography. 2000. 99 pp.
- 53 Åslund, Olof. Health, Immigration, and Settlement Policies. 2000. 224 pp.
- 54 Bali Swain, Ranjula. Demand, Segmentation and Rationing in the Rural Credit Markets of Puri. 2001. 160 pp.
- 55 Löfqvist, Richard. Tax Avoidance, Dividend Signaling and Shareholder Taxation in an Open Economy. 2001. 145 pp.
- 56 Vejsiu, Altin. Essays on Labor Market Dynamics. 2001. 209 pp.
- 57 Zetterström, Erik. Residential Mobility and Tenure Choice in the Swedish Housing Market. 2001. 125 pp.
- 58 Grahn, Sofia. Topics in Cooperative Game Theory. 2001. 106 pp.
- 59 Laséen, Stefan. Macroeconomic Fluctuations and Microeconomic Adjustments. Wages, Capital, and Labor Market Policy. 2001. 142 pp.
- 60 Arnek, Magnus. Empirical Essays on Procurement and Regulation. 2002. 155 pp.
- Jordahl, Henrik. Essays on Voting Behavior, Labor Market Policy, and Taxation. 2002.
 172 pp.

- 62 Lindhe, Tobias. Corporate Tax Integration and the Cost of Capital. 2002. 102 pp.
- 63 Hallberg, Daniel. Essays on Household Behavior and Time-Use. 2002. 170 pp.
- 64 Larsson, Laura. Evaluating Social Programs: Active Labor Market Policies and Social Insurance. 2002. 126 pp.
- 65 Bergvall, Anders. Essays on Exchange Rates and Macroeconomic Stability. 2002. 122 pp.
- 66 Nordström Skans, Oskar. Labour Market Effects of Working Time Reductions and Demographic Changes. 2002. 118 pp.
- 67 Jansson, Joakim. Empirical Studies in Corporate Finance, Taxation and Investment. 2002. 132 pp.
- 68 Carlsson, Mikael. Macroeconomic Fluctuations and Firm Dynamics: Technology, Production and Capital Formation. 2002. 149 pp.
- 69 Eriksson, Stefan. The Persistence of Unemployment: Does Competition between Employed and Unemployed Job Applicants Matter? 2002. 154 pp.
- 70 Huitfeldt, Henrik. Labour Market Behaviour in a Transition Economy: The Czech Experience. 2003. 110 pp.
- 71 Johnsson, Richard. Transport Tax Policy Simulations and Satellite Accounting within a CGE Framework. 2003. 84 pp.
- 72 Öberg, Ann. Essays on Capital Income Taxation in the Corporate and Housing Sectors. 2003. 183 pp.
- 73 Andersson, Fredrik. Causes and Labor Market Consequences of Producer Heterogeneity. 2003. 197 pp.
- 74 Engström, Per. Optimal Taxation in Search Equilibrium. 2003. 127 pp.
- 75 Lundin, Magnus. The Dynamic Behavior of Prices and Investment: Financial Constraints and Customer Markets. 2003. 125 pp.
- 76 Ekström, Erika. Essays on Inequality and Education. 2003. 166 pp.
- 77 Barot, Bharat. Empirical Studies in Consumption, House Prices and the Accuracy of European Growth and Inflation Forecasts. 2003. 137 pp.
- 78 Österholm, Pär. Time Series and Macroeconomics: Studies in Demography and Monetary Policy. 2004. 116 pp.
- 79 Bruér, Mattias. Empirical Studies in Demography and Macroeconomics. 2004. 113 pp.
- 80 Gustavsson, Magnus. Empirical Essays on Earnings Inequality. 2004. 154 pp.
- 81 Toll, Stefan. Studies in Mortgage Pricing and Finance Theory. 2004. 100 pp.
- 82 Hesselius, Patrik. Sickness Absence and Labour Market Outcomes. 2004. 109 pp.
- 83 Häkkinen, Iida. Essays on School Resources, Academic Achievement and Student Employment. 2004. 123 pp.
- 84 Armelius, Hanna. Distributional Side Effects of Tax Policies: An Analysis of Tax Avoidance and Congestion Tolls. 2004. 96 pp.
- 85 Ahlin, Åsa. Compulsory Schooling in a Decentralized Setting: Studies of the Swedish Case. 2004. 148 pp.
- Heldt, Tobias. Sustainable Nature Tourism and the Nature of Tourists' Cooperative Behavior: Recreation Conflicts, Conditional Cooperation and the Public Good Problem.
 2005. 148 pp.
- 87 Holmberg, Pär. Modelling Bidding Behaviour in Electricity Auctions: Supply Function Equilibria with Uncertain Demand and Capacity Constraints. 2005. 43 pp.
- 88 Welz, Peter. Quantitative new Keynesian macroeconomics and monetary policy 2005. 128 pp.
- 89 Ågren, Hanna. Essays on Political Representation, Electoral Accountability and Strategic Interactions. 2005. 147 pp.
- 90 Budh, Erika. Essays on environmental economics. 2005. 115 pp.
- 91 Chen, Jie. Empirical Essays on Housing Allowances, Housing Wealth and Aggregate Consumption. 2005. 192 pp.
- 92 Angelov, Nikolay. Essays on Unit-Root Testing and on Discrete-Response Modelling of Firm Mergers. 2006. 127 pp.
- 93 Savvidou, Eleni. Technology, Human Capital and Labor Demand. 2006. 151 pp.
- 94 Lindvall, Lars. Public Expenditures and Youth Crime. 2006. 112 pp.
- 95 Söderström, Martin. Evaluating Institutional Changes in Education and Wage Policy. 2006. 131 pp.
- 96 Lagerström, Jonas. Discrimination, Sickness Absence, and Labor Market Policy. 2006. 105 pp.
- 97 Johansson, Kerstin. Empirical essays on labor-force participation, matching, and trade. 2006. 168 pp.
- 98 Ågren, Martin. Essays on Prospect Theory and the Statistical Modeling of Financial Returns. 2006. 105 pp.

- 99 Nahum, Ruth-Aïda. Studies on the Determinants and Effects of Health, Inequality and Labour Supply: Micro and Macro Evidence. 2006. 153 pp.
- 100 Žamac, Jovan. Education, Pensions, and Demography. 2007. 105 pp.
- 101 Post, Erik. Macroeconomic Uncertainty and Exchange Rate Policy. 2007. 129 pp.
- 102 Nordberg, Mikael. Allies Yet Rivals: Input Joint Ventures and Their Competitive Effects. 2007. 122 pp.
- 103 Johansson, Fredrik. Essays on Measurement Error and Nonresponse. 2007. 130 pp.
- 104 Haraldsson, Mattias. Essays on Transport Economics. 2007. 104 pp.
- 105 Edmark, Karin. Strategic Interactions among Swedish Local Governments. 2007. 141 pp.
- 106 Oreland, Carl. Family Control in Swedish Public Companies. Implications for Firm Performance, Dividends and CEO Cash Compensation. 2007. 121 pp.
- 107 Andersson, Christian. Teachers and Student Outcomes: Evidence using Swedish Data. 2007. 154 pp.
- 108 Kjellberg, David. Expectations, Uncertainty, and Monetary Policy. 2007. 132 pp.
- 109 Nykvist, Jenny. Self-employment Entry and Survival Evidence from Sweden. 2008. 94 pp.
- 110 Selin, Håkan. Four Empirical Essays on Responses to Income Taxation. 2008. 133 pp.
- 111 Lindahl, Erica. Empirical studies of public policies within the primary school and the sickness insurance. 2008. 143 pp.
- 112 Liang, Che-Yuan. Essays in Political Economics and Public Finance. 2008. 125 pp.
- Elinder, Mikael. Essays on Economic Voting, Cognitive Dissonance, and Trust. 2008.120 pp.
- 114 Grönqvist, Hans. Essays in Labor and Demographic Economics. 2009. 120 pp.
- 115 Bengtsson, Niklas. Essays in Development and Labor Economics. 2009. 93 pp.
- 116 Vikström, Johan. Incentives and Norms in Social Insurance: Applications, Identification and Inference. 2009. 205 pp.
- 117 Liu, Qian. Essays on Labor Economics: Education, Employment, and Gender. 2009. 133 pp.
- 118 Glans, Erik. Pension reforms and retirement behaviour. 2009. 126 pp.
- 119 Douhan, Robin. Development, Education and Entrepreneurship. 2009.

- 120 Nilsson, Peter. Essays on Social Interactions and the Long-term Effects of Early-life Conditions. 2009. 180 pp.
- 121 Johansson, Elly-Ann. Essays on schooling, gender, and parental leave. 2010. 131 pp.
- 122 Hall, Caroline. Empirical Essays on Education and Social Insurance Policies. 2010. 147 pp.
- 123 Enström-Öst, Cecilia. Housing policy and family formation. 2010. 98 pp.
- 124 Winstrand, Jakob. Essays on Valuation of Environmental Attributes. 2010. 96 pp.
- 125 Söderberg, Johan. Price Setting, Inflation Dynamics, and Monetary Policy. 2010. 102 pp.
- 126 Rickne, Johanna. Essays in Development, Institutions and Gender. 2011. 138 pp.
- Hensvik, Lena. The effects of markets, managers and peers on worker outcomes. 2011.179 pp.
- 128 Lundqvist, Heléne. Empirical Essays in Political and Public. 2011. 157 pp.
- 129 Bastani, Spencer. Essays on the Economics of Income Taxation. 2012. 257 pp.
- 130 Corbo, Vesna. Monetary Policy, Trade Dynamics, and Labor Markets in Open Economies. 2012. 262 pp.
- 131 Nordin, Mattias. Information, Voting Behavior and Electoral Accountability. 2012. 187 pp.
- 132 Vikman, Ulrika. Benefits or Work? Social Programs and Labor Supply. 2013. 161 pp.
- 133 Ek, Susanne. Essays on unemployment insurance design. 2013. 136 pp.
- 134 Österholm, Göran. Essays on Managerial Compensation. 2013. 143 pp.
- 135 Adermon, Adrian. Essays on the transmission of human capital and the impact of technological change. 2013. 138 pp.
- 136 Kolsrud, Jonas. Insuring Against Unemployment 2013. 140 pp.
- 137 Hanspers, Kajsa. Essays on Welfare Dependency and the Privatization of Welfare Services. 2013. 208 pp.
- 138 Persson, Anna. Activation Programs, Benefit Take-Up, and Labor Market Attachment. 2013. 164 pp.
- 139 Engdahl, Mattias. International Mobility and the Labor Market. 2013. 216 pp.
- 140 Krzysztof Karbownik. Essays in education and family economics. 2013. 182 pp.

- 141 Oscar Erixson. Economic Decisions and Social Norms in Life and Death Situations. 2013.183 pp.
- 142 Pia Fromlet. Essays on Inflation Targeting and Export Price Dynamics. 2013. 145 pp.
- 143 Daniel Avdic. Microeconometric Analyses of Individual Behavior in Public Welfare Systems. Applications in Health and Education Economics. 2014. 176 pp.
- 144 Arizo Karimi. Impacts of Policies, Peers and Parenthood on Labor Market Outcomes. 2014. 221 pp.
- 145 Karolina Stadin. Employment Dynamics. 2014. 134 pp.
- 146 Haishan Yu. Essays on Environmental and Energy Economics. 132 pp.
- 147 Martin Nilsson. Essays on Health Shocks and Social Insurance. 139 pp.
- 148 Tove Eliasson. Empirical Essays on Wage Setting and Immigrant Labor Market Opportunities. 2014. 144 pp.
- 149 Erik Spector. Financial Frictions and Firm Dynamics. 2014. 129 pp.
- 150 Michihito Ando. Essays on the Evaluation of Public Policies. 2015. 193 pp.
- 151 Selva Bahar Baziki. Firms, International Competition, and the Labor Market. 2015. 183 pp.
- 152 Fredrik Sävje. What would have happened? Four essays investigating causality. 2015. 229 pp.
- 153 Ina Blind. Essays on Urban Economics. 2015. 197 pp.
- 154 Jonas Poulsen. Essays on Development and Politics in Sub-Saharan Africa. 2015. 240 pp.
- 155 Lovisa Persson. Essays on Politics, Fiscal Institutions, and Public Finance. 2015. 137 pp.
- 156 Gabriella Chirico Willstedt. Demand, Competition and Redistribution in Swedish Dental Care. 2015. 119 pp.
- 157 Yuwei Zhao de Gosson de Varennes. Benefit Design, Retirement Decisions and Welfare Within and Across Generations in Defined Contribution Pension Schemes. 2016. 148 pp.
- 158 Johannes Hagen. Essays on Pensions, Retirement and Tax Evasion. 2016. 195 pp.
- 159 Rachatar Nilavongse. Housing, Banking and the Macro Economy. 2016. 156 pp.
- 160 Linna Martén. Essays on Politics, Law, and Economics. 2016. 150 pp.
- 161 Olof Rosenqvist. Essays on Determinants of Individual Performance and Labor Market Outcomes. 2016. 151 pp.
- 162 Linuz Aggeborn. Essays on Politics and Health Economics. 2016. 203 pp.

- 163 Glenn Mickelsson. DSGE Model Estimation and Labor Market Dynamics. 2016. 166 pp.
- 164 Sebastian Axbard. Crime, Corruption and Development. 2016. 150 pp.
- 165 Mattias Öhman. Essays on Cognitive Development and Medical Care. 2016. 181 pp.
- 166 Jon Frank. Essays on Corporate Finance and Asset Pricing. 2017. 160 pp.
- 167 Ylva Moberg. Gender, Incentives, and the Division of Labor. 2017. 220 pp.
- 168 Sebastian Escobar. Essays on inheritance, small businesses and energy consumption. 2017. 194 pp.
- 169 Evelina Björkegren. Family, Neighborhoods, and Health. 2017. 226 pp.
- 170 Jenny Jans. Causes and Consequences of Early-life Conditions. Alcohol, Pollution and Parental Leave Policies. 2017. 209 pp.
- 171 Josefine Andersson. Insurances against job loss and disability. Private and public interventions and their effects on job search and labor supply. 2017. 175 pp.
- 172 Jacob Lundberg. Essays on Income Taxation and Wealth Inequality. 2017. 173 pp.
- 173 Anna Norén. Caring, Sharing, and Childbearing. Essays on Labor Supply, Infant Health, and Family Policies. 2017. 206 pp.
- 174 Irina Andone. Exchange Rates, Exports, Inflation, and International Monetary Cooperation. 2018. 174 pp.
- 175 Henrik Andersson. Immigration and the Neighborhood. Essays on the Causes and Consequences of International Migration. 2018. 181 pp.
- 176 Aino-Maija Aalto. Incentives and Inequalities in Family and Working Life. 2018. 131 pp.
- 177 Gunnar Brandén. Understanding Intergenerational Mobility. Inequality, Student Aid and Nature-Nurture Interactions. 2018. 125 pp.
- 178 Mohammad H. Sepahvand. Essays on Risk Attitudes in Sub-Saharan Africa. 2019. 215 pp.
- 179 Mathias von Buxhoeveden. Partial and General Equilibrium Effects of Unemployment Insurance. Identification, Estimation and Inference. 2019. 89 pp.
- 180 Stefano Lombardi. Essays on Event History Analysis and the Effects of Social Programs on Individuals and Firms. 2019. 150 pp.
- 181 Arnaldur Stefansson. Essays in Public Finance and Behavioral Economics. 2019. 191 pp.
- 182 Cristina Bratu. Immigration: Policies, Mobility and Integration. 2019. 173 pp.
- 183 Tamás Vasi. Banks, Shocks and Monetary Policy. 2020. 148 pp.

- 184 Jonas Cederlöf. Job Loss: Consequences and Labor Market Policy. 2020. 213 pp.
- 185 Dmytro Stoyko. Expectations, Financial Markets and Monetary Policy. 2020. 153 pp.
- 186 Paula Roth. Essays on Inequality, Insolvency and Innovation. 2020. 191 pp.
- 187 Fredrik Hansson. Consequences of Poor Housing, Essays on Urban and Health Economics. 2020. 143 pp.
- 188 Maria Olsson. Essays on Macroeconomics: Wage Rigidity and Aggregate Fluctuations. 2020. 130 pp.
- 189 Dagmar Müller. Social Networks and the School-to-Work Transition. 2020. 146 pp.
- 190 Maria Sandström. Essays on Savings and Intangible Capital. 2020. 129 pp.
- Anna Thoresson. Wages and Their Impact on Individuals, Households and Firms. 2020.
 220 pp.
- 192. Jonas Klarin. Empirical Essays in Public and Political Economics. 2020. 129 pp.
- André Reslow. Electoral Incentives and Information Content in Macroeconomic Forecasts. 2021. 184 pp.
- Davide Cipullo. Political Careers, Government Stability, and Electoral Cycles. 2021.
 308 pp.
- 195. Olle Hammar. The Mystery of Inequality: Essays on Culture, Development, and Distributions. 2021. 210 pp.
- 196. J. Lucas Tilley. Inputs and Incentives in Education. 2021. 184 pp.
- 197. Sebastian Jävervall. Corruption, Distortions and Development. 2021. 215 pp.
- Vivika Halapuu. Upper Secondary Education: Access, Choices and Graduation. 2021. 141 pp.
- Charlotte Paulie. Essays on the Distribution of Production, Prices and Wealth. 2021. 141 pp.
- Kerstin Westergren. Essays on Inflation Expectations, Monetary Policy and Tax Reform. 2021. 124 pp.
- 201. Melinda Süveg. Finance, Shocks, Competition and Price Setting. 2021. 137 pp.
- Adrian Poignant. Gold, Coal and Iron. Essays on Industrialization and Economic Development. 2022. 214 pp.
- 203. Daniel Bougt. A Sequence of Essays on Sequences of Auctions. 2022. 188 pp.

- 204. Lillit Ottosson. From Welfare to Work. Financial Incentives, Active Labor Market Policies, and Integration Programs. 2022. 219 pp.
- 205. Yaroslav Yakymovych. Workers and Occupations in a Changing Labour Market. The Heterogeneous Effects of Mass Layoffs and Social Safety Nets. 2022. 212 pp.
- 206. Raoul van Maarseveen. Urbanization and Education. The Effect of Childhood Urban Residency on Educational Attainment. 2022. 210 pp.
- 207. Sofia Hernnäs. Automation and the Consequences of Occupational Decline. 2022. 229 pp.
- 208. Simon Ek, Structural Change, Match Quality, and Integration in the Labor Market. 2023. 176 pp.
- 209. Maximiliano Sosa Andres, A Risky and Polarized World: Essays on Uncertainty, Ideology and Foreign Policy. 2023. 226 pp.